UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D. C. 20549

Form 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15 (d) **OF THE SECURITIES EXCHANGE ACT OF 1934**

For the quarter ended December 31, 2004

Commission file number 1-14643

STERIS Corporation (Exact name of registrant as specified in its charter)

Ohio (State or other jurisdiction of incorporation or organization)

5960 Heisley Road, Mentor, Ohio 44060-1834 (Address of principal executive offices)

34-1482024 (IRS Employer Identification No.)

440-354-2600 (Registrant's telephone number, including area code)

Indicate by check mark whether the Registrant (1) has filed all reports required to be filed by Section 13 or 15 (d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the Registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act). Yes 🗵 No 🗆

The number of Common Shares outstanding as of January 31, 2005: 69,294,481

STERIS Corporation Form 10-Q Index

Page

-		· ·
Part I	 Financial I 	nformation
-		

	Item 1.	Financial Statements (unaudited, except Consolidated Balance Sheet as of March 31, 2004)	3
		Consolidated Balance Sheets as of December 31, 2004 and March 31, 2004	3
		Consolidated Statements of Income for the Three and Nine Month Periods Ended December 31, 2004 and December 31, 2003	4
		Consolidated Statements of Cash Flows for the Nine Month Periods Ended December 31, 2004 and December 31, 2003	5
		Notes to Consolidated Financial Statements	6
		Report of Independent Registered Public Accounting Firm	15
	Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	16
	Item 3.	Quantitative and Qualitative Disclosures About Market Risk	30
	Item 4.	Controls and Procedures	30
<u>Part II -</u>	Other Info	<u>ormation</u>	
	Item 1.	Legal Proceedings	31
	Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	31
	Item 5.	Other Information	31
	Item 6.	Exhibits	31
		Signature	33

PART I - FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

STERIS CORPORATION CONSOLIDATED BALANCE SHEETS (in thousands)

	December 31, 2004	March 31, 2004
	(Unaudited)	
Assets		
Current assets:	* ****	
Cash and cash equivalents	\$ 62,196	\$ 80,408
Accounts receivable (net of allowances of \$8,846 and \$8,623, respectively)	249,003	253,797
Inventories	103,646	98,249
Current portion of deferred income taxes	18,454	18,246
Prepaid expenses and other current assets	9,955	10,338
Total current assets	443,254	461,038
Property, plant, and equipment, net	389,119	374,102
Goodwill and intangibles, net	283,901	230,993
Other assets	2,482	2,037
Total assets	\$1,118,756	\$1,068,170
Liabilities and shareholders' equity		
Current liabilities:		
Current portion of long-term indebtedness	\$ 2,169	\$ 4,049
Accounts payable	52,919	67,988
Accrued income taxes	12,057	2,277
Accrued payroll and other related liabilities	35,941	41,972
Accrued expenses and other	82,730	72,502
Total current liabilities	185,816	188,788
Long-term indebtedness	103,432	109,090
Deferred income taxes	35,110	29,568
Other liabilities	63,350	60,025
Total liabilities	387,708	387,471
Serial preferred shares, without par value; 3,000 shares authorized; no shares issued or outstanding		
Common Shares, without par value; 300,000 shares authorized; issued and outstanding shares of 69,176 and 69,946, respectively	202,341	224,999
Retained earnings	512,513	451,546
Accumulated other comprehensive (loss) income:	012,010	101,010
Minimum pension liability	(4,582)	(4,582)
Cumulative foreign currency translation adjustment	20,776	8,736
Total shareholders' equity	731,048	680,699
Total liabilities and shareholders' equity	\$1,118,756	\$1,068,170
	, , , , , , , , , , , , , , , , , , , ,	. ,,

See notes to consolidated financial statements.

STERIS CORPORATION CONSOLIDATED STATEMENTS OF INCOME (in thousands, except per share amounts) (Unaudited)

		Three Months Ended December 31,		Nine Months December				
	2004		2003		2004		2003	
Revenues:								
Product	\$ 193,11	1 \$1	91,362	\$ 5	538,148	\$ 5	549,481	
Service	91,27	'8	82,924	2	265,884	2	241,474	
Total revenues	284,38	89 2	74,286	8	304,032	7	790,955	
Cost of revenues:								
Product	112,22	.3 1	12,120	3	810,005	3	320,007	
Service	52,60)4	47,069	1	153,085	1	38,666	
Total cost of revenues	164,82	.7 1	59,189	Z	463,090	Z	158,673	
Gross profit	119,56	52 1	15,097	3	340,942	3	332,282	
Operating expenses:								
Selling, general, and administrative	70,46		69,623	2	214,975	2	214,182	
Research and development	9,18	85	7,648		27,173		21,726	
Total operating expenses	79,65	50	77,271	2	242,148	2	235,908	
Income from operations	39,91	2	37,826		98,794		96,374	
Interest expense, net	63		448	_	2,020		1,432	
Income before income tax expense	39,27	/8	37,378		96,774		94,942	
Income tax expense	14,82	.1	10,285		35,807		31,008	
Net income	\$ 24,45	57 \$	27,093	\$	60,967	\$	63,934	
Net income per common share - basic	\$ 0.3	s \$	0.39	\$	0.88	\$	0.92	
				*				
Net income per common share - diluted	\$ 0.3	\$5	0.38	\$	0.87	\$	0.91	

See notes to consolidated financial statements.

STERIS CORPORATION CONSOLIDATED STATEMENTS OF CASH FLOWS (in thousands) (Unaudited)

	Nine Mon Decem	
	2004	2003
Operating activities:		
Net income	\$ 60,967	\$ 63,934
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation and amortization	37,493	36,488
Deferred income taxes	5,432	144
Other items	(3,072)	6,882
Changes in operating assets and liabilities, excluding the effects of business acquisitions:		
Accounts receivable	10,476	1,772
Inventories	3,343	(704
Other current assets	629	918
Accounts payable	(19,414)	(30,938
Accruals and other, net	3,203	(11,281
Net cash provided by operating activities	99,057	67,215
nvesting activities:		
Purchases of property, plant, equipment, and intangibles	(38,793)	(48,430
Purchase of business related assets		(2,900
Investments in businesses, net of cash acquired	(53,323)	(36,814
Net cash used in investing activities	(92,116)	(88,144
Financing activities:		
Net (payments) proceeds under credit facilities	(4,398)	46,800
Payments on long-term obligations and capital leases	(3,273)	(4,933
Repurchases of Common Shares	(33,868)	(16,609
Deferred financing fees and debt issuance costs		(537
Stock option and other equity transactions, net	14,581	7,748
Net cash (used in) provided by financing activities	(26,958)	32,469
Effect of exchange rate changes on cash and cash equivalents	1,805	1,371
Decrease) increase in cash and cash equivalents	(18,212)	12,911
Cash and cash equivalents at beginning of period	80,408	25,941
Cash and cash equivalents at end of period	\$ 62,196	\$ 38,852

See notes to consolidated financial statements.

1. Basis of Presentation

Throughout this document, references to "STERIS Corporation," "STERIS," or the "Company," mean STERIS Corporation and its subsidiaries.

The Company's fiscal year ends on March 31. Reference to a particular "year" or "year-end" refers to the Company's fiscal year.

Nature of Operations

The Company develops, manufactures, and markets a combination of equipment, consumables, and services to healthcare, pharmaceutical, industrial, and government customers throughout the world. The Company operates in three business segments: Healthcare, Life Sciences, and STERIS Isomedix Services.

Interim Financial Statements

The accompanying unaudited consolidated financial statements of the Company have been prepared in accordance with U.S. generally accepted accounting principles for interim financial information and the instructions to Form 10-Q and Rule 10-01 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by U.S. generally accepted accounting principles for complete financial statements. In the opinion of management, the accompanying unaudited consolidated financial statements contain all material adjustments (including normal recurring accruals and adjustments) necessary to present fairly the financial condition, results of operations, and cash flows of the Company for the periods presented.

The information included in this Quarterly Report on Form 10-Q should be read in conjunction with "Management's Discussion and Analysis of Financial Condition and Results of Operations," "Quantitative and Qualitative Disclosures About Market Risk," and the consolidated financial statements and notes thereto included in Items 7, 7A, and 8, respectively, of the Company's Annual Report on Form 10-K for the year ended March 31, 2004, filed with the Securities and Exchange Commission on June 14, 2004. The consolidated balance sheet at March 31, 2004 has been derived from the audited consolidated financial statements at that date, but does not include all of the information and footnotes required by U.S. generally accepted accounting principles for complete financial statements.

Principles of Consolidation

The consolidated financial statements include the accounts of the Company and its wholly-owned subsidiaries. Intercompany accounts and transactions have been eliminated upon consolidation.

Use of Estimates

The preparation of financial statements in conformity with U.S. generally accepted accounting principles requires management to make estimates and assumptions in certain circumstances that affect the reported amounts of assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting periods. Actual results could differ from these estimates and, therefore, operating results for the three and nine month periods ended December 31, 2004 are not necessarily indicative of results that may be expected for the full fiscal year ending March 31, 2005.

Reclassifications

Certain prior period amounts have been reclassified to conform to the current period's presentation.

Significant Accounting Policies

A detailed description of the Company's significant and critical accounting policies, estimates, and assumptions is included in the Company's Annual Report on Form 10-K for the year ended March 31, 2004 filed with the Securities and Exchange Commission on June 14, 2004 in the section of Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations," titled, "Critical Accounting Policies, Estimates, and Assumptions," and in Note 1 to the consolidated financial statements, "Accounting Policies." The Company's significant and critical accounting policies, estimates, and assumptions have not changed materially from March 31, 2004.

Stock-based Compensation

The Company has granted nonqualified stock options to certain employees to purchase the Company's Common Shares at the market price on the date of grant. Stock options granted generally become exercisable to the extent of one-fourth of the optioned shares for each full year of employment following the date of grant and expire approximately 10 years after the date of grant, or earlier if an option holder ceases to be employed by the Company. Certain option agreements have provisions that provide for an adjustment to the normal vesting schedule, whereby, options vest on a prorated basis as defined by specific option agreements in the event of employment termination. The Company accounts for stock-based compensation under the provisions of Accounting Principles Board Opinion No. 25 ("APB No. 25"), "Accounting for Stock Issued to Employees," as permitted by Statement of Financial Accounting Standards No. 123 (SFAS No. 123), "Accounting for Stock-Based Compensation," as amended by Statement of Financial Accounting Standards No. 148, "Accounting for Stock-Based Compensation - Transition and Disclosure," and accordingly recognizes no compensation expense when the exercise price equals the market price of the stock on the date of grant.

The following table illustrates the effect on the Company's net income, earnings per basic Common Share, and earnings per diluted Common Share, had compensation cost for all options been determined based upon the fair market value recognition provisions of SFAS No. 123:

	T	hree Mon Decem	ths Ended ber 31,		
		2004	2003	2004	2003
t income:					
As reported	\$2	4,457	\$27,093	\$60,967	\$63,93
Less: Stock-based compensation expense, net of income taxes, assuming the fa	r value method	1,528	1,626	4,501	4,49
Pro forma	\$2	2,929	\$25,467	\$56,466	\$59,43
rnings per common share:					
Basic:					
As reported	\$	0.35	\$ 0.39	\$ 0.88	\$ 0.
Pro forma		0.33	0.37	0.81	0.
Diluted:					
As reported		0.35	0.38	0.87	0.
Pro forma		0.33	0.36	0.81	0.

For the purpose of computing pro forma net income, the fair value of option grants was estimated at their grant date using the Black-Scholes option pricing model and the following assumptions for the three and nine months ended December 31, 2004 and 2003:

	Three Mon Decem			ths Ended ber 31,	
	2004	2003	2004	2003	
	3.54%- 5.50%	3.54%-6.46%	3.54%-5.50%	3.54%-6.46%	
	0%	0%	0%	0%	
	45%	45%	45%	45%	
in years)	5	5	5	5	

The Black-Scholes option pricing model was developed for use in estimating the fair value of traded options that have no vesting restrictions and are fully transferable, characteristics that are not present in the Company's option grants. If the model permitted consideration of the unique characteristics of employee stock options, the resulting estimate of the fair value of the stock options could be different.

Recently Issued Accounting Standards and Accounting Standards Pending Adoption

In December 2004, the Financial Accounting Standards Board finalized Statement of Financial Accounting Standards No. 123R ("SFAS No. 123R"), "Share-Based Payment." This revised standard addresses the accounting for share-based payment transactions in which a company receives employee services in exchange for either equity instruments of the company or liabilities that are based on the fair value of the company's equity instruments. Under the revised standard, companies will no longer be able to account for share-based compensation transactions using the intrinsic value method in accordance with APB No. 25. Instead, companies will be required to account for such transactions using a fair value method and recognize expense in the consolidated statements of income. SFAS No. 123R is effective for interim or annual periods beginning after June 15, 2005 and allows, but does not require, companies to restate the full fiscal year during the period of adoption or restate all periods presented to reflect the impact of expensing share-based payments under SFAS No. 123R. The Company will adopt SFAS No. 123R as of July 1, 2005. The Company has not yet determined which fair value model and transitional provision it will follow. Note 1, "Basis of Presentation," contains pro forma disclosures regarding the effect on the Company's net income, earnings per basic Common Share, and earnings per diluted Common Share, had the Company applied a fair value method of accounting for share-based compensation as prescribed by SFAS No. 123. Depending on the model used to calculate share-based compensation expense in the future and other requirements of SFAS No. 123R, the pro forma disclosures currently used by the Company may not be indicative of the share-based compensation expense that will be recognized in the Company's future financial statements. Further, the structure and timing of future grants may also have differing impacts on future results.

In October 2004, the American Jobs Creation Act of 2004 (the Jobs Creation Act) was signed into law. The Jobs Creation Act contains a number of provisions that might affect the Company's future effective income tax rate. The most significant provisions would allow the Company to elect to deduct from its taxable income 85% of certain eligible dividends received by the Company from non-U.S. subsidiaries before the end of 2005 if those dividends are reinvested in the U.S. for eligible purposes. On December 21, 2004, the Financial Accounting Standards Board issued FASB Staff Position No. FAS 109-2 (FSP No. 109-2), "Accounting and Disclosure Guidance for the Foreign Earnings Repatriation Provision within the American Jobs Creation Act of 2004." FSP No. 109-2 provides accounting and disclosure guidance for the repatriation provision and became effective upon issuance. The Company is currently evaluating the amount, if any, of such eligible dividends that its non-U.S. subsidiaries will remit, as well as the effects the Jobs Creation Act will have on its effective income tax rate and deferred assets and liabilities.

In December 2003, the Medicare Prescription Drug, Improvement and Modernization Act of 2003 (the "Act") was signed into law. The Act expands Medicare benefits, primarily adding a prescription drug benefit for Medicare-eligible retirees beginning in 2006. The law provides a federal subsidy to companies that sponsor post-retirement healthcare plans that provide prescription drug coverage. In January 2004, the Financial Accounting Standards Board issued FSP No. FAS 106-1 and in May 2004 issued FSP No. FAS 106-2 both titled "Accounting and Disclosure Requirements Related to the Medicare Prescription Drug, Improvement and Modernization Act of 2003" ("FSP No. 106-1" and "FSP No. 106-2"). FSP No. 106-1 allows companies to make a one-time election to defer the accounting for the effects of the Act.

FSP No. 106-2 provides guidance on the accounting for the effects of the Act for employers that sponsor post-retirement healthcare plans with prescription drug benefits. Under the guidance of FSP No. 106-1, the Company elected to defer accounting for the effects of the Act. This deferral remains in effect until the appropriate effective date of FSP No. 106-2. For entities that elected deferral and for which the impact of the Act is significant, FSP No. 106-2 is effective for the first interim or annual period beginning after June 15, 2004. Entities for which FSP No. 106-2 does not have a significant impact are permitted to delay recognition of the effects of the Act until the next regularly scheduled measurement date following the issuance of FSP No. 106-2. The Company does not expect that the effects of the Act will have a significant impact on the Company's financial position, results of operations, or cash flows and will therefore delay recognition until the Company's regularly scheduled measurement date. The reported post-retirement benefit cost for the three and nine months ended December 31, 2004, presented in Note 7, "Benefit Plans," does not reflect the effects of the Act on the Company's post-retirement benefit plan.

2. Business Acquisition

On September 15, 2004, the Company announced that a subsidiary had acquired Albert Browne Limited and its subsidiaries ("Browne"), a privately-held manufacturer of chemical indicators, headquartered in Leicester, England for 28.9 million British pounds sterling (\$52,132), net of 3.2 million British pounds sterling (\$5,807) of cash acquired at the close of the transaction. Browne is a leading provider of chemical indicators to healthcare facilities in over 70 countries. The acquisition provides the Company with an established European distribution channel and expands the Company's offering of consumable products used with its broad line of capital equipment for infection control and decontamination.

The purchase price is subject to the final settlement of certain working capital adjustments as guaranteed by the seller. The purchase price of \$59,130, which includes \$1,191 of direct acquisition costs associated with the transaction, has been preliminarily allocated to net assets and goodwill and is subject to further adjustment as the Company finalizes costs associated with the acquisition and the valuation of identified intangible assets. Through December 31, 2004, \$23,108 has been preliminarily allocated to goodwill within the Company's Healthcare segment and \$23,871 has been preliminarily allocated to identified intangible assets, such as trademarks, technology and patents, customer lists, and non-competition agreements, which are expected to be amortized over a weighted average life of approximately seven years. The consolidated financial statements include the operating results of Browne from the date of acquisition. Pro forma results of operations have not been presented because the effect of this acquisition is not deemed significant.

3. Common Shares

Basic earnings per share is calculated based upon the weighted average number of Common Shares outstanding. Diluted earnings per share is calculated based upon the weighted average number of Common Shares outstanding plus the dilutive effect of Common Share equivalents determined using the treasury stock method. The following is a summary of Common Shares and Common Share equivalents outstanding used in the calculations of basic and diluted earnings per share:

	Three Mon Decem			
	2004 2003		2004	2003
		(shares in t	housands)	
Weighted average Common Shares outstanding - basic	69,134	69,468	69,206	69,425
Dilutive effect of Common Share equivalents	704	1,091	802	1,206
Weighted average Common Shares and equivalents - diluted	69,838	70,559	70,008	70,631

Options to purchase the following number of Common Shares at the following weighted average exercise prices were outstanding but excluded from the computation of diluted earnings per share because the exercise prices were greater than the average market price for the Common Shares during the period:

		nths Ended Iber 31,		ths Ended ber 31,
	2004	2003	2004	2003
		(shares in t	housands)	
Number of Common Share options	2,554	1,766	2,546	699
Weighted average exercise price	\$ 26.05	\$ 25.29	\$ 26.06	\$ 29.43

4. Comprehensive Income

Statement of Financial Accounting Standards No. 130, "Reporting Comprehensive Income," establishes standards for reporting comprehensive income. Comprehensive income includes net income as currently reported under U.S. generally accepted accounting principles and other comprehensive income. Other comprehensive income considers the effect of additional economic events that are not required to be recorded in determining net income, but rather are reported as a separate component of shareholders' equity. The following table illustrates the components of the Company's comprehensive income for the three and nine month periods ended December 31, 2004 and 2003:

		nths Ended ber 31,		ths Ended ber 31,
	2004	2003	2004	2003
Net income	\$24,457	\$27,093	\$60,967	\$63,934
Foreign currency translation adjustments	6,306	8,425	12,040	13,736
Total comprehensive income	\$30,763	\$35,518	\$73,007	\$77,670

5. Inventories

Inventories are stated at the lower of cost or market. The Company uses the last-in, first-out (LIFO) and first-in, first-out (FIFO) cost methods. An actual valuation of inventory under the LIFO method is made only at the end of the fiscal year based on the inventory levels and costs at that time. Accordingly, interim LIFO calculations are based on management's estimates of expected year-end inventory levels and are subject to the final year-end LIFO inventory valuation. Inventory costs include material, labor, and overhead. Inventories consisted of the following:

	December 31, 2004	March 31, 2004
Raw materials	\$ 28,447	\$ 27,916
Work in process	20,264	24,420
Finished goods	54,935	45,913
Total inventories	\$ 103,646	\$ 98,249

6. Income Taxes

Income tax expense includes U.S. federal, state and local, and foreign income taxes, and is based on reported pre-tax income. Income tax expense is provided on an interim basis based upon the Company's estimate of the annual effective income tax rate. In determining the estimated annual effective income tax rate, the Company analyzes various factors, including projections of the Company's annual earnings and taxing jurisdictions in which the earnings will be generated, the impact of state and local income

taxes, the Company's ability to use tax credits and net operating loss carryforwards, and available tax planning alternatives. The effective income tax rates for the three month periods ended December 31, 2004 and 2003 were 37.7% and 27.5%, respectively. For the nine month periods ended December 31, 2004 and 2003, the effective income tax rates were 37.0% and 32.7%, respectively. The higher effective income tax rates in both fiscal 2005 periods resulted primarily from a reduction of operating profits generated in international tax jurisdictions and the resulting inability of the Company to fully utilize a portion of foreign tax credits against foreign profits taxed in the United States.

7. Benefit Plans

The Company provides defined benefit pension plans for certain manufacturing and plant administrative personnel throughout the world as determined by collective bargaining agreements or employee benefit standards set at the time of acquisition of certain businesses. In addition to providing pension benefits to certain employees, the Company sponsors an unfunded post-retirement medical benefit plan for two groups of U.S. employees comprised substantially of the same employees who receive pension benefits under the U.S. defined benefit plans. Benefits under this plan include retiree life insurance and retiree medical insurance, including prescription drug coverage and Medicare supplemental coverage. Additional information regarding the Company's defined benefit pension plans and post-retirement medical benefit plan, including detailed information regarding employees covered and assumptions used to determine the benefit obligation and net periodic benefit cost is included in Note 9, "Benefit Plans," to the Company's consolidated financial statements included in the Company's Annual Report on Form 10-K filed with the Securities and Exchange Commission on June 14, 2004.

The components of net periodic benefit cost for the Company's defined benefit pension plans and other post-retirement benefit plan were as follows:

		Defined Benefit Pension Plans Othe						
Three-Months Ended December 31,		2004		2003		2004		2003
Service cost	\$	377	\$	374	\$	235	\$	235
Interest cost		793		776		1,172		1,158
Expected return on plan assets		(822)		(720)				
Net amortization and deferral		144		256		476		422
Net periodic benefit cost	\$	492	\$	686	\$	1,883	\$	1,815
	_	Defined Benefi	it Pension			Other Post-R		
Nine-Months Ended December 31,		2004		2003		2004		2003
Service cost	\$	1,131	\$	1,122	\$	705	\$	705
Interest cost		2,379		2,328		3,516		3,474
Expected return on plan assets		(2,466)		(2,160)				
Net amortization and deferral		432		768		1,428		1,266
								1,200
Net periodic benefit cost	\$	1,476	\$	2,058	\$	5,649	\$	5,445

8. Contingencies

The Company is involved in various patent, product liability, consumer, commercial, environmental, tax proceedings and claims, government investigations, and other legal and regulatory proceedings and claims that arise from time to time in the ordinary course of business. In accordance with Statement of Financial Accounting Standards No. 5, "Accounting for Contingencies," the Company records accruals for such contingencies to the extent that the Company concludes that their occurrence is both probable and estimable. The Company considers many factors in making these assessments, including the professional judgment of experienced members of management and the Company's legal counsel. The Company has made estimates as to the likelihood of unfavorable outcomes and the amounts of such potential losses. In the opinion of management, the ultimate outcome of these proceedings

and claims is not anticipated to have a material adverse effect on the Company's consolidated financial position, results of operations, or cash flows. Litigation is inherently unpredictable and actual results could materially differ from the Company's estimates. The Company records anticipated recoveries under applicable insurance contracts when assured of recovery.

To the extent that management of the Company believes it is probable that a taxing authority will take a sustainable position on a matter contrary to the position taken by the Company, the Company provides tax accruals. If the Company does not prevail in matters for which accruals have been established or is required to pay amounts in excess of established accruals, the Company's effective income tax rate in a given financial statement period may be materially impacted.

Refer to Part II, Item 1, "Legal Proceedings," on page 31 of this Form 10-Q.

9. Business Segment Information

The Company operates and reports in three business segments: Healthcare, Life Sciences, and STERIS Isomedix Services. Note 12, "Business Segment Information," to the Company's consolidated financial statements included in the Company's Annual Report on Form 10-K filed with the Securities and Exchange Commission on June 14, 2004 provides detailed information regarding each business segment. Operating income (loss) for each business segment reflects the full allocation of all distribution, corporate, and research and development expenses to the reporting segment. The accounting policies for reportable business segments are the same as those for the consolidated Company. Financial information for each of the Company's reportable business segments is presented in the following table:

		nths Ended ıber 31,	Nine Mon Decem	
	2004	2003	2004	2003
Revenues:				
Healthcare	\$204,942	\$188,583	\$568,657	\$547,824
Life Sciences	54,522	63,917	161,769	179,217
STERIS Isomedix Services	24,925	21,786	73,606	63,914
Total revenues	\$284,389	\$274,286	\$804,032	\$790,955
Operating income (loss):				
Healthcare	\$ 38,067	\$ 31,364	\$ 94,516	\$ 83,863
Life Sciences	(3,112)	3,255	(10,575)	3,127
STERIS Isomedix Services	4,957	3,207	14,853	9,384
Total operating income	\$ 39,912	\$ 37,826	\$ 98,794	\$ 96,374

Financial information for each of the Company's U.S. and International geographic areas is presented in the following table. Revenues are based on the location of these operations and their customers. Long-lived assets are those assets that are identified within the operations in each geographic area, including property, plant, equipment, goodwill, intangibles, and other assets.

STERIS CORPORATION NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited) For the Three and Nine Months Ended December 31, 2004 and 2003

(dollars in thousands, except per share amounts)

		Three Months Ended December 31,		ths Ended ber 31,
	2004	2003	2004	2003
Revenues:				
United States	\$ 219,533	\$213,585	\$ 633,894	\$ 616,560
International	64,856	60,701	170,138	174,395
			<u> </u>	
Total revenues	\$ 284,389	\$274,286	\$ 804,032	\$ 790,955
	December 3 2004	31, M	arch 31, 2004	

Long-lived assets:		
United States	\$ 531,006 \$	525,980
International (1)	144,496	81,152
Total long-lived assets	\$ 675,502 \$	607,132

(1) Long-lived tangible and intangible assets and goodwill related to the acquisiton of Browne have been included as international assets at December 31, 2004 in the above table. Refer to Note 2, "Business Acquisition," for further information.

10. Repurchases of Common Shares

On July 28, 2004, the Company announced that its Board of Directors had authorized the repurchase of up to 3.0 million STERIS Common Shares. This Common Share repurchase authorization replaced the Common Share repurchase authorization of July 24, 2002. During the first six months of fiscal 2005, the Company repurchased 1,539,100 of its Common Shares for \$33,868, representing an average price of \$22.01 per Common Share. The Company did not repurchase any Common Shares during the third quarter of fiscal 2005. At December 31, 2004, 2,726,000 Common Shares remained authorized for repurchase and 825,146 Common Shares were held in treasury.

11. Financial and Other Guarantees

The Company generally offers a limited one-year parts and labor warranty on its capital equipment. The specific terms and conditions of those warranties vary depending on the product sold and the country where the Company conducts business. The Company provides for the estimated cost of product warranties at the time product revenue is recognized. Amounts due to customers for the Company's future performance under these warranties are recorded as a current liability on the accompanying consolidated balance sheets. Factors that affect the Company's warranty liability include the number and type of installed units, historical and anticipated rates of product failures, and material and service costs per claim. The Company periodically assesses the adequacy of its recorded warranty liabilities and adjusts the amounts as necessary.

Changes to the Company's warranty liability during the first nine months of fiscal 2005 were as follows:

Balance, March 31, 2004	\$ 5,914
Warranties issued during the period	6,295
Settlements made during the period	(6,288)
Balance, December 31, 2004	\$ 5,921

The Company also issues product maintenance contracts to its customers that are accounted for in accordance with the requirements of FASB Technical Bulletin No. 90-1, "Accounting for Separately Priced Extended Warranty and Product Maintenance Contracts." Such contracts generally range in terms from 1 to 5 years and require the Company to maintain and repair the Company's product over the maintenance contract term. Amounts received from customers under these contracts are initially recorded as a liability for deferred service contract revenue on the accompanying consolidated balance sheets.

The liability recorded for deferred service contract revenue was \$10,713 and \$12,342 as of December 31, 2004 and March 31, 2004, respectively. Such deferred service contract revenue is then amortized on a straight-line basis over the contract term and recognized as service revenue on the accompanying consolidated statements of income. The activity related to the liability for deferred service contract revenue has been excluded from the table presented above.

12. Foreign Currency Forward Contracts

The Company has entered into foreign currency forward contracts with notional values of 9.6 million euro and 5.3 million British pounds sterling to reduce its transactional foreign currency exposures. The Company has recorded an accrued loss of \$1.3 million related to these contracts at December 31, 2004, which is included in "Accrued expenses and other" on the accompanying consolidated balance sheets. The contracts are marked to market, with gains and losses recognized in the consolidated statements of income.

13. Subsequent Events

On January 7, 2005, the Company announced that it had completed the previously announced (November 16, 2004) agreement to acquire certain assets of Cosmed Group, Inc. ("Cosmed"), a privately-held contract sterilization service provider with corporate offices located in Jamestown, Rhode Island, for \$73,000. The acquisition was funded through a combination of cash and the Company's existing revolving credit facility. The acquired business will be integrated into the Company's Isomedix Services segment. As a result of this transaction, Isomedix added five ethylene oxide processing facilities to its current network of locations. The Company's financial position, results of operations, and cash flows as of and for the three and nine month periods ended December 31, 2004 do not reflect the effects of this transaction. The operating results of Cosmed will be included in the Company's consolidated financial statements from the date of the acquisition.

On January 26, 2005, the Company announced, in its fiscal 2005 third quarter earnings release, that it had completed a detailed analysis of strategic alternatives within its Life Sciences segment designed to reshape the segment's product portfolio and improve profitability. As a first step in this process, the Company announced the sale of its DetachTM product line (automated cleaning systems for comparative medicine) subsequent to the end of the third quarter. The sale of the DetachTM product line did not have a material impact to the Company's financial position, results of operations, or cash flows subsequent to December 31, 2004. In addition, the Company announced that it intends to explore the sale of its lyophilizer (freeze dryer), pure steam generator, and water still product lines, which currently represent approximately 30% of Life Sciences revenues.

Report of Independent Registered Public Accounting Firm

Board of Directors and Shareholders STERIS Corporation

We have reviewed the consolidated balance sheet of STERIS Corporation and subsidiaries as of December 31, 2004, and the related consolidated statements of income for the three and nine month periods ended December 31, 2004 and 2003, and the consolidated statements of cash flows for the nine month periods ended December 31, 2004 and 2003. These financial statements are the responsibility of the Company's management.

We conducted our review in accordance with the standards of the Public Company Accounting Oversight Board (United States). A review of interim financial information consists principally of applying analytical procedures and making inquiries of persons responsible for financial and accounting matters. It is substantially less in scope than an audit conducted in accordance with the standards of the Public Company Accounting Oversight Board, the objective of which is the expression of an opinion regarding the financial statements taken as a whole. Accordingly, we do not express such an opinion.

Based on our review, we are not aware of any material modifications that should be made to the consolidated financial statements referred to above for them to be in conformity with U.S. generally accepted accounting principles.

We have previously audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated balance sheet of STERIS Corporation and subsidiaries as of March 31, 2004 and the related consolidated statements of income, shareholders' equity, and cash flows for the year then ended, not presented herein, and in our report dated June 4, 2004, we expressed an unqualified opinion on those consolidated financial statements. In our opinion, the information set forth in the accompanying consolidated balance sheet as of March 31, 2004, is fairly stated, in all material respects, in relation to the consolidated balance sheet from which it has been derived.

/s/ Ernst & Young LLP

Cleveland, Ohio January 24, 2005

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Financial Measures

In the following sections of Management's Discussion and Analysis of Financial Condition and Results of Operations, the Company, at times, may refer to financial measures that are not required to be presented in the financial statements under U.S. generally accepted accounting principles. The Company has used the following financial measures that are not required to be presented under U.S. generally accepted accounting principles in the context of this report: Backlog, debt to capital, and days sales outstanding. The Company defines these financial measures as follows:

- Backlog- is defined by the Company as the amount of unfilled capital purchase orders at a point in time. At December 31, 2004, September 30, 2004, and December 31, 2003, the Company's backlog amounted to \$140.4 million, \$133.5 million, and \$146.2 million, respectively. Backlog at December 31, 2004 was comprised of \$76.1 million related to the Company's Healthcare segment and \$64.3 million related to the Company's Life Sciences segment. Management uses this figure as a measure to assist in the projection of short-term financial results and inventory requirements.
- Debt to capital- is defined by the Company as total long-term debt divided by the sum of long-term debt and shareholders' equity. At December 31, 2004 and March 31, 2004, the Company's debt to capital amounted to 12.4% and 13.8%, respectively. Management uses this figure as a financial liquidity measure to gauge the Company's ability to borrow, provide strength/protection against creditors, fund growth, develop outside of current business operations, and measure the risk of the Company's financial structure.
- Days sales outstanding- is defined by the Company as the average collection period for sales revenue. It is calculated as net accounts receivable divided by the trailing four quarters revenues, multiplied by 365. At December 31, 2004 and March 31, 2004, the Company's days sales outstanding amounted to 83 days and 85 days, respectively. Management uses this figure to help gauge the quality of accounts receivable.

The Company has presented these financial measures because it believes that meaningful analysis of the Company's financial performance is enhanced by an understanding of certain additional factors underlying that performance and management's judgments about those particular factors. These financial measures should not be considered an alternative to measurements required by GAAP. The Company's calculation of these measures may differ from similar measures used by other companies and investors should be careful when comparing the these financial measures to those of other companies.

Executive Overview

Revenues for the third quarter of fiscal 2005 were \$284.4 million, an increase of \$10.1 million, or 3.7%, as compared to \$274.3 million during the prior year third quarter. For the first nine months of fiscal 2005, revenues were \$804.0 million as compared to \$791.0 million for the same period of fiscal 2004, representing an increase of \$13.1 million, or 1.7%. Increased revenues in the Company's Healthcare and Isomedix Services segments were the drivers of overall revenue growth during both periods. These factors were partially offset by a decline in revenues in the Company's Life Sciences segment during both periods. Additional discussion of segment revenues for the three and nine month periods is presented in subsequent sections of Management's Discussion and Analysis of Financial Condition and results of Operations titled, "Revenues" and "Business Segment Results of Operations."

The Company's gross margin percentage was relatively flat for both fiscal 2005 periods presented as compared to the respective fiscal 2004 periods. Strong gross margin growth in the Company's Isomedix Services segment offset continued margin erosion within the Company's Life Sciences segment for both periods presented. Healthcare margins decreased slightly quarter over quarter due to changes in revenue mix and remained flat for the 2005 year to date period as compared to the same prior year period. The addition of the Albert Browne Limited ("Browne") product line during the fiscal 2005 third quarter benefited the Healthcare segment's gross margins. Additional discussion of the Company's gross margin is presented in a subsequent section of Management's Discussion and Analysis of Financial Condition and Results of Operations titled, "Gross Profit."

The Company's net income was \$24.5 million, or \$0.35 per diluted share, for the third quarter of fiscal 2005, representing a decrease of 9.7% from the comparable prior year period's net income of \$27.1 million, or \$0.38 per diluted share. For the first nine months of fiscal 2005, net income was \$61.0 million, or \$0.87 per diluted share, as compared to \$64.0 million, or \$0.91 per diluted share, representing a decrease of \$3.0 million, or 4.6%. The decline in net income for both periods reflects significantly higher effective income tax rates during fiscal 2005, which offset operating income increases of 5.5% and 2.5% for the three and nine month periods ended December 31, 2004, respectively. Information related to the effective income tax rate changes for both periods is discussed in a subsequent section of Management's Discussion and Analysis of Financial Condition and Results of Operations titled, "Income Tax Expense."

The Company continues to focus its capital spending on investments in new and existing facilities, business expansion projects, and information technology improvements. For the first nine months of fiscal 2005, capital expenditures amounted to \$38.8 million, of which \$13.3 million related to third quarter spending. The Company also continued to step up its investment in research and development to strengthen its competitive position in global marketplaces. The increase in research and development expenditures during the first nine months of fiscal 2005 reflects an increased emphasis on new product development, product improvements, and the development of new technological innovations.

On September 15, 2004, the Company announced that a subsidiary had acquired Browne, a manufacturer of chemical indicators based in Leicester, England. The acquired business, which has been integrated into the Company's Healthcare segment, provides the Company with an established European distribution channel and expands its offering of consumable products used with its broad line of capital equipment for infection control and decontamination. For the quarter ended December 31, 2004, revenues resulting from the Browne acquisition amounted to 1.4% of total Company revenues and contributed 2.3% to the Company's overall gross margins.

On January 7, 2005, the Company announced that it had completed the previously announced (November 16, 2004) acquisition of certain assets of Cosmed Group, Inc. ("Cosmed"), a contract sterilization service provider with corporate offices located in Jamestown, Rhode Island. This acquired business will be integrated into the Company's Isomedix Services segment. As a result of this transaction, Isomedix Services added five ethylene oxide processing facilities to its current network of locations.

On January 26, 2005, the Company announced, in its fiscal 2005 third quarter earnings release, that it had completed a detailed analysis of its customers' needs in the Life Sciences segment and identified several steps to reshape the segment's product portfolio and improve profitability. As a first step of this strategy, the Company announced the sale of its DetachTM product line (automated cleaning systems for comparative medicine) subsequent to the end of the third quarter. The sale of the DetachTM product line did not have a material impact to the Company's financial position, results of operations, or cash flows subsequent to December 31, 2004. In addition, the Company announced that it is exploring the sale of its lyophilizer (freeze dryer), pure steam generator, and water still product lines, which account for approximately 30% of Life Sciences segment revenues. These strategic steps will enable the Company to dedicate more management resources to further develop its core sterilization, washing, and decontamination product offerings to the pharmaceutical, biopharmaceutical, and research markets. Assuming all of these steps are executed during fiscal 2006, the Company anticipates a reduction in pre-tax earnings of approximately \$2.0 to \$3.0 million, excluding any gain or loss from asset sales. In the first full fiscal year following completion, the Company expects these steps to add approximately \$3.0 million to \$5.0 million to annual pre-tax earnings.

Results of Operations

In the following subsections, the Company has presented its fiscal 2005 third quarter and fiscal 2005 year to date results of operations, and has compared these results to the corresponding periods of the prior fiscal year.

Revenues

As prescribed by Regulation S-X, the Company has presented separately on its Consolidated Statements of Income for each period presented, revenues generated as either product revenues or service revenues. In discussing revenues, the Company, at times, may refer to revenues in differing detail than that which is prescribed by Regulation S-X. The terminology, definitions, and applications of terms that the Company employs to describe revenues may differ significantly from terms used by other companies. The Company uses the following terms to describe revenues:

- <u>Revenues</u>- The Company's revenues are presented net of sales returns and allowances.
- <u>Product Revenues</u>- Product revenues are defined by the Company as revenues generated from sales of capital equipment, which includes steam and low temperature liquid sterilizers, washing systems, freeze dryers, VHP technology, water stills, and pure steam generators; surgical lights and tables; and the consumable family of products, which includes STERIS SYSTEM 1[®] consumables, sterility assurance products, skin care products, and cleaning consumables.
- <u>Service Revenues</u>- Service revenues are defined by the Company as revenues generated from parts and labor associated with the maintenance, repair, and installation of the Company's capital equipment, as well as revenues generated from contract sterilization offered through the Company's Isomedix Services segment.
- <u>Capital Revenues</u>- Capital revenues are defined by the Company as revenues generated from sales of capital equipment, which includes steam and low temperature liquid sterilizers, washing systems, freeze dryers, VHP technology, water stills, and pure steam generators; and surgical lights and tables.
- <u>Consumable Revenues</u>- Consumable revenues are defined by the Company as revenues generated from sales of the consumable family of products which includes STERIS SYSTEM 1[®] consumables, sterility assurance products, skin care products, and cleaning consumables.
- <u>*Recurring Revenues*</u>- Recurring revenues are defined by the Company as revenues generated from sales of consumable products and service revenues.

The following table illustrates the changes in the Company's revenues for the three and nine month periods ended December 31, 2004 as compared to the same periods in the prior year:

(dollars in thousands)

		nths Ended ber 31,			Percent of Tota	al Revenues
	2004	2003	Change	Percent Change	2004(1)	2003(1)
Capital Revenues	\$132,231	\$133,549	\$ (1,318)	-1.0%	46.5%	48.7%
Consumable Revenues	60,880	57,813	3,067	5.3%	21.4%	21.1%
Product Revenues	193,111	191,362	1,749	0.9%	67.9%	69.8%
Service Revenues	91,278	82,924	8,354	10.1%	32.1%	30.2%
Total Revenues	\$284,389	\$274,286	\$ 10,103	3.7%	100.0%	100.0%
Service Revenues	\$ 91,278	\$ 82,924	\$ 8,354	10.1%	32.1%	30.2%
Consumable Revenues	60,880	57,813	3,067	5.3%	21.4%	21.1%
Recurring Revenues	152,158	140,737	11,421	8.1%	53.5%	51.3%
Capital Revenues	132,231	133,549	(1,318)	-1.0%	46.5%	48.7%
Total Revenues	\$284,389	\$274,286	\$ 10,103	3.7%	100.0%	100.0%
United States	\$219,533	\$213,585	\$ 5,948	2.8%	77.2%	77.9%
International	64,856	60,701	4,155	6.8%	22.8%	22.1%
Total Revenues	\$284,389	\$274,286	\$ 10,103	3.7%	100.0%	100.0%

	Nine Mon Decem				Percent of Tota	al Revenues
	2004	2003	Change	Percent Change	2004(1)	2003(1)
Capital Revenues	\$362,714	\$379,844	\$(17,130)	-4.5%	45.1%	48.0%
Consumable Revenues	175,434	169,637	5,797	3.4%	21.8%	21.5%
Product Revenues	538,148	549,481	(11,333)	-2.1%	66.9%	69.5%
Service Revenues	265,884	241,474	24,410	10.1%	33.1%	30.5%
Total Revenues	\$804,032	\$790,955	\$ 13,077	1.7%	100.0%	100.0%
Service Revenues	\$265,884	\$241,474	\$ 24,410	10.1%	33.1%	30.5%
Consumable Revenues	175,434	169,637	5,797	3.4%	21.8%	21.5%
Recurring Revenues	441,318	411,111	30,207	7.3%	54.9%	52.0%
Capital Revenues	362,714	379,844	(17,130)	-4.5%	45.1%	48.0%
Total Revenues	\$804,032	\$790,955	\$ 13,077	1.7%	100.0%	100.0%
United States	\$633,894	\$616,560	\$ 17,334	2.8%	78.8%	78.0%
International	170,138	174,395	(4,257)	-2.4%	21.2%	22.0%
Total Revenues	\$804,032	\$790,955	\$ 13,077	1.7%	100.0%	100.0%

(1) Certain percentages may not calculate precisely due to rounding.

Quarter over Quarter Comparison

Revenues for the third quarter of fiscal 2005 were \$284.4 million, an increase of \$10.1 million, or 3.7%, as compared to \$274.3 million during the third quarter of the prior year. Improved overall revenues, quarter over quarter, were driven by an increase of 8.1% in recurring revenues. A 14.4% increase in service revenues within the Company's Isomedix Services segment, resulting from increased demand, higher utilization of recently expanded facilities capacity, and a temporary reduction in industry processing capacity, contributed to the overall increase in consumable revenues. Double digit growth of capital revenues and improvements in recurring revenue streams within the Healthcare segment, which resulted from strengthening demand in the United States hospital market, also contributed to the increase in overall Company revenues. Overall revenues for the quarter were negatively impacted by a 14.7% decline in Life Sciences revenues, primarily the result of continued declines in overall capital spending within the pharmaceutical industry in the European and United States marketplaces.



International revenues for the quarter amounted to \$64.9 million, representing an increase of \$4.2 million, or 6.8%, as compared to the third quarter of fiscal 2004. International revenues were positively impacted by an increase in recurring revenues of 37.4% quarter over quarter. Contributing to international recurring revenue growth was an increase of 78.1% in consumables revenues, primarily the result of additional revenues generated from the integration of the Browne product line. International revenues were negatively impacted by a decrease in capital revenues of 7.1% quarter over quarter, primarily the result of weaknesses in the Life Sciences segment throughout Europe.

United States revenues for the quarter amounted to \$219.5 million, representing an increase of \$5.9 million, or 2.8%, as compared to the third quarter of fiscal 2004. United States revenues were positively impacted by an increase in service revenues of 9.0%, which resulted from increased demand within the Company's Healthcare segment and increased volumes and capacity utilization within the Company's Isomedix Services segment. United States revenues were also positively impacted during the quarter by a net increase in capital revenues of 1.8%, which represents a 16.1% increase in United States Healthcare capital revenues resulting from increased demand from the United States hospital market. The United States Healthcare capital revenue growth was offset by a decrease of 40.7% in United States Life Sciences capital revenue.

Revenues are further discussed on a segment basis in the section of Management's Discussion and Analysis of Financial Condition and Results of Operations, titled "Business Segment Results of Operations."

Year over Year Comparison

Revenues for the first nine months of fiscal 2005 were \$804.0 million, an increase of \$13.1 million, or 1.7%, as compared to \$791.0 million during the first nine months of fiscal 2004. For the year to date period, a 10.1% increase in service revenues and a 3.4% increase in consumable revenues resulted in an increase in overall recurring revenues of 7.3% as compared to the same period in the prior year. Service revenues, which increased across all segments, were driven by a \$9.7 million, or 15.2%, increase in the Isomedix Services segment. Within the Company's Healthcare and Life Sciences segments, service revenues for the year to date period increased 5.4% and 28.8%, respectively, as compared to the same prior year period. Capital revenues declined \$17.1 million, or 4.5%, as compared to the first nine months of fiscal 2004. Strong Healthcare capital sales within the United States hospital market during the fiscal 2005 third quarter resulted in a 4.1% increase in the segment's year to date capital revenues as compared to the same period in fiscal 2004. This performance was offset by a 20.7% decrease in Life Sciences capital revenues, year over year, as a result of reduced capital spending within the pharmaceutical industry in the European and United States marketplaces.

International revenues for the first nine months of fiscal 2005 amounted to \$170.1 million, a decrease of \$4.3 million, or 2.4%, as compared to the first nine months of fiscal 2004. The decline in year over year international revenues was attributable to a 15.7% decline in capital equipment sales within the Company's Healthcare and Life Sciences segments, primarily within the European marketplace. Within Europe, year to date capital revenues within the Company's Healthcare and Life Sciences segments decreased 13.4% and 32.0%, respectively, as compared to the same prior year period. The decline in international capital equipment sales was partially offset by a 27.8% increase in international recurring revenue streams year over year.

United States revenues for the first nine months of fiscal 2005 amounted to \$633.9 million, an increase of \$17.3 million, or 2.8%, as compared to the first nine months of fiscal 2004. United States revenues were positively impacted by an increase of 4.3% in recurring revenues, driven by an increase of 15.2% in service revenues within the Company's Isomedix Services segment. Increased volumes within the Company's Healthcare segment also contributed to the year over year increase in consumable revenues. Year over year, United States capital revenues remained relatively flat. A surge in Healthcare capital sales within the United States hospital marketplace during the latter part of the second quarter and throughout the third quarter of fiscal 2005 resulted in a 7.5% increase in capital revenues in the segment, year over year. A year over year decrease in United States capital revenues of 23.1% within the Company's Life Sciences segment, as a result of reduced capital spending within the pharmaceutical industry, offset the Healthcare segment's performance.

Revenues are further discussed on a segment basis in the section of Management's Discussion and Analysis of Financial Condition and Results of Operations, titled "Business Segment Results of Operations."

Gross Profit

The following table illustrates the changes in gross profit for the three and nine month periods ended December 31, 2004 and 2003:

(dollars in thousands)

	Three Mon Decemb			
	2004	2003	Change	Percent Change
Gross Profit:				
Product	\$ 80,888	\$ 79,242	\$ 1,646	2.1%
Service	38,674	35,855	2,819	7.9%
Total Gross Profit	\$ 119,562	\$115,097	\$ 4,465	3.9%
Gross Profit Percentage:				
Product	41.9%	41.4%		
Service	42.4%	43.2%		
Total Gross Profit Percentage	42.0%	42.0%		
-				

	Nine Mont Decemb			
	2004	2003	Change	Percent Change
Gross Profit:				
Product	\$228,143	\$229,474	\$(1,331)	-0.6%
Service	112,799	102,808	9,991	9.7%
Total Gross Profit	\$340,942	\$332,282	\$ 8,660	2.6%
Gross Profit Percentage:				
Product	42.4%	41.8%		
Service	42.4%	42.6%		
Total Gross Profit Percentage	42.4%	42.0%		

Gross profit is impacted by the volume, pricing, and mix of sales of the Company's products and services, as well as the costs associated with the products and services that are sold. The Company's gross profit percentage was flat quarter over quarter and increased 40 basis points year over year. Strong margin growth, resulting from higher volumes, within the Company's Isomedix Services segment offset continued margin erosion within the Company's Life Sciences segment for both periods presented. Healthcare margins were negatively impacted during fiscal 2005 as a result of a change in revenue mix. However, the addition of the Albert Browne Limited ("Browne") product line during the 2005 third quarter benefited the Healthcare segment's gross margins. Gross margins for both periods presented were positively impacted by changes in foreign currency exchange rates in relation to the U.S. dollar. The Company's gross profit percentage for the nine month period ending December 31, 2003 was negatively impacted by costs associated with the closing of an Australian manufacturing facility.

Operating Expenses

The following table illustrates the changes in operating expenses for the three and nine month periods ended December 31, 2004 and 2003:

(dollars in thousands)

	Three Months December			
	2004	2003	Change	Percent Change
Operating Expenses:				
Selling, General, and Administrative	\$ 70,465 \$	69,623	\$ 842	1.2%
Research and Development	9,185	7,648	1,537	20.1%
Total Operating Expenses	\$ 79,650 \$	5 77,271	\$2,379	3.1%
	Nine Months December			
			Change	Percent Change
Operating Expenses:	December	r 31,	Change	
Operating Expenses: Selling, General, and Administrative		r 31,	Change \$ 793	
		2003		Change
Selling, General, and Administrative	2004 \$214,975 \$	2003 5214,182	\$ 793	Change 0.4%
Selling, General, and Administrative	December 2004 \$214,975 \$27,173	2003 5214,182	\$ 793	Change 0.4%

Significant components of total selling, general, and administrative expenses (SG&A) are compensation and associated costs, fees for professional services, travel and entertainment, and other general and administrative expenses. As a percentage of total revenues, SG&A expenses were 24.8% and 26.7% for the three and nine month periods ended December 31, 2004, respectively, as compared to 25.4% and 27.1%, respectively, for the comparable prior year periods. As a percentage of revenues, SG&A expenses decreased in both fiscal 2005 periods. SG&A expenses remained relatively flat for the three and nine month periods ended December 31, 2004, as compared to the same prior year periods.

As a percentage of total revenues, research and development expenses were 3.2% and 3.4% for the three and nine month periods ended December 31, 2004, respectively, as compared to 2.8% and 2.7%, respectively, for the same prior year periods. For the three and nine month periods ended December 31, 2004, research and development expenses increased 20.1% and 25.1% to \$9.2 million and \$27.2 million, respectively, as compared to \$7.6 million and \$21.7 million, respectively, during the same prior year periods. The increase in research and development expenses during both periods is attributable to an increased emphasis on new product development, product improvements, and the development of new technological innovations.

Interest Expense, Net

The following table illustrates the changes in interest expense, net for the three and nine month periods ended December 31, 2004 and 2003: *(dollars in thousands)*

		nths Ended iber 31,	
	2004	2003	Change
Interest Expense, Net:			
Interest Expense	\$ 1,118	\$ 440	\$ 678
Interest and Miscellaneous (Income) Expense	(484)	8	(492)
Interest Expense, Net	\$ 634	\$ 448	\$ 186
		nths Ended aber 31,	
	2004	2003	Change
Interest Expense, Net:			
Interest Expense	\$ 3,008	\$ 1,486	\$1,522
Interest and Miscellaneous (Income) Expense	(988)	(54)	(934)
		·	
		A 1 100	+
Interest Expense, Net	\$ 2,020	\$ 1,432	\$ 588

Interest expense, net consists of interest expense on debt, offset by interest earned on cash, cash equivalents, short-term investment balances, and other miscellaneous income. The most significant driver of the increase in both fiscal 2005 periods is interest expense, which resulted from higher interest rates on the Company's outstanding debt during fiscal 2005 as compared to the prior year. A detailed discussion of the Company's outstanding debt is included in the Company's Annual Report on Form 10-K for the year ended March 31, 2004 filed with the Securities and Exchange Commission on June 14, 2004 in the section of Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations," titled, "Sources of Credit," and in Note 6 to the Company's consolidated financial statements, "Long-Term Debt." The Company's credit facilities have not changed materially from March 31, 2004.



Income Tax Expense

The following table illustrates the changes in income tax expense for the three and nine month periods ended December 31, 2004 and 2003, and provides a comparison of the effective income tax rates for the aforementioned periods:

(dollars in thousands)

	Three Mon Decemi			
	2004	2003	Change	Percent Change
Income Tax Expense	\$14,821	\$10,285	\$4,536	44.1%
Effective Income Tax Rate	37.7%	27.5%		
	Nine Mont Decemi			
	2004	2003	Change	Percent Change
Income Tax Expense	\$35,807	\$31,008	\$4,799	15.5%
Effective Income Tax Rate	37.0%	32.7%		

The Company provides for income tax expense on an interim basis based upon the Company's estimate of the annual effective income tax rate. In determining the estimated annual effective income tax rate, the Company analyzes various factors, including projections of the Company's annual earnings and the taxing jurisdictions in which the earnings will be generated, the impact of state and local income taxes, the Company's ability to use tax credits and net operating loss carryforwards, and available tax planning alternatives. The effective income tax rates for the third quarters of fiscal 2005 and 2004 were 37.7% and 27.5%, respectively. For the first nine months of fiscal 2005 and 2004, the effective income tax rates were 37.0% and 32.7%, respectively. The higher effective income tax rates during both fiscal 2005 periods resulted primarily from a reduction of operating profits generated in international tax jurisdictions and the resulting inability of the Company to fully utilize a portion of foreign tax credits against foreign profits taxed in the United States.

Business Segment Results of Operations

The Company operates and reports in three business segments: Healthcare, Life Sciences, and STERIS Isomedix Services. Note 12, "Business Segment Information," to the Company's consolidated financial statements included in the Company's Annual Report on Form 10-K filed with the Securities and Exchange Commission on June 14, 2004, provides detailed information regarding each business segment. The following table provides a summary of the Company's revenues by business segment for the three and nine month periods ended December 31, 2004 and 2003:

(dollars in thousands)

	Three Mon Decemi			
	2004	2003	Change	Percent Change
Revenues:				
Healthcare	\$204,942	\$188,583	\$ 16,359	8.7%
Life Sciences	54,522	63,917	(9,395)	-14.7%
STERIS Isomedix Services	24,925	21,786	3,139	14.4%
Total Revenues	\$284,389	\$274,286	\$ 10,103	3.7%
	Nine Mon Decem			
			Change	Percent Change
Revenues:	Decem	ber 31,	Change	
Revenues: Healthcare	Decem	ber 31,	Change \$ 20,833	
		ber 31, 2003		Change
	Decemi 2004 \$568,657	\$547,824	\$ 20,833	Change 3.8%
Healthcare Life Sciences	Decemi 2004 \$568,657 161,769	ber 31, 2003 \$547,824 179,217	\$ 20,833 (17,448)	Change 3.8% -9.7%

Healthcare Segment

Healthcare segment revenues represented 72.0% of total revenues for the third quarter of fiscal 2005 as compared to 68.8% of total revenues for the third quarter of fiscal 2004. Healthcare revenues increased \$16.4 million, or 8.7%, to \$204.9 million for the third quarter of fiscal 2005 as compared to \$188.6 million for the third quarter of fiscal 2004. The increase in Healthcare revenues for the quarter was primarily driven by a 12.3% increase in capital revenues, which resulted from increased demand generated in the United States hospital market. At December 31, 2004, the Healthcare segment's backlog amounted to \$76.1 million, representing increases of \$3.7 million, or 5.0%, and \$7.6 million, or 11.2%, over the September 30, 2004 and December 31, 2003 levels, respectively. The Healthcare segment's fiscal 2005 third quarter revenues were also positively impacted by a 4.5% increase in consumable revenues, primarily attributable to increased chemical indicator sales resulting from the business integration of Browne.

Healthcare segment revenues represented 70.7% of total revenues for the first nine months of fiscal 2005 as compared to 69.2% of total revenues for the third quarter of fiscal 2004. Healthcare revenues increased \$20.8 million, or 3.8%, to \$568.7 million for the first nine months of fiscal 2005 as compared to \$547.8 million for the first nine months of fiscal 2004. Healthcare capital revenues increased 4.1% year over year, for the reasons presented in the preceding paragraph. Service revenues within the segment increased 5.4%, year over year, and contributed to the overall increase in segment revenues.

Life Sciences Segment

Life Sciences segment revenues represented 19.2% of total revenues for the third quarter of fiscal 2005 as compared to 23.3% of total revenues for the third quarter of fiscal 2004. Life Sciences revenues decreased \$9.4 million, or 14.7%, to \$54.5 million for the third quarter of fiscal 2005 as compared to \$63.9 million for the third quarter of fiscal 2004. The decrease in Life Sciences revenues for the quarter was driven by a 25.1% decrease in capital revenues. Consolidations within the European pharmaceutical marketplace during the Company's first two fiscal 2005 quarters continued to impact European capital equipment sales. The quarter over quarter decline in capital revenues was further impacted by more prevalent cautiousness by pharmaceutical customers in the North American marketplace regarding capital spending. At December 31, 2004, backlog for the segment was \$64.3 million, compared to backlogs of \$61.1 million at September 30, 2004, and \$77.8 million at December 31, 2003. An increase of 15.5% in recurring revenue streams partially offset the segment's quarter over quarter decline in capital revenues.

Life Sciences segment revenues represented 20.1% of total revenues for the first nine months of fiscal 2005 as compared to 22.7% of total revenues for the first nine months of fiscal 2004. Life Sciences revenues decreased \$17.4 million, or 9.7%, to \$161.8 million for the first nine months of fiscal 2005 as compared to \$179.2 million for the first nine months of fiscal 2004. The decrease in Life Sciences revenues for the year to date period was driven by a 20.7% decline in capital revenues, the reasons for which were presented in the previous paragraph. An increase of 20.6% in recurring revenue streams partially offset the segment's year over year decline in capital revenues.

STERIS Isomedix Services Segment

STERIS Isomedix Services segment revenues represented 8.8% and 9.2% of total revenues for the three and nine month periods ended December 31, 2004 as compared to 7.9% and 8.1% during the comparable prior year periods. The segment experienced revenue growth of 14.4% and 15.2%, respectively, for the three and nine month periods ended December 31, 2004, as compared to the same periods in the prior year. The revenue growth in both periods presented is the result of increased demand, higher utilization of recently expanded facilities in the segment, and a temporary reduction in industry processing capacity.

The following table provides a summary of the Company's operating results by business segment for the three and nine month periods ended December 31, 2004 and 2003:

(dollars in thousands)

		Three Months Ended December 31,		
	2004	2003	Change	Percent Change
Operating Income (Loss):				
Healthcare	\$ 38,067	\$31,364	\$ 6,703	21.4%
Life Sciences	(3,112)	3,255	(6,367)	NM
STERIS Isomedix Services	4,957	3,207	1,750	54.6%
	<u> </u>		·	
Total Operating Income	\$ 39,912	\$37,826	\$ 2,086	5.5%
	Nine Montl Decemb			
			Change	Percent Change
Operating Income (Loss):	Decemb	er 31,	Change	
Operating Income (Loss): Healthcare	Decemb	er 31,	<u>Change</u> \$ 10,653	
		er 31, 2003		Change
Healthcare	Decemb 2004 \$ 94,516	er 31, 2003 \$83,863	\$ 10,653	Change
Healthcare Life Sciences	Decemb 2004 \$ 94,516 (10,575)	er 31, 2003 \$83,863 3,127	\$ 10,653 (13,702)	Change 12.7% NM

NM- Not meaningful

To determine segment operating income (loss), the Company reduces the respective segment's revenues by direct expenses and indirect cost allocations, which reflect the full allocation of all distribution, corporate, and research and development expenses. Corporate cost allocations are based on each segment's portion of revenues, headcount, or other variables in relation to the total Company.

Healthcare Segment

Healthcare- Operating income increased \$6.7 million, or 21.4%, to \$38.1 million for the third quarter of fiscal 2005, as compared to \$31.4 million during the comparable prior year period. As a percentage of segment revenues, operating income represented 18.6% and 16.6% for the quarters ended December 31, 2004 and 2003, respectively, reflecting lower operating expenses as a percentage of revenue. Healthcare gross margins were 47.3% for the third quarter of fiscal 2005 as compared to 47.7% during the same prior year period, and were impacted by a continued shift in revenue mix toward capital equipment sales which carry lower margins. For the 2005 year to date period, the segment's operating income was \$94.5 million, representing an increase of \$10.7 million, or 12.7%, as compared to the first nine months of fiscal 2004. For the year to date periods of fiscal 2005 and 2004, gross margins remained relatively flat at 47.7% and 47.6%, respectively.

Life Sciences Segment

Life Sciences- Operating loss was \$3.1 million and \$10.6 million for the third quarter and first nine months of fiscal 2005 as compared to operating income of \$3.3 million and \$3.1 million during the respective prior year periods. For the third quarter and year to date period of fiscal 2005, gross margins were 29.3% and 30.7%, as compared to 32.9% and 32.2% during the comparable prior year periods. Both fiscal 2005 periods were negatively impacted by reduced sales volumes and the resulting lower fixed cost absorption in the segment.

STERIS Isomedix Services Segment

STERIS Isomedix Services- Operating income increased \$1.8 million, or 54.6%, to \$5.0 million for the third quarter of fiscal 2005, as compared to \$3.2 million during the third quarter of fiscal 2004. For the nine month period ended December 31, 2004, the segment's operating income was \$14.9 million, representing an increase of \$5.5 million, or 58.3%, over the comparable prior year period's operating income of \$9.4 million. The drivers of the increase in both periods were more favorable gross margins. Gross margins benefited from increased volume and improvements in processing utilization as a result of capital investments made during the past year.

Liquidity and Capital Resources

The following table summarizes significant components of the Company's cash flow for the nine months ended December 31, 2004 and 2003:

Cash Flows

(dollars in thousands)

	Nine Months Ended December 31,			
	2004	2003	Change	Percent Change
Operating activities:				
Net income	\$ 60,967	\$ 63,934	\$ (2,967)	-4.6%
Non-cash items	39,853	43,514	(3,661)	-8.4%
Changes in operating assets and liabilities, excluding the effects of business acquisitions	(1,763)	(40,233)	38,470	95.6%
Net cash provided by operating activities	\$ 99,057	\$ 67,215	\$ 31,842	47.4%
Investing activities:				
Purchases of property, plant, equipment, and intangibles	\$(38,793)	\$(48,430)	\$ 9,637	19.9%
Purchase of business related assets	_	(2,900)	2,900	NM
Investments in businesses, net of cash acquired	(53,323)	(36,814)	(16,509)	-44.8%
Net cash used in investing activities	\$(92,116)	\$(88,144)	\$ (3,972)	-4.5%
Financing activities:				
Net (payments) proceeds under credit facilities	\$ (4,398)	\$ 46,800	\$(51,198)	NM
Payments of long-term obligations and capital leases	(3,273)	(4,933)	1,660	33.7%
Repurchases of Common Shares	(33,868)	(16,609)	(17,259)	-103.9%
Stock option and other equity transactions, net	14,581	7,748	6,833	88.2%
Deferred financing fees and debt issuance costs		(537)	537	NM
Net cash (used in) provided by financing activities	\$(26,958)	\$ 32,469	\$(59,427)	-183.0%

NM- Not meaningful

Net Cash Provided by Operating Activities

Net cash provided by operating activities was \$99.1 million for the first nine months of fiscal 2005 as compared to \$67.2 million for the same prior year period. Non-cash items include depreciation and amortization and fluctuations in deferred income taxes and other items. Deferred income tax liabilities increased during the first nine months of fiscal 2005 due to increases in deferred income taxes attributable to long-term customer contracts and utilization of net operating loss carryforwards. Significant components of the Company's operating cash flows for the nine months ended December 31, 2004 and 2003 driving the increase in net cash inflow from operating activities of \$31.8 million include:

- Accounts receivables, net-Excluding the impact of foreign currency translation adjustments and businesses acquired, accounts receivable, net decreased \$10.5 million and \$1.8 million during the first nine months of fiscal 2005 and 2004, respectively. Accounts receivable balances are influenced by the timing of revenues, customer payments, and progress billings for contracts which are accounted for under the percentage of completion method of accounting for construction-type contracts. At December 31, 2004 and March 31, 2004, accounts receivables represent 83 days and 85 days outstanding, respectively.
- Inventories- Excluding the impact of foreign currency translation adjustments and businesses acquired, inventories decreased \$3.3 million during the first nine months of fiscal 2005, as compared to an increase of \$0.7 million during the same period of the prior year. The Company has established targeted inventory production levels at manufacturing facilities in a process called modified level-loading, whereby a relatively constant stream of inventory production occurs, which may result in varying inventory levels during the year as a result of customer demand fluctuations.

- Accounts payable, net- Excluding the impact of foreign currency translation adjustments and businesses acquired, accounts payable decreased \$19.4 million and \$30.9 million during the first nine months of fiscal 2005 and 2004, respectively. Based upon varying payment due dates of accounts payable obligations and the Company's cash management strategies, accounts payable balances may fluctuate from period to period.
- Accruals and other, net- Excluding the impact of foreign currency translation adjustments and businesses acquired, accruals and other, net decreased \$3.2 million and increased \$11.3 million during the first nine months of fiscal 2005 and 2004, respectively. Accruals and other, net decreased during the first nine months of fiscal 2005 due to decreased accrued payroll and related liabilities that resulted from the timing of compensation related payments, reduced projected bonus payouts under the Company's profit sharing plan, and reduced 401(k) obligations due to the timing of their funding. The decrease in accruals and other, net was further impacted by reduced levels of accrued employee and dealer commissions and deferred revenues.

Net Cash Used In Investing Activities

For the first nine months of fiscal 2005, net cash used in investing activities amounted to \$92.1 million as compared to \$88.1 million during the same prior year period. The following discussion summarizes the significant components of the Company's investing cash flows for the nine months ended December 31, 2004 and 2003:

- Purchases of property, plant, equipment, and intangibles- During the first nine months of fiscal 2005, the Company's capital expenditures amounted to \$38.8 million as compared to \$48.4 million during the comparable prior year period. The decrease in capital spending year over year resulted from a reduction of in process corporate facilities expansion projects during fiscal 2005 as compared to the prior year. In addition, certain information technology initiatives were completed during the first half of fiscal 2005, thus reducing the level of capital expenditures during the current year. Additionally, during fiscal 2004, as a result of Isomedix Services facilities expansions, incremental cobalt requirements were funded by capital expenditures.
- Investments in businesses, net of cash acquired- During the second quarter of fiscal 2005, the Company announced that a subsidiary had acquired Albert Browne Limited ("Browne"). Investment in this business, including direct acquisition costs, net of cash acquired, amounted to \$53.3 million. Further discussion of the Company's acquisition of Browne is included in Note 2, "Business Acquisition." Net cash flows used in investing activities for the first nine months of fiscal 2004 reflect the acquisition of Hamo Holdings AG ("Hamo") of \$36.8 million. The Company completed this acquisition during the first quarter of fiscal 2004.
- Purchases of business related assets- During the first quarter of fiscal 2004, the Company acquired certain assets related to the sterilization container business from Sterion Incorporated for \$2.9 million. The purchase of these assets is presented as cash used in investing activities for the nine month period ended December 31, 2003.

Net Cash (Used In) Provided By Financing Activities

For the first nine months of fiscal 2005, net cash used in financing activities amounted to \$27.0 million as compared to net cash provided by financing activities of \$32.5 million during the same prior year period. The following discussion summarizes the significant components of the Company's financing cash flows for the nine months ended December 31, 2004 and 2003:

Net (payments) proceeds under credit facilities- During the first nine months of fiscal 2005, net payments under credit facilities amounted to \$4.4 million. During the first nine months of fiscal 2004, net proceeds under the Company's credit facilities amounted to \$46.8 million and were partially used by the Company to fund the acquisition of Hamo. A detailed description of the Company's sources of credit is included in the Company's Annual Report on Form 10-K for the year ended March 31, 2004 filed with the Securities and Exchange Commission on June 14, 2004 in the section of Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations," titled, "Sources of Credit," and in Note 6, "Long-Term Debt," to the consolidated financial statements. There have been no significant changes in the Company's sources of credit since March 31, 2004.

- Repurchases of Common Shares- As discussed in Note 10, "Repurchases of Common Shares," the Company's Board of Directors has authorized the periodic repurchase of the Company's Common Shares. From time to time, the Company engages in open market transactions to repurchase its Common Shares. During the first nine months of fiscal 2005, the Company repurchased 1.5 million of its Common Shares for \$33.9 million, representing an average purchase price of \$22.01 per Common Share. During the first nine months of fiscal 2004, the Company repurchased 0.8 million of its Common Shares for \$16.6 million, representing an average purchase price of \$21.82 per Common Share.
- Stock option and other equity transactions- Cash flows from stock option and other equity transactions are primarily derived from the issuance of the Company's Common Shares under various employee stock compensation programs. During the first nine months of fiscal 2005, cash proceeds, net of tax benefits, from the issuance of Common Shares under these programs totaled \$4.7 million as compared to \$3.9 million during the same period of fiscal 2004.

Cash Requirements

The Company believes that its available cash, cash flows from operations, and sources of credit will be adequate to satisfy its operating and capital needs for the next twelve months. The Company's management and Board of Directors continually review the operating and capital requirements, as well as the overall capital structure of the Company.

Sources of Credit and Contractual and Commercial Commitments

A discussion of the Company's sources of credit and contractual and commercial commitments is included in Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations," included in the Company's Annual Report on Form 10-K for the year ended March 31, 2004, filed with the Securities and Exchange Commission on June 14, 2004. The Company's sources of credit and contractual and commercial commitments have not changed materially from March 31, 2004. At December 31, 2004 and March 31, 2004, debt to capital (as defined on page 16) amounted to 12.4% and 13.8%, respectively.

Critical Accounting Policies, Estimates, and Assumptions

A discussion of the Company's critical accounting policies, estimates, and assumptions is included in Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations," included in the Company's Annual Report on Form 10-K for the year ended March 31, 2004, filed with the Securities and Exchange Commission on June 14, 2004. The Company's critical accounting policies, estimates, and assumptions have not changed materially from March 31, 2004.

Contingencies

The Company is involved in various patent, product liability, consumer, commercial, environmental, tax proceedings and claims, government investigations, and other legal and regulatory proceedings and claims that arise from time to time in the ordinary course of business. In accordance with Statement of Financial Accounting Standards No. 5, "Accounting for Contingencies," the Company records accruals for such contingencies to the extent that the Company concludes that their occurrence is both probable and estimable. The Company considers many factors in making these assessments, including the professional judgment of experienced members of management and the Company's legal counsel. The Company has made estimates as to the likelihood of unfavorable outcomes and the amounts of such potential losses. In the opinion of management, the ultimate outcome of these proceedings and claims is not anticipated to have a material adverse effect on the Company's consolidated financial position, results of operations, or cash flows. Litigation is inherently unpredictable and actual results could materially differ from the Company's estimates. The Company records anticipated recoveries under applicable insurance contracts when assured of recovery.

To the extent that management of the Company believes it is probable that a taxing authority will take a sustainable position on a matter contrary to the position taken by the Company, the Company provides tax accruals. If the Company does not prevail in matters for which accruals have been established or is required to pay amounts in excess of established accruals, the Company's effective income tax rate in a given financial statement period may be materially impacted.

Refer to Part II, Item 1, "Legal Proceedings," on page 31 of this Form 10-Q.

International Operations

The Company conducts operations outside of the United States through its subsidiaries in the same business segments as the Company's domestic operations. Depending on the direction of change relative to the U.S. dollar, foreign currency values can increase or reduce the reported dollar value of the Company's net assets and results of operations. Revenues were favorably impacted by \$5.4 million, or 1.9%, and net income was favorably impacted by \$1.1 million, or 4.5%, during the third quarter of fiscal 2005 as compared to a favorable impact to revenues and net income of \$6.4 million, or 2.4%, and \$0.6 million, or 2.3%, respectively, during the comparable prior year period. Foreign currency fluctuations favorably impacted revenues during the first nine months of fiscal 2005 by \$10.4 million, or 1.3%, and net income by \$0.7 million, or 1.2%, as compared to a favorable impact to revenues and net income of \$17.2 million, or 2.2%, and \$1.5 million, or 2.4%, respectively, during the comparable prior year period.

Seasonality

The Company's financial results have been subject to recurring seasonal fluctuations. A number of factors have contributed to the seasonal patterns, including sales promotion and compensation programs, customer buying patterns, and international business practices. Sales and profitability of certain of the Company's product lines have historically been disproportionately weighted toward the latter part of each quarter and generally weighted toward the latter part of each fiscal year. However, there is no assurance that these patterns will continue.

Forward-Looking Statements

This document may contain statements and data concerning certain trends, expectations, forecasts, estimates, or other forward-looking information affecting or relating to the Company or its industry that are intended to qualify for the protections afforded "forward-looking statements" under the Private Securities Litigation Reform Act of 1995 and other laws and regulations. Forward-looking statements speak only as to the date of this report and may be identified by the use of forward-looking terms such as "may," "will," "expects," "believes," "anticipates," "plans," "estimates," "projects," "targets," "forecasts," and "seeks," or the negative of such terms or other variations on such terms or comparable terminology. Many important factors could cause actual results to differ materially from those in the forward-looking statements including, without limitation, disruption of production or supplies, changes in market conditions, political events, pending or future claims or litigation, competitive factors, technology advances, and changes in government regulations or the application or interpretation thereof. Other risk factors are described in the Company's Form 10-K and other securities filings. Many of these important factors are outside STERIS's control. No assurances can be provided as to any future financial results. Unless legally required, the Company does not undertake to update or revise any forward-looking statements even if events make clear that any projected results, express or implied, will not be realized. Other potential risks and uncertainties that could cause actual results to differ materially from those in the forward-looking statements include, without limitation, (a) the potential for increased pressure on pricing that leads to erosion of profit margins, (b) the possibility that market demand will not develop for new technologies, products, or applications, or the Company's business initiatives will take longer, cost more, or produce lower benefits than anticipated, (c) the possibility that compliance with laws, court rulings, regulations, or certification requirements of domestic and foreign authorities may delay or prevent new product introductions, affect the production and marketing of existing products, or otherwise affect Company performance, (d) the potential of international unrest or effects of fluctuations in foreign currencies of countries where the Company does a sizeable amount of business, and (e) the possibility of reduced demand, or reductions in the rate of growth in demand, for the Company's products and services, and (f) the ability of the Company to successfully complete and/or realize the anticipated benefits of the proposed divestiture of certain Life Sciences product lines.

Availability of Securities and Exchange Commission Filings

The Company files annual reports on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K, amendments to those reports, and other information with the Securities and Exchange Commission (the "SEC"). Copies of these materials can be obtained by visiting the SEC's Public Reference Room at 450 Fifth Street, NW, Washington, D.C. 20549 or by accessing the SEC's website at <u>http://www.sec.gov</u>. Information may be obtained by calling the SEC at 1-800-SEC-0330. In addition, as soon as reasonably practicable after such materials are filed with or furnished to the SEC, the Company makes copies available to the public, free of charge, on or through the investor relations section of its website at <u>http://www.steris.com</u>. Information on the Company's website is not incorporated by reference into this report.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

A discussion of market risk exposures is included in Part II, Item 7A, "Quantitative and Qualitative Disclosures about Market Risk," of the Company's Annual Report on Form 10-K for the year ended March 31, 2004 filed with the Securities and Exchange Commission on June 14, 2004. The Company's exposures to market risk have not changed materially since March 31, 2004. During the second quarter of fiscal 2005, the Company began to manage its foreign currency risk by entering into forward contracts. These contracts are entered into in order to mitigate the impact of currency fluctuations on transactions and other exposures. The Company does not enter into forward contracts for speculative purposes.

ITEM 4. CONTROLS AND PROCEDURES

The Company's management is responsible for establishing and maintaining effective disclosure controls and procedures, as defined under Rules 13a-15(e) and 15d-15(e) of the Securities and Exchange Act of 1934 (the "Exchange Act"). As of December 31, 2004, an evaluation was performed under the supervision and with the participation of the Company's management, including the Chief Executive Officer ("CEO") and the Chief Financial Officer ("CFO"), of the effectiveness of the Company's disclosure controls and procedures. Based on that evaluation, the Company's management, including the CFO, concluded that the Company's disclosure controls and procedures as of December 31, 2004 were effective in ensuring material information required to be disclosed in this Quarterly Report on Form 10-Q was recorded, processed, summarized, and reported on a timely basis. In addition, there were no changes in the Company's internal control over financial reporting as defined under Exchange Act Rules 13a-15(f) and 15d-15(f) that occurred during the quarter ended December 31, 2004 that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

PART II - OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

The Company is involved in a number of legal proceedings and claims which the Company believes arise from the ordinary course of its business, given its size, history, complexity, nature of its business, and industries in which it participates. These legal proceedings and claims generally involve a variety of legal theories and allegations, including without limitation, personal injury (e.g., slip and falls, automobile accidents), product liability (e.g., based on the operation or claimed malfunction of products), product exposure (e.g., claimed exposure to chemicals, asbestos, contaminants), property damage (e.g., claimed damage due to leaking equipment, fire), economic loss (e.g., breach of contract, other commercial claims), employment (e.g., wrongful termination), and other claims for damage and relief. In fiscal 2004, the Company settled a wrongful discharge lawsuit with a former employee. In connection with that settlement, the Company became aware of an investigation which the Company believes was initiated based on discussions between the former employee and the FDA regarding the Company's SYSTEM 1 ® sterile processing system. The investigation is currently being conducted by the FDA and the U.S. Department of Justice and is ongoing. The Company has offered and intends to cooperate with the government agencies regarding this matter and has received a subpoena for documents relating to this matter.

The Company believes it has adequately reserved for its current litigation and that the ultimate outcome of its pending lawsuits and claims will not have a material adverse effect on the Company's consolidated financial position or results of operations taken as a whole. Due to their inherent uncertainty, however, there can be no assurance of the ultimate outcome of current or future litigation, proceedings, investigations, or claims or their effect. The Company presently maintains product liability insurance coverage and other liability coverage in amounts and with deductibles that it believes are prudent.

From time to time, STERIS is also involved in legal proceedings as a plaintiff involving contract, patent protection, and other claims asserted by the Company. Gains, if any, from these proceedings are recognized when they are realized. The Company believes there have been no material recent developments concerning the Company's legal proceedings since March 31, 2004 and no new material pending legal proceedings required to be reported.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

No repurchases of Common Shares were made by or on behalf of the Company during the third quarter of fiscal 2005. As of December 31, 2004, 2,726,000 shares remained authorized for repurchase under the new share repurchase program that was approved by the Company's Board of Directors and announced on July 28, 2004. This Common Share repurchase authorization does not have a stated maturity date.

ITEM 5. OTHER INFORMATION

On February 8, 2005, the Company entered into an executive retention agreement with Peter A. Burke, the Company's Senior Vice President and Chief Technology Officer. The agreement provides that Dr. Burke will receive a base salary of \$24,167 per month for the fiscal year ending March 31, 2005, as well as the opportunity to participate in the Company's cash bonus plan, benefit plans and stock option programs.

Under the terms of the agreement, if Dr. Burke's employment with the Company is terminated prior to the third anniversary of the agreement (or, after an extension of the agreement, prior to the third anniversary of the extension) by the Company without "cause" (as defined in the agreement) or by Dr. Burke for "good reason" (as defined in the agreement), Dr. Burke will be entitled to receive, subject to his execution of a release of all claims against the Company and his compliance with his obligations under the agreement (a) his then-current salary for either (1) the number of months remaining from the date of his termination to the date of the third anniversary of the agreement or any extension thereof, as applicable, or (2) twelve months, whichever is greater, (b) continuation of medical and dental benefits for such period, and (c) a one-time payment equal to the amount Dr. Burke would have been entitled to receive as a bonus relating to the fiscal year in which the termination occurred, pro rated to the date of termination. In addition, if Dr. Burke's employment with the Company is terminated during the first twelve months of the agreement by the Company without cause or by Dr. Burke for good reason, for purposes of calculating Dr. Burke's service requirements under the Company's 2002 Stock Option Plan, Dr. Burke will be considered to have additional service credit with the Company equal to the number of months from the date of such termination to the date of the first anniversary of the agreement.

ITEM 6. EXHIBITS

(a) Exhibits required by Item 601 of Regulation S-K

Exhibit Number	Exhibit Description
3.1	1992 Amended Articles of Incorporation of STERIS Corporation, as amended on May 14, 1996, November 6, 1996, and August 6, 1998 (filed as Exhibit 3.1 to Form 10-K for the fiscal year ended March 31, 2000, and incorporated herein by reference).
3.2	Amended and Restated Regulations of STERIS Corporation effective July 28, 2004 (filed as Exhibit 3.2 to Form 10-Q for the fiscal quarter ended September 30, 2004, as originally filed, and incorporated herein by reference).

4.1 Specimen Form of Common Stock Certificate (filed as Exhibit 4.1 to Form 10-K for the fiscal year ended March 31, 2002, and incorporated herein by reference).

- 4.2 Amended and Restated Rights Agreement, dated as of January 21, 1999, between STERIS Corporation and National City Bank, as successor Rights Agent (filed as Exhibit 4.2 to the Registration Statement on Form 8-A filed April 16, 1999, and incorporated herein by reference).
- 4.3 Amendment No. 1, dated June 7, 2002, to Amended and Restated Rights Agreement, dated as of January 21, 1999, between STERIS Corporation and National City Bank, as successor Rights Agent (filed as Exhibit 4.1 to the Registration Statement on Form 8-A/A filed June 10, 2002, and incorporated by reference).
- 10.1 Asset Purchase Agreement dated as of November 15, 2004 between Cosmed Group, Inc. and Registrant.
- 10.2 Form of STERIS Corporation Nonqualified Stock Option Agreement for Employees.
- 10.3 Form of STERIS Corporation Notice of Restricted Grant for Directors.
- 10.4 Form of STERIS Corporation Nonqualified Stock Option Grant Agreement for Directors.
- 15.1 Letter Regarding Unaudited Interim Financial Information.
- 31.1 Certification of the Chief Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 31.2 Certification of the Chief Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 32.1 Certification of the Chief Executive Officer and Chief Financial Officer Pursuant to 18 U.S.C. Section 1350 as Adopted Pursuant Section 906 of the Sarbanes-Oxley Act of 2002.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

STERIS Corporation

/s/ Laurie Brlas

Laurie Brlas Senior Vice President and Chief Financial Officer February 8, 2005

<u>EXHIBIT 10.1</u>

EXECUTION VERSION

ASSET PURCHASE AGREEMENT

between

COSMED GROUP, INC.

and

STERIS CORPORATION

Dated as of November 15, 2004

_

TABLE OF CONTENTS

		Page
ARTICLE I	SALE AND PURCHASE OF THE ASSETS AND ASSUMPTION OF LIABILITIES	1
1.1	Assets	1
1.2	Excluded Assets	3
1.3	Assumption of Liabilities	4
1.4	Excluded Liabilities	5
1.5	Transfer of Additional Chambers	5
ARTICLE II	THE CLOSING	6
2.1	Place and Date	6
2.2	Purchase Price	6
2.3	Purchase Price Adjustment	6
2.4	[Intentionally Omitted]	7
2.5	Consent of Third Parties	7
2.6	Release of the Chamber Holdback	8
2.7	Proration	8
ARTICLE III	REPRESENTATIONS AND WARRANTIES	9
3.1	Representations and Warranties of the Seller	9
3.2	Representations and Warranties of the Buyer	24
ARTICLE IV	COVENANTS	25
4.1	Covenants of the Seller	25
4.2	Covenants of the Buyer	30
4.3	Antitrust Notification	31
4.4	Change Notices	32
4.5	Consent Decree	32
4.6	Environmental Permits	33
ARTICLE V	CLOSING CONDITIONS	33
5.1	Conditions to Obligations of Each Party	33
5.2	Conditions to Obligations of the Buyer	33
5.3	Conditions to Obligations of the Seller	36

ARTICLE VI 6.1 6.2	EMPLOYEE MATTERS Employment of the Seller's Employees Employee and Related Matters	37 37 37 37
ARTICLE VII 7.1 7.2	TERMINATION Termination Effect of Termination	38 38 39
ARTICLE VIII 8.1	DEFINITIONS Definition of Certain Terms	39 39
ARTICLE IX 9.1 9.2 9.3 9.4 9.5 9.6 9.7	INDEMNIFICATION Indemnification By the Seller Indemnification by the Buyer Notice and Opportunity to Defend Survival Limitations in Indemnification Specific Performance; General Provisions Holdback Amount	49 49 50 51 52 52 53
ARTICLE X 10.1 10.2 10.3 10.4 10.5 10.6 10.7 10.8 10.9 10.10 10.11	MISCELLANEOUS Expenses Severability Notices Headings Entire Agreement Counterparts Governing Law; Consent to Jurisdiction Binding Effect Assignment No Third Party Beneficiaries Amendment: Waivers, etc	54 54 54 55 55 56 56 56 56 56 56 56 56
10.12	Construction	56

- ii -

Table of Exhibits

- Exhibit AForm of Consent DecreeExhibit BForm of Non-Compete AgreementExhibit CForm of License AgreementExhibit DForm of Maintenance AgreementExhibit EForm of Supply Agreement
- Exhibit F Form of Bill of Sale, Assignment and General Conveyance
- Exhibit G Form of Assignment and Assumption Agreement
- Exhibit H Environmental Audits Performed by Buyer

- iii -

ASSET PURCHASE AGREEMENT

This ASSET PURCHASE AGREEMENT ("<u>Agreement</u>") dated as of November 15, 2004, is by and between Cosmed Group, Inc., a Maryland corporation (the "<u>Seller</u>"), and STERIS Corporation, an Ohio corporation (the "<u>Buyer</u>").

RECITALS

A. The Seller operates a medical device sterilization division (the "<u>Division</u>"), which provides contract ethylene oxide sterilization services for the sterilization of medical products, devices, supplies, and equipment, and Other Services to the Non-Medical Customers, through the Facilities (as defined herein) of the Seller (the "<u>Business</u>");

B. The Buyer wishes to purchase or acquire from the Seller, and the Seller wishes to sell, assign, convey, deliver and transfer ("<u>Transfer</u>") to the Buyer, substantially all of the assets and properties held in connection with, necessary for, or material to the business and operations of the Business (except for the Excluded Assets, as defined in <u>Section 1.2</u>), and the Buyer has agreed to assume the Assumed Liabilities, all for the Purchase Price (as defined herein) and upon the terms and subject to the conditions hereinafter set forth;

C. Michael L. Howe and David G. Howe (the "<u>Principal Stockholders</u>") have entered into a Limited Guaranty Agreement, dated as of the date hereof, whereby such stockholders of Seller have agreed to guarantee the obligations of the Seller as set forth and subject to the limitations in such Limited Guaranty Agreement; and

D. Certain capitalized terms used herein shall have the meanings set forth in Article VIII.

NOW, THEREFORE, in consideration of the mutual covenants, representations and warranties made herein, the mutual benefits to be derived hereby, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

ARTICLE I SALE AND PURCHASE OF THE ASSETS AND ASSUMPTION OF LIABILITIES

1.1 <u>Assets</u>. Subject to and upon the terms and conditions set forth in this Agreement, at the Closing (except as otherwise provided in <u>Section 1.5</u>), the Seller will Transfer, or cause to be Transferred, to the Buyer, and the Buyer will purchase or acquire from the Seller, all right, title and interest of the Seller, directly or indirectly, in and to the properties, assets and rights of every nature, kind and description, tangible and intangible (including goodwill), whether real, personal or mixed, and whether now existing or hereinafter acquired (other than the Excluded Assets) that are located at the Facilities or otherwise relating to or used by or held for the benefit of the Business as the same may exist on the Closing Date (collectively, together with the assets to be conveyed pursuant to <u>Section 1.5</u>, the <u>"Assets</u>"), free and clear of all Liens other than Permitted Liens, including the following assets of the Business:

(a) All chambers (including an aggregate of eight chambers at the San Diego Facility, two of which are the subject of <u>Section 1.5</u>), facilities, machinery, equipment, computer hardware, furniture, furnishings, vehicles, tools and similar property held for use at the Facilities or by the Business (including any of the foregoing purchased subject to any conditional sales or title retention agreement in favor of any other Person), including those chambers, forklifts and vehicles listed on <u>Schedule 1.1(a)</u>;

(b) All raw materials (including process gases), spare parts, replacement and component parts and office and other supplies, including such items previously purchased and in transit to the Seller for the benefit of the Business or located at or to be delivered to the Facilities;

(c) Subject to <u>Section 2.5</u>, all contracts, agreements, leases, commitments, instruments, guarantees, purchase orders, licenses and other agreements ("<u>Contracts</u>") related to the Business to which the Seller is a party specifically listed or described on <u>Schedule 1.1(c)</u> (collectively, the "<u>Assumed</u> <u>Contracts</u>");

(d) All credits, prepaid expenses, deferred charges, advance payments, deposits, prepaid items, rebates and allowances of the Business;

(e) All accounts receivable, unpaid accounts and other receivables and all notes, bonds and other evidences of indebtedness of, or rights to receive payments from, any Person arising out of the Business, including any amounts received by the Seller with respect to the foregoing after the Closing;

(f) All Intellectual Property and all rights thereunder or in respect thereof, including rights to sue for and remedies against past, present and future infringements thereof, and rights of priority and protection of interests therein under Applicable Law, including those items listed on <u>Schedule 1.1(f)</u> but excluding (i) the Licensed IP and (ii) the Excluded Non-Licensed IP;

(g) All Division or Business books, records, manuals and other materials (in any form or medium), advertising matter, catalogues, price lists, correspondence, mailing lists, lists of customers, distribution lists, photographs, production data, sales and promotional materials and records, purchasing materials and records, personnel records (subject to Applicable Law), manufacturing and quality control records and procedures, specifications, engineering data, equipment manuals, test data, blueprints, research and development files, records, data and laboratory books, Intellectual Property disclosures (relating to the Business Intellectual Property Assets other than the Licensed IP (solely to the extent that such Licensed IP is not located in a Facility) and the Excluded Non-Licensed IP), media materials, accounting records, sales order files, litigation files, instruction outlines and operating guides and booklets;

(h) Except as set forth on <u>Schedule 3.1(c)</u> with respect to non-transferrable Governmental Approvals, all Governmental Approvals, including all applications therefor and waivers thereof, including those set forth on <u>Schedule 1.1(h)</u>;

(i) All Real Property related to the Facilities;

- 2 -

(j) All rights to causes of action, lawsuits, judgments, claims and demands of any nature available to or being pursued by the Seller arising out of the Business, whether arising by way of counterclaim or otherwise, including those listed or described on <u>Schedule 1.1(j)</u>;

(k) Except as set forth on <u>Schedule 1.1(k)</u>, all guarantees, warranties, indemnities and similar obligations of any party in favor of the Seller with respect to the Business or to any of the Assets, and all claims or causes of action against any manufacturer, supplier or other transferor related to any of the Assets;

(1) All other assets of the Business shown or reflected on the balance sheet included in the Interim Financial Statements and all property and assets acquired by the Seller for the Business between the date of such balance sheet and the Closing (and, in each case, not disposed of in the Ordinary Course of Business as permitted by this Agreement);

(m) All insurance claims relating to the Assets or Assumed Liabilities; and

(n) All rights to receive and retain mail and other communications and all telephone numbers used by the Business (other than those primarily related to Excluded Assets or Excluded Liabilities).

1.2 <u>Excluded Assets</u>. The Seller and its Affiliates will retain and not Transfer, and the Buyer will not purchase or acquire, the following assets (collectively, the "<u>Excluded Assets</u>"):

(a) The assets listed on <u>Schedule 1.2(a)</u>;

(b) The rights of the Seller and its Affiliates under, and the funds and other property held in trust or under any funding vehicle with respect to, any of the Plans;

(c) All cash and cash equivalents, other than petty cash held at the Facilities on the Closing Date;

(d) All rights and incidents of interest of, and benefits accruing to, the Seller and its Affiliates in and to any Contracts of the Seller other than the Assumed Contracts;

(e) All assets owned or leased by Seller and its Affiliates at its locations in Jamestown, Rhode Island, Linden, New Jersey, Sparks, Nevada and Baltimore, Maryland;

(f) All of Seller's and its Affiliates' books, records, manuals and other materials (in any form or medium), advertising matter, catalogues, price lists, correspondence, mailing lists, lists of customers, distribution lists, photographs, production data, sales and promotional materials and records, purchasing materials and records, personnel records, manufacturing and quality control records and procedures, specifications, engineering data, equipment manuals, test data, blueprints, research and development files, records, data and laboratory books, Intellectual Property disclosures, media materials, accounting records, sales order files and litigation files that, in each case, do not relate to the Business or that relate to the Seller's corporate-wide functions;

- 3 -

(g) The Intellectual Property licensed to the Buyer pursuant to the License Agreement (the "Licensed IP"); and

(h) The Intellectual Property set forth on <u>Schedule 1.2(h)</u> (the "Excluded Non-Licensed IP").

1.3 <u>Assumption of Liabilities</u>. Subject to the terms and conditions set forth herein, at the Closing, the Buyer shall assume, pay, honor and discharge when due only the following liabilities of Seller arising out of the Assets and existing at or arising on or after the Closing Date (collectively, the "<u>Assumed Liabilities</u>"):

(a) all liabilities and obligations of the Seller to the extent reflected on the balance sheet included in the Interim Financial Statements and all liabilities and obligations of the Seller incurred or accrued for since the Balance Sheet Date in the Ordinary Course of Business, but only to the extent that such liabilities and obligations constitute trade payables of the Business;

(b) subject to <u>Section 2.5</u>, any and all liabilities, obligations and commitments under the Assumed Contracts; <u>provided</u>, <u>however</u>, that the Buyer shall not assume or be responsible for any such liabilities, obligations or commitments to be performed prior to the Closing Date or that arise from breaches by Seller of such Assumed Contracts or defaults by Seller occurring prior to the Closing or as a result of the Closing under such Assumed Contracts, all of which liabilities, obligations and commitments constitute Excluded Liabilities;

(c) if the Consent Decree is executed prior to the Closing in substantially the same form (including with respect to the nationwide facility audit provisions) as attached hereto as <u>Exhibit A</u>, the obligation to perform, and any penalties attributable to the Buyer's failure to meet compliance deadlines for, those obligations relating to the Facilities under the Consent Decree, but only to the extent set forth in Appendix B to the Consent Decree; <u>provided</u> that the costs of compliance with such obligations shall be allocated as set forth in Section 4.5.

(d) if the Consent Decree is not executed prior to the Closing, Buyer shall assume the obligation to join the negotiations of and to execute the Consent Decree in a form reasonably acceptable to the Buyer as the owner of the Facilities; <u>provided</u> that the cost of compliance with the executed Consent Decree shall be allocated as set forth in <u>Section 4.5</u>;

(e) all liabilities and obligations occurring on or after the Closing Date first arising out of the Buyer's ownership of the Assets and its conduct of the Business; and

(f) the liabilities specifically set forth on <u>Schedule 1.3(f)</u>.

- 4 -

1.4 <u>Excluded Liabilities</u>. Notwithstanding the provisions of <u>Section 1.3</u> or any other provision hereof or any Schedule or Exhibit hereto and regardless of any disclosure to the Buyer, the Buyer shall not assume any liabilities, obligations or commitments of the Seller other than the Assumed Liabilities (the "<u>Excluded Liabilities</u>"), including the following, all of which Excluded Liabilities shall be paid, performed and observed by the Seller:

(a) any liability or obligation of the Seller under any Indebtedness of the Seller, except with regard to capital leases included in the Assumed Contracts;

(b) any liability or obligation of the Seller for Taxes;

(c) any liability or obligation of the Seller for costs and expenses in connection with the negotiation and execution of this Agreement or the consummation of the transactions contemplated hereby;

(d) all liabilities and obligations relating to the compensation (including stock options and other equity-based compensation), salary, commissions and bonuses payable or granted to, or earned or accrued in respect of service performed by, Employees on or prior to the Closing Date;

(e) all intercompany accounts between the Seller or any of the Seller's Affiliates and the Business;

(f) any liability or obligation of the Seller arising out of or related to any breach of warranty, product defect or violation of Applicable Law by Seller relating to any service provided by the Business prior to the Closing Date;

(g) all liabilities related to recording the chain of title of all Business Intellectual Property Assets into the name of Seller before the Transfer to the Buyer pursuant to this Agreement;

(h) any and all Benefit Liabilities;

(i) all Environmental Liabilities and Costs including those relating to or arising out of matters addressed in the Consent Decree except to the extent assumed by the Buyer pursuant to <u>Section 1.3(c)</u> or <u>1.3(d)</u>;

(j) all liabilities and obligations related to the Excluded Assets; and

(k) all liabilities and obligations of the Seller that are not Assumed Liabilities, including any liabilities and obligations of the Seller related to or arising out of the operation of the Business or the ownership of the Assets on or prior to the Closing Date other than the Assumed Liabilities, and any liabilities and obligations of the Seller related to or arising out of the operation of any business other than the Business by the Seller, whether before or after the Closing.

1.5 <u>Transfer of Additional Chambers</u>. The Seller shall complete the production of the Additional Chambers, and install, in a fully operational manner, the Additional Chambers at the San Diego Facility, using commercially reasonable efforts to complete this work as quickly as

- 5 -

possible after the date hereof, and the Seller shall keep the Buyer promptly informed about the status of such production and installation. The parties shall reasonably cooperate with each other, at their own expense, to determine whether each such Additional Chamber is Commissioned, and the Seller, at its own expense, shall correct any deficiencies found by either party in any Additional Chamber that prevent such Additional Chamber from being fully operational and Commissioned. Buyer shall use commercially reasonable efforts to cooperate with correctional measures taken by Seller. Any Additional Chamber that is installed and Commissioned prior to the Closing shall be Transferred to the Buyer on the Closing Date together with the other Assets pursuant to <u>Section 1.1</u>. If any Additional Chamber is installed and Commissioned subsequent to the Closing, such Additional Chamber shall be automatically Transferred to the Buyer immediately upon being Commissioned.

ARTICLE II THE CLOSING

2.1 <u>Place and Date</u>. Unless this Agreement is terminated in accordance with <u>Section 7.1</u>, the closing of the sale and purchase of the Assets and the assignment and assumption of the Assumed Liabilities (the "<u>Closing</u>") shall take place at the offices of Edwards & Angell, LLP, 2800 Financial Plaza, Providence, Rhode Island 02903, on the second Business Day following the satisfaction or waiver of each of the conditions set forth in <u>Article V</u> (other than those conditions that are to be satisfied at the Closing) or such other time and place upon which the parties may agree. The day on which the Closing actually occurs is referred to as the "<u>Closing Date</u>". The Closing shall be deemed to have taken place at 12:01 a.m. on the Closing Date.

2.2 <u>Purchase Price</u>. On the terms and subject to the conditions set forth in this Agreement and subject to adjustment as provided in <u>Section 2.3</u>, the Buyer shall (a) pay or cause to be paid to the Seller Seventy-Three Million Dollars (\$73,000,000) in cash <u>minus</u> the sum of the Holdback Amount and the Chamber Holdback and (b) assume the Assumed Liabilities. The "<u>Purchase Price</u>" shall be \$73,000,000, as adjusted pursuant to <u>Section 2.3</u>. The Purchase Price (unadjusted), <u>less</u> the Holdback Amount and <u>less</u> the Chamber Holdback, shall be payable at the Closing by wire transfer of immediately available funds to such bank account or accounts designated in writing by the Seller to the Buyer at least five days prior to the Closing.

2.3 Purchase Price Adjustment.

(a) Within 90 days of the Closing Date, the Buyer shall prepare and deliver, or cause to be prepared and delivered, to the Seller a net working capital statement (the "<u>Working Capital Statement</u>"), setting forth the calculation of the Working Capital of the Business as of the Closing Date ("<u>Closing Working Capital</u>") together with detailed supporting documentation of such calculation. The Working Capital Statement must be prepared in accordance with generally accepted accounting principles ("<u>GAAP</u>") and in a manner consistent with the working capital statement attached hereto as <u>Schedule 2.3(a)</u> and the principles of preparation included on such Schedule.

(b) Within 30 days following receipt by the Seller of the Working Capital Statement, the Seller shall deliver written notice to the Buyer of any dispute the Seller has with respect to the preparation or content of the Working Capital Statement, and such notice shall include specific detail and backup information regarding the basis of such

- 6 -

dispute. In the event that the Seller does not notify the Buyer of a dispute with respect to the Working Capital Statement within such 30-day period, such statement will be final, conclusive and binding on the parties. In the event of such notification of a dispute, the Buyer and the Seller shall negotiate in good faith to resolve such dispute. If the Buyer and the Seller, notwithstanding such good faith effort, fail to resolve such dispute within 30 days after the Seller advises the Buyer of the Seller's objections, then the Buyer and the Seller jointly shall engage the firm of KPMG LLP (the "Arbitration Firm"), to resolve such dispute. The Arbitration Firm shall utilize the Rules for Commercial Arbitration of the American Arbitration Association to the extent applicable. All determinations made by the Arbitration Firm shall be final, conclusive and binding on the parties. The Buyer and the Seller shall share equally the fees and expenses of the Arbitration Firm.

(c) For purposes of complying with the terms set forth in this <u>Section 2.3</u>, each party shall cooperate with and make available to the other party and its representatives all information, records, data and working papers and shall permit access to its facilities and personnel, as may be reasonably required in connection with the preparation and analysis of the Working Capital Statement and the resolution of any disputes under the Working Capital Statement.

(d) Subject to <u>Section 2.3(f)</u> below, if Closing Working Capital (as finally determined pursuant to <u>Section 2.3(b)</u>) is less than an amount equal to the Target Working Capital, then the Purchase Price will be adjusted by the amount of such shortfall and the Seller shall pay, or caused to be paid, by bank wire transfer of immediately available funds to an account designated in writing by the Buyer, an amount in cash equal to such shortfall within five business days from the date on which Closing Working Capital is finally determined pursuant to <u>Section 2.3(b)</u>.

(e) Subject to <u>Section 2.3(f)</u> below, if Closing Working Capital (as finally determined pursuant to <u>Section 2.3(b)</u>) is greater than an amount equal to the Target Working Capital, then the Purchase Price will be adjusted by the amount of such excess and the Buyer shall pay or cause to be paid to the Seller by bank wire transfer of immediately available funds to an account designated in writing by the Seller, an amount in cash equal to such excess within five Business Days from the date on which Closing Working Capital is finally determined pursuant to <u>Section 2.3(b)</u>.

(f) Notwithstanding anything to the contrary in this <u>Section 2.3</u>, no adjustment to the Purchase Price shall be made, and no amounts shall be payable, pursuant to this <u>Section 2.3</u> unless the difference between Closing Working Capital (as finally determined pursuant to <u>Section 2.3(b)</u>) and Target Working Capital is at least \$10,000.

2.4 [Intentionally Omitted]

2.5 <u>Consent of Third Parties</u>. Notwithstanding anything to the contrary in this Agreement, this Agreement shall not constitute an agreement to Transfer any Governmental Approval, Contract or other agreement or arrangement or any claim, right or benefit arising thereunder or resulting therefrom if a Transfer or an attempt to make such a Transfer without the Consent of a Government Authority or third party would constitute a breach or violation thereof

- 7 -

or adversely affect the rights of the Buyer or the Seller thereunder; and any Transfer to the Buyer by the Seller of any interest under any such Governmental Approval, Contract or other agreement or arrangement that requires the Consent of a Government Authority or third party shall be made subject to such Consent being obtained. In the event any such Consent is not obtained on or prior to the Closing Date, the Seller shall continue to use all reasonable efforts to obtain any such Consent after the Closing Date until such time as such Consent has been obtained, and the Seller and Buyer will reasonable cooperate with each other in any lawful arrangement to provide that the Buyer will receive the interest of the Seller in the benefits (and will fulfill all obligations) under any such Government Approval, Contract or other agreement or arrangement, including performance by the Seller as agent. The Seller shall pay and discharge, and shall indemnify and hold the Buyer harmless from and against any and all costs, expenses or liabilities related to Seller performing as Buyer's agent as contemplated by this <u>Section 2.5</u> with respect to those Governmental Approvals and Consents listed in <u>Schedule 3.1(c)</u>. Nothing in this <u>Section 2.5</u> shall be deemed a waiver by the Buyer of its right to have received on or before the Closing an effective assignment of all of the Assets nor shall this <u>Section 2.5</u> be deemed to constitute an agreement to exclude from the Assets any assets described under <u>Section 1.1</u>. Notwithstanding anything to the contrary herein, the Seller's obligations under this <u>Section 2.5</u> shall terminate on the second anniversary of the Closing Date or for any specific Governmental Approval or Consent upon Buyer receiving such or the equivalent.

2.6 <u>Release of the Chamber Holdback</u>. If neither of the Additional Chambers is Transferred to the Buyer at the Closing, within three Business Days of the Transfer of each Additional Chamber to the Buyer pursuant to <u>Section 1.1</u> and <u>Section 1.5</u>, the Buyer shall pay or cause to be paid by bank wire transfer of immediately available funds, to the account designated in writing by the Seller, an amount equal to one-half of the Chamber Holdback. If one and only one of the Additional Chambers is Transferred to the Buyer at the Closing, within three Business Days of the Transfer of the other Additional Chamber to the Buyer at the Closing, within three Business Days of the Transfer of the other Additional Chamber to the Buyer pursuant to <u>Section 1.1</u> and <u>Section 1.5</u>, the Buyer shall pay or cause to be paid by bank wire transfer of immediately available funds, to the account designated in writing by the Seller, an amount equal to the Chamber Holdback.

2.7 <u>Proration</u>. At the Closing, the parties shall apportion all Real Property Taxes (as defined below), Personal Property Taxes (as defined below), Utility Charges (as defined below), license fees, lease fees, rent and other costs and expenses directly attributable to the Business to any period that begins before the Closing and ends after the Closing (a "<u>Straddle Period</u>") on a pro-rata basis, with Seller being liable for any taxes or charges attributable to the days in the Straddle Period up to and including the day immediately preceding the Closing and Buyer being liable for taxes or charges attributable to the remaining days in the Straddle Period (including the Closing Date). Proration of Real Property Taxes and Personal Property Taxes shall be made on the basis of the most recent tax valuation and assessment for the Assets. The Buyer and the Seller shall cooperate in assuring that the such apportioned obligations are properly paid. For purposes of this <u>Section 2.7</u>, "<u>Personal Property Taxes</u>" shall mean ad valorem taxes with respect to the Owned Real Property and the Leased Real Property to the extent the Leases for such Leased Real Property require payment of such taxes by the lessee; and "<u>Utility Charges</u>" shall mean water, sewer, electricity, natural gas, telephone and other utility charges, if any, applicable to the Assets.

- 8 -

ARTICLE III REPRESENTATIONS AND WARRANTIES

Certain agreements and other matters are listed in the disclosure schedules to this <u>Section 3.1</u> (the "<u>Disclosure Schedules</u>") for disclosure purposes only and shall not be taken as an admission by Seller that such disclosures are required under or responsive to the terms of any representations and warranties contained in this Agreement. The disclosure of any agreement or other matter in the Disclosure Schedules do not evidence a determination by Seller that such disclosure rises above any applicable materiality thresholds. Any matter disclosed in one section of the Disclosure Schedules is only deemed to be disclosed in such other sections of the Disclosure Schedules for which it is readily apparent to a reader (on its face or by way of an appropriate cross reference) that such item is required by this Agreement to be disclosed in such other sections (by way of example, without limiting the generality of the foregoing, disclosure of a breach of contract shall be disclosure of such breach for all representations in which is readily apparent (on its face or by way of an appropriate cross reference) that such breach is required to be disclosed by this Agreement, including the representation relating to the undisclosed liabilities, but disclosure of the existence of a contract shall not be disclosure of the need for consent under such contract in connection with the transaction provided for hereby in the absence of a statement of the need for such consent, or of a breach of such contract in the absence of such breach).

3.1 <u>Representations and Warranties of the Seller</u>. The Seller represents and warrants to the Buyer as follows:

(a) <u>Authorization</u>. The Seller has the corporate power and authority to execute and deliver this Agreement and each of the Ancillary Agreements, to perform fully its obligations hereunder and thereunder, and to consummate the transactions contemplated hereby and thereby. The execution and delivery by the Seller of this Agreement, and the consummation of the transactions contemplated hereby, have been, and on the Closing Date the execution and delivery by the Seller of each of the Ancillary Agreements and the consummation of the transactions contemplated hereby will have been, duly authorized by all requisite corporate and stockholder action of the Seller. The Seller has duly executed and delivered this Agreement and on the Closing Date the Seller will have duly executed and delivered each of the Ancillary Agreements. This Agreement is, and on the Closing Date each of the Ancillary Agreements will be, legal, valid and binding obligations of the Seller, enforceable against it in accordance with their respective terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium, liquidation, fraudulent conveyance and other Applicable Laws and principles of equity affecting creditors' rights and remedies generally (the "<u>General Enforceability Exceptions</u>").

(b) <u>Corporate Status</u>. (i) The Seller is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation, with full corporate power and authority to carry on the Business and to own or lease and to operate its properties as and in the places where the Business is conducted and such properties are owned or leased and operated.

- 9 -

(ii) The Seller is duly qualified or authorized to do business and is in good standing in each of the jurisdictions specified in <u>Schedule 3.1(b)</u>, which are the only jurisdictions in which the operation of the Business or the character of the properties owned, leased or operated by it in connection with the Business makes such qualification or authorization necessary.

(iii) The Seller has delivered to the Buyer complete and correct copies of its articles of incorporation and bylaws or other organizational documents, in each case, as amended and in effect on the date hereof. The Seller is not in violation of any of the provisions of its articles of incorporation or bylaws or other organizational documents.

(c) <u>No Conflicts</u>. The execution, delivery and performance by the Seller of this Agreement and each of the Ancillary Agreements, and the consummation of the transactions contemplated hereby and thereby, do not and will not (i) violate or conflict with the articles of incorporation or the bylaws or other organizational documents of the Seller or any Applicable Law, (ii) violate, conflict with or result in a breach or termination of, or otherwise give any Person additional rights or compensation under, or the right to terminate or accelerate, or constitute (with notice or lapse of time, or both) a default under the terms of, any Assumed Contract or by which any of the Assets are bound or (iii) result in the creation or imposition of any Lien with respect to, or otherwise have an adverse effect upon, the Business or any of the Assets. Except as specified in <u>Schedule 3.1(c)</u>, no Consent is required to be obtained from any Person or Governmental Authority or made by the Seller in connection with the execution and delivery of this Agreement and the Ancillary Agreements or the consummation of the transactions contemplated hereby or thereby.

(d) <u>Financial Statements</u>. (i) <u>Schedule 3.1(d)</u> sets forth true and complete copies of (A) the unaudited balance sheets of the Business as of December 31, 2001, December 31, 2002, and December 31, 2003, and the related unaudited statements of income for the fiscal years then ended, together with the notes thereto, and the other financial information included therewith (collectively, the "<u>Financial Statements</u>"), and (B) the unaudited balance sheet of the Business as of September 30, 2004, and the related unaudited statement of income for the nine-month period then ended (the "<u>Interim Financial</u> Statements"). The Financial Statements were derived from the audited financial statements of the Seller.

(ii) The Financial Statements present fairly, in all material respects, the financial position and results of operations of the Business at the dates and for the time periods indicated and have been prepared and reviewed by the management of the Business in accordance with GAAP, consistently applied throughout the periods indicated. The Interim Financial Statements present fairly, in all material respects, the financial position and results of operations of the Business at the date and for the period indicated and have been prepared and reviewed by the management of the Seller in accordance with GAAP, consistent with the Financial Statements, except for the absence

- 10 -

of footnote disclosure and any customary year-end adjustments. The Financial Statements and the Interim Financial Statements were derived from the books and records of the Seller.

(e) <u>Absence of Undisclosed Liabilities</u>. The Seller does not have any liabilities related to the Business, the Assets or the Assumed Liabilities (whether accrued, absolute, contingent, unliquidated or otherwise, whether due or to become due, whether known or unknown, regardless of when asserted) arising out of transactions or events entered into prior to the date of this Agreement, or any action or inaction, or any state of facts existing, with respect to or based upon transactions or events occurring prior to the date of this Agreement, in each case relating to the Business, the Assets or the Assumed Liabilities except (i) liabilities reflected in the Financial Statements or the Interim Financial Statements, (ii) liabilities that have arisen after the date of the Interim Financial Statements in the Ordinary Course of Business; or (iii) as otherwise set forth on <u>Schedule 3.1(e)</u>.

(f) <u>Taxes</u>. (i) The Seller has (or by the Closing will have) duly and timely filed all Tax Returns relating to the Business required to be filed on or before the Closing Date except those subject to a validly filed extension ("<u>Covered Returns</u>"). All such Covered Returns are correct and complete in all material respects. All Taxes owed by the Seller and relating to the Business, whether or not shown on a Covered Return ("<u>Covered Taxes</u>"), have (or by the Closing Date will have) been duly and timely paid or properly accrued on Seller's most recent Interim Financial Statement or the balance sheet. All Taxes required to be withheld by or on behalf of the Seller in connection with amounts paid or owing to any employee, independent contractor, creditor or other party with respect to the Business ("<u>Withholding Taxes</u>") have been withheld and either duly and timely paid to the proper Governmental Authorities or set aside in accounts for such purpose. There are no Liens on any of the Assets that arose in connection with any failure (or alleged failure) to pay any Tax.

(ii) Except as set forth on <u>Schedule 3.1(f)(ii)</u>, no agreement or other document extending, or having the effect of extending, the period of assessment or collection of any Covered Taxes or Withholding Taxes, and no power of attorney with respect to any such Taxes, has been filed with the IRS or any other Governmental Authority.

(iii) Except as set forth on <u>Schedule 3.1(f)(iii)</u>, (i) there are no written, or to the knowledge of the Seller other, claims regarding Covered Taxes or Withholding Taxes asserted by any Governmental Authority to be due and (ii) no issue has been raised in writing by any Governmental Authority within the past 3 years in the course of any audit with respect to Covered Taxes or Withholding Taxes. Except as set forth on <u>Schedule 3.1(f)</u>, no Covered Returns are currently under audit by any Governmental Authority. Except as set forth on <u>Schedule 3.1(f)</u>, neither the IRS nor any other Governmental Authority is now asserting or, to the knowledge of the Seller, threatening to assert against the Seller any deficiency or claim for additional Covered Taxes.

- 11 -

(iv) Except as set forth on <u>Schedule 3.1(f)(iv)</u>, there is no litigation or administrative appeal pending or, to the knowledge of the Seller, threatened against or relating to the Seller in connection with Covered Taxes.

(v) Seller is not a foreign person within the meaning of Section 1445 of the Code.

(vi) Seller is not a party, and the Business is not subject, to any Tax allocation or sharing agreement under which Buyer or the Assets could be subject to Tax or other liability after the Closing.

(vii) None of the Assets is "tax exempt use property" within the meaning of Section 168(h) of the Code. None of the Assets is required to be treated as owned by any person or entity other than the Seller pursuant to the so-called "safe harbor lease" provisions of the Internal Revenue Code of 1954. None of the Assets directly or indirectly secures any debt the interest of which is exempt from federal income tax under Section 103 of the Code.

(viii) None of the Assumed Liabilities is an obligation to make a payment that will not be deductible under Section 280G of the Code.

(g) <u>Absence of Changes</u>. Except as set forth the relevant sub-section of <u>Schedule 3.1(g)</u>, since December 31, 2003 (the "<u>Balance Sheet Date</u>"), the Seller has conducted the Business only in the Ordinary Course of Business and has not, on behalf of, in connection with or relating to, the Business or the Assets:

(i) suffered any Material Adverse Effect;

(ii) incurred any obligation or liability, absolute, accrued, contingent or otherwise, whether due or to become due, except current liabilities for trade or business obligations incurred in connection with the purchase of goods or services in the Ordinary Course of Business or for capital expenditures reflected in the Interim Financial Statements;

(iii) discharged or satisfied any Lien other than those then required to be discharged or satisfied, or paid any obligation or liability, absolute, accrued, contingent or otherwise, whether due or to become due, other than current liabilities reflected on the balance sheet included in the Financial Statements and trade payables incurred since the date thereof in the Ordinary Course of Business;

(iv) mortgaged, pledged or subjected to a Lien, any property, business or assets, tangible or intangible, held in connection with the Business;

(v) sold, Transferred, leased to others or otherwise disposed of any of the Assets, except for inventory sold in the Ordinary Course of Business, or canceled or materially compromised any debt or claim, or waived or released any right of substantial value;

- 12 -

(vi) received any notice of termination of any material Contract or suffered any material damage, destruction or loss (whether or not covered by insurance);

(vii) Transferred or granted any rights under, or entered into any settlement regarding the breach or infringement of, any Business Intellectual Property Assets or modified any existing rights with respect thereto;

(viii) made any change in the rate of compensation, commission, bonus or other direct or indirect remuneration payable, or paid or agreed or orally promised to pay, conditionally or otherwise, any bonus, incentive, retention or other compensation, retirement, welfare, fringe or severance benefit or vacation pay, to or in respect of any employee, salesman, distributor or agent of the Seller relating to the Business;

(ix) encountered any labor union organizing activity, had any actual or threatened employee strikes, work stoppages, slowdowns or lockouts, or had any material change in its relations with its employees, agents, customers or suppliers;

(x) failed to maintain a quantity and quality of inventories and supplies of the Business (including nitrogen gas and ethylene oxide) consistent with past practice and usable in the Ordinary Course of Business, or made any purchase commitment in excess of the normal, ordinary and usual requirements of its business or at any price in excess of the then current market price or upon terms and conditions more onerous than those usual and customary in the industry, or made any change in its selling, pricing, advertising or personnel practices inconsistent with its prior practice and prudent business practices prevailing in the industry;

(xi) made any capital expenditures or capital additions or improvements in excess of an aggregate of \$50,000 other than the installation of chambers in the San Diego Facility as set forth in <u>Section 2.5</u>;

(xii) instituted, settled or agreed to settle any litigation, action or proceeding before any Governmental Authority relating to the Business or the Assets;

(xiii) made any change prior to the date of this Agreement in its general pricing practices or policies or any change in its credit or allowance practices or policies;

(xiv) entered into any amendment, modification, termination (partial or complete) or granted any waiver under or given any consent with respect to any Contract that is required (or had it been in effect on the date of this Agreement would have been required) to be disclosed in the Schedules to this Agreement;

(xv) received written or, to the knowledge of the Seller, oral notice from any customer or supplier that such customer or supplier has ceased, may cease or will cease to do business with it; or

(xvi) taken any action or, to the knowledge of the Seller, omitted to take any action that would result in the occurrence of any of the foregoing.

(h) <u>Litigation</u>. Except as set forth on <u>Schedule 3.1(h)</u>, there is no pending, and there has not been in the three years prior to the date of this Agreement any, action, claim, demand, suit, proceeding, arbitration, grievance, citation, notice of violation, notice of potential liability, summons, subpoena, inquiry or investigation of any nature, civil, criminal, regulatory or otherwise, in law or in equity, and there are no, and there has not been in the three years prior to the date of this Agreement any, Orders pending or, to the knowledge of the Seller, threatened against or relating to the Seller in connection with the Assets or the Business or against or relating to the transactions contemplated by this Agreement. Except as set forth in <u>Schedule 3.1(h)</u>, in the three years prior to the date of this Agreement there have not been any citations, fines or penalties asserted against the Seller with respect to the Business under any Applicable Law.

(i) <u>Compliance with Laws and Governmental Approvals</u>. (i) Except as disclosed in <u>Schedule 3.1(i)(i)</u>, since three years prior to the date of this Agreement, the Seller has complied in all material respects with all Applicable Laws applicable to the Business or the Assets, and the Seller has not received any notice alleging any conflict, violation, breach or default of any Applicable Law.

(ii) <u>Schedule 3.1(i)(ii)</u> sets forth all Governmental Approvals necessary for, or otherwise material to, the conduct of the Business. Except as set forth in <u>Schedule 3.1(i)(ii)</u>, all such Governmental Approvals have been duly obtained or submitted and are in full force and effect, and the Seller is in compliance, in all material respects, with each of such Governmental Approvals held by it with respect to the Assets and the Business.

(iii) To the knowledge of the Seller, there are no proposed laws, rules, regulations, ordinances, orders, judgments, decrees, governmental takings, condemnations or other proceedings pending against Seller which would be applicable to the Business or its operations or properties that would reasonably be expected to have a Material Adverse Effect.

(j) <u>Operation of the Business</u>. Except as set forth in <u>Schedule 3.1(j</u>), the Seller has conducted the Business only through the Seller and not through any other divisions or any direct or indirect Subsidiary or Affiliate of the Seller.

(k) <u>Assets</u>. Except as disclosed in <u>Schedule 3.1(k)</u>, the Seller has directly, or indirectly with respect to the Owned Real Property, good and marketable title to, or in the case of leased property, has valid leasehold interests in, all the Assets free and clear of any and all Liens other than Permitted Liens. The Assets comprise all of the assets required for the continued conduct of the Business by the Buyer as now being conducted. The Assets, taken as a whole, constitute all the properties and assets relating to or used or held for use in connection with the Business during the past twelve months (except cash disposed of, accounts receivable collected, prepaid expenses realized, Contracts fully performed, properties or assets replaced by substantially equivalent properties or assets, in each case in the Ordinary Course of Business, and except for the Excluded Assets). Except as disclosed in <u>Schedule 3.1(k)</u>, the Assets are in all material respects adequate for the purposes for which such assets are currently used or are held for use, and are in reasonably good repair and operating condition (subject to normal wear and tear) and, to

- 14 -

the knowledge of the Seller, there are no facts or conditions affecting the Assets which could, individually or in the aggregate, interfere in any material respect with the use, occupancy or operation thereof as currently used, occupied or operated, or their adequacy for such use.

(l) <u>Contracts</u>. (i) <u>Schedule 3.1(1)(i)</u> contains a complete and correct list of all written Contracts, and oral Contracts of which the Seller has knowledge, of the types described below (x) by which any of the Assets are bound or affected or (y) to which the Seller is a party and by which it is bound in connection with the Business or the Assets:

(A) Leases, licenses, permits, franchises, insurance policies, Governmental Approvals and other Contracts concerning or relating to the Real Property;

(B) Employment, consulting, agency, collective bargaining or other similar Contracts relating to or for the benefit of current, future or former employees, officers, directors, sales representatives, distributors, dealers, agents, independent contractors or consultants;

(C) Loan agreements, indentures, letters of credit, mortgages, security agreements, pledge agreements, deeds of trust, bonds, notes, guarantees and other Contracts relating to the borrowing of money or obtaining of or extension of credit, to the extent they will be an Assumed Contract;

(D) Licenses, licensing arrangements and other Contracts providing in whole or in part for the use of, or limiting the use of, any Intellectual Property other than "shrink-wrap" or off the shelf software licenses;

(E) Brokerage or finder's Contracts, to the extent they will be an Assumed Contract;

(F) Joint venture, partnership and similar Contracts involving a sharing of profits or expenses (including joint research and development and joint marketing Contracts);

(G) Asset purchase agreements and other acquisition or divestiture Contracts, including any Contracts relating to the sale, lease or disposal of any assets (other than sales of inventory in the Ordinary Course of Business) or involving continuing indemnity or other obligations;

(H) Orders and other Contracts for the purchase of materials, supplies, products or services, each of which involves aggregate payments in excess of \$50,000;

(I) Contracts in effect on the date of this Agreement for the provision of services or products to customers of the Business, each of which involves aggregate payments in excess of \$250,000 per annum or \$750,000 in the aggregate;

- 15 -

(J) Contracts with respect to which the aggregate amount that could reasonably be expected to be paid by the Seller thereunder in the future exceeds \$50,000 per annum or \$75,000 in the aggregate;

(K) Sales agency or marketing Contracts;

(L) Contracts with respect to the representation of the Business in foreign countries;

(M) Contracts providing for the leasing of personal property used in, or held for use in connection with, the Business which involve aggregate annual payments in excess of \$25,000;

(N) Contracts with any employee, director, officer, stockholder or Affiliate of the Seller; and

(O) Any other Contracts that are material to the Business.

(ii) The Seller has delivered to the Buyer complete and correct copies of all such written Contracts (other than Contracts with customers, of which Buyer has received only the top 25 based on revenues (which 25 contracts represented at least 75% of the total revenues of the Business for the eight month period ended August 31, 2004)), together with all amendments thereto, and accurate descriptions of all material terms of all such oral Contracts.

(iii) All Contracts are in full force and effect and enforceable against each party thereto. There does not exist under any Contract any event of default or event or condition that, after notice or lapse of time or both, would constitute a violation, breach or event of default thereunder on the part of the Seller or, to the knowledge of the Seller, any other party thereto except as set forth in <u>Schedule 3.1(1)(iii)</u>. Except as set forth in <u>Schedule 3.1(1)(iii)</u>, no Consent of any third party is required under any Contract as a result of or in connection with, and the enforceability of any Contract will not be affected in any manner by, the execution, delivery and performance of this Agreement or any of the Ancillary Agreements or the consummation of the transactions contemplated hereby or thereby.

(iv) Except as set forth in <u>Schedule 3.1(1)(iv</u>), the Seller is not restricted by any Contract with any other Person from carrying on the Business anywhere in the world. The Buyer, as a result of its purchase of the Business from the Seller pursuant hereto and the assumption of the Assumed Liabilities, will not thereby become restricted in carrying on any business anywhere in the world.

(m) <u>Receivables</u>. All of the Seller's receivables (including accounts receivable, loans receivable and advances) which have arisen in connection with the Business and which are reflected in the Financial Statements, and all such receivables which will have arisen since the Balance Sheet Date and are reflected on the accounting records of the Seller (collectively, the "<u>Accounts Receivable</u>") represent or will represent valid obligations arising from sales actually made or services actually performed in the

- 16 -

Ordinary Course of Business. Unless paid prior to the Closing Date, the Accounts Receivable are or will be as of the Closing current and collectible, except to the extent of any reserve included in the calculation of Closing Working Capital.

(n) <u>Customers and Suppliers</u>. (i) <u>Schedule 3.1(n)(i)</u> sets forth all customers that accounted for 3% or more of the revenues of the Seller in connection with the Business for each of the years ended December 31, 2002 and December 31, 2003 ("<u>Material Customers</u>"). Except as set forth on <u>Schedule 3.1(n)</u> (i), (i) all Material Customers continue to be customers of the Business and, other than fluctuations in the Ordinary Course of Business, none of the Material Customers has reduced materially its business with the Business from the levels achieved during the year ended December 31, 2003, and the Seller does not have any knowledge that any such reduction will occur; (ii) no Material Customer has terminated its relationship with the Seller or has threatened to do so; (iii) the Seller is not involved in any claim, dispute or controversy with any Material Customer relating to the Business; and (iv) the Seller is not involved in any claim, dispute or controversy with any other customers of the Business that, individually or in the aggregate, would reasonably be anticipated to have a Material Adverse Effect.

(ii) <u>Schedule 3.1(n)(ii)</u> sets forth the 10 largest suppliers, based on the dollar amount of purchases, of the Seller in connection with the Business for each of the years ended December 31, 2002 and December 31, 2003 ("<u>Material Suppliers</u>"). Except as set forth on <u>Schedule 3.1(n)(ii</u>), (i) all Material Suppliers continue to be suppliers of the Business and, other than fluctuations in the Ordinary Course of Business, none of the Material Suppliers has reduced materially its business with the Business from the levels achieved during the year ended December 31, 2003, and the Seller does not have any knowledge that any such reduction will occur; (ii) no Material Supplier has terminated its relationship with the Seller or has threatened to do so; (iii) the Seller is not involved in any claim, dispute or controversy with any Material Supplier relating to the Business; and (iv) the Seller is not involved in any claim, dispute or controversy with any of its other suppliers relating to the Business that, individually or in the aggregate, could reasonably be anticipated to have a Material Adverse Effect. Except as set forth on <u>Schedule 3.1(n)(ii</u>), no supplier to the Business represents a sole source of supply for goods and services used in the conduct of the Business.

(o) <u>Warranties</u>. Each service performed or otherwise delivered or provided, as the case may be, by the Seller in the operation of the Business, has been in conformity with all applicable contractual commitments and all express and implied warranties, and the Seller does not have any liability (and, to the knowledge of the Seller, there is no basis for any action, suit, proceeding, hearing, investigation, charge, complaint, claim or demand against the Seller) for any damages in connection therewith, subject only to the reserve for warranty claims set forth in the Financial Statements. No service provided by the Seller in the operation of the Business is subject to any guaranty, warranty or other indemnity beyond the terms and conditions of service contained in the customer Contracts, and except as set forth on <u>Schedule 3.1(0)</u>, the maximum potential liability of the Seller under any guaranty, warranty or other indemnity set forth in the Contracts with the 25 largest customers of the Business (as such Contracts are referenced in <u>Section 3.1(1)(ii)</u>) is limited to the cost of reprocessing the products (if required) and the manufacturing costs of the customer's product being processed.

- 17 -

(p) <u>Intellectual Property</u>. (i) <u>Schedule 3.1(p)(i)</u> contains a complete and correct list of all Business Intellectual Property Assets that are owned by the Seller (the "<u>Owned Business Intellectual Property</u>") and that constitute a patent, trademark registration, copyright registration, domain name registration or application for any of the foregoing. Except as set forth on <u>Schedule 3.1(p)(i)</u>, the Seller owns solely all Owned Business Intellectual Property, free from any Liens other than Permitted Liens. Except as set forth on <u>Schedule 3.1(p)(i)</u> or pursuant to a Contract set forth on <u>Schedule 3.1(l)(i)</u>. Seller has the right to use all Business Intellectual Property Assets free from any Liens other than Permitted Liens and free from any requirement of any past, present or future royalty payments, license fees, charges or other payments, or conditions or restrictions whatsoever. The Business Intellectual Property Assets comprise all of the Intellectual Property necessary for the Buyer to conduct and operate the Business as now being conducted by the Seller.

(ii) Immediately after the Closing, the Buyer will own all of the Owned Business Intellectual Property (except for the Licensed IP) and, except as set forth on <u>Schedule 3.1(p)(ii)</u>, will have a right to use all other Business Intellectual Property Assets, free from any Liens other than Permitted Liens and on the same terms and conditions as in effect prior to the Closing.

(iii) The conduct of the Business does not infringe or otherwise conflict with any rights of any Person in respect of any Intellectual Property. Except pursuant to a Contract set forth on <u>Schedule 3.1(l)(i)</u>, to the knowledge of the Seller, none of the Owned Business Intellectual Property is being infringed or otherwise used or available for use, by any Person other than the Seller.

(iv) To Seller's knowledge no claim or demand of any Person has been made and there is not any proceeding that is pending, or to the knowledge of the Seller, threatened, and, to the knowledge of the Seller, there is not a reasonable basis therefor, which (i) challenges the rights of the Seller in respect of any Business Intellectual Property Assets or (ii) asserts that the Seller is infringing or otherwise in conflict with, or is, except as set forth in <u>Schedule 3.1(p)(iv)</u> or pursuant to Contract set forth on <u>Schedule 3.1(l)(i)</u>, required to pay any royalty, license fee, charge or other amount with regard to, any Business Intellectual Property Assets. None of the Owned Business Intellectual Property and, to the knowledge of Seller, other Business Intellectual Property Assets is subject to any Order or administrative agency, or has been the subject of any litigation within the last five years, whether or not resolved in favor of the Seller.

(v) The Owned Business Intellectual Property identified on <u>Schedule 3.1(p)(i)</u> has been duly registered with, filed in or issued by, as the case may be, the United States Patent and Trademark Office or any other applicable filing offices, and the Seller has paid all fees and taken such other actions to ensure that such registrations, filings and issuances remain in full force and effect.

- 18 -

(q) <u>Insurance</u>. <u>Schedule 3.1(q)</u> contains a complete and correct list and summary description of all insurance policies maintained by the Seller for the benefit of or in connection with the Assets or the Business, including any occurrence-based liability policies in effect at any time during the Seller's operation of the Business. The Seller has made available to the Buyer complete and correct copies of all such policies together with all riders and amendments thereto. Such policies are in full force and effect, and all premiums due thereon have been paid. The Seller has complied in all material respects with the terms and provisions of such policies. The insurance coverage provided by such policies is adequate and customary for the Business as conducted by the Seller. <u>Schedule 3.1(q)</u> sets out all claims made by the Seller under any policy of insurance during the past two years with respect to the Business.

(r) <u>Real Property</u>. (i) <u>Schedule 3.1(r)(i)</u> contains a complete and correct list of all Owned Real Property setting forth the address and owner of each parcel of Owned Real Property. The Seller or its Affiliates have good and marketable fee simple title to the Owned Real Property free and clear of all Liens other than Permitted Liens. There are no outstanding options or rights of first refusal to purchase the Owned Real Property, or any portion thereof or interest therein.

(ii) <u>Schedule 3.1(r)(ii)</u> contains a complete and correct list of all Leases setting forth the address, landlord and tenant for each Lease. The Real Property is not subject to any Other Lease. The Seller has delivered to the Buyer correct and complete copies of the Leases. Each Lease is legal, valid, binding, enforceable, and in full force and effect, subject to the General Enforceability Exceptions. Neither the Seller nor, to the knowledge of the Seller, any other party, is in material default, violation or breach in any respect under any Lease, and no event has occurred and is continuing that constitutes or, with notice or the passage of time or both, would constitute a default, violation or breach by Seller, or to the Seller's knowledge, by any landlord, in any respect under any Lease. Each Lease grants the tenant under the Lease the right to use and occupy the demised premises thereunder. The Seller has good and marketable title to the leasehold estate under each Lease free and clear of all Liens other than Permitted Liens. The Seller enjoys peaceful and undisturbed possession under its respective Leases for the Leased Real Property.

(iii) Except for the Excluded Assets, the Real Property constitutes all the fee and leasehold interests in real property held for use in connection with, necessary for the conduct of the Business.

(iv) To Seller's knowledge, there are no eminent domain or other similar proceedings pending or, to the knowledge of the Seller, threatened affecting any portion of the Real Property, and Seller has received no written notice of any pending or threatened eminent domain or similar proceedings. There is no writ, injunction, decree, order or judgment outstanding, nor any action, claim, suit or proceeding, pending or, to the knowledge of the Seller, threatened, relating to the ownership, lease, use, occupancy or operation of any Real Property.

(v) Except as set forth on <u>Schedule 3.1(r)(v)</u>, the use and operation of the Real Property in the conduct of the Business does not violate in any material respect

- 19 -

any instrument of record or agreement with Seller affecting the Real Property. Except as set forth on <u>Schedule 3.1(r)(v)</u>, to the knowledge of Seller, there is no violation of any covenant, condition, restriction, easement or order of any Governmental Authority having jurisdiction over such property or of any <u>other Person</u> entitled to enforce the same affecting the Real Property or the use or occupancy thereof. No material damage or destruction has occurred with respect to any of the Real Property since three years prior to the date of this Agreement.

(vi) Except as set forth on <u>Schedule 3.1(r)(vi</u>), the Real Property is in full compliance with all applicable building, zoning, subdivision and other land use and similar Applicable Laws (subject to any applicable variance or special use permit) affecting the Real Property (collectively, the "<u>Real</u> <u>Property Laws</u>"), and the Seller has not received any notice of violation or claimed violation of any Real Property Law. There is no pending or, to the knowledge of the Seller, anticipated, change in any Real Property Law that would reasonably be expected to have or result in a Material Adverse Effect upon the ownership, use, occupancy or operation of the Real Property. Except as set forth on <u>Schedule 3.1(r)(vi</u>), no current use by the Seller of the Real Property is dependent on a nonconforming use or other Governmental Approval the absence of which would materially limit the use of such properties or assets held for use in connection with, necessary for the conduct of, or otherwise material to, the Business.

(vii) Except as set forth on <u>Schedule 3.1(r)(vii)</u>, each parcel included in the Real Property is assessed for real property tax purposes as a wholly independent tax lot, separate from adjoining land or improvements not constituting a party of that parcel.

(viii) The structures, improvements and fixtures at or upon the Real Property, including roofs and structural elements thereof and the electrical, plumbing, heating, ventilation, air conditioning, have to date been reasonably maintained and are in reasonable operating condition (normal wear and tear excepted) for their intended use subject to the provision of usual and customary maintenance and repair performed in the ordinary course with respect to similar properties of like age and construction, including normal wear and tear. All structures located on the Real Property are supplied with utilities and other services necessary for the operation of such structures as presently operated.

(s) <u>Environmental Matters</u>. (i) Except as set forth on <u>Schedule 3.1(s)(i)</u> and as provided in <u>Section 4.6</u>, all Environmental Permits will be Transferred to the Buyer on the Closing Date subject to the receipt of the necessary approvals. The Seller has not been notified by any relevant Governmental Authority that any Environmental Permit will be modified, suspended, canceled or revoked, or cannot be renewed in the Ordinary Course of Business.

(ii) Except as set forth in <u>Schedule 3.1(s)(ii)</u>, (A) the Seller has complied and is in compliance in all material respects with all Environmental Permits and all applicable Environmental Laws pertaining to the Real Property (and the use, ownership or transferability thereof) and the Business and (B) no Person has alleged any violation by the Seller of any Environmental Permits or any applicable Environmental Law relating to the conduct of the Business or the use, ownership or transferability of the Real Property.

- 20 -

(iii) Except as set forth in <u>Schedule 3.1(s)(iii</u>), the Seller has not caused or taken any action that has resulted or may result in, or has been or is subject to, any liability or obligation relating to (i) the environmental conditions on, under, or about any Real Property, the Assets or other properties or assets owned, leased or used by the Seller held for use in connection with, necessary for the conduct of, or otherwise material to, the Business, or (ii) the past or present use, management, handling, transport, treatment, generation, storage, disposal, arrangement for disposal or Release of any Hazardous Substances at any location.

(iv) Except as set forth in <u>Schedule 3.1(s)(iv)</u>:

(A) None of current or past operations conducted by the Seller, or any by-product thereof, and none of the currently or, to the knowledge of the Seller, formerly owned or operated property or assets of the Seller used in the Business, including the Assets and the Real Property, is or has been related to or subject to any investigation or evaluation by any Governmental Authority, as to whether any Remedial Action is needed to respond to a Release or threatened Release of any Hazardous Substances, nor, to the knowledge of the Seller, are there any conditions with respect to such property or assets which could give rise to such an investigation or evaluation in the future.

(B) The Seller is not subject to any outstanding order, judgment, injunction, decree or writ from, or contractual or other obligation to or with, any Governmental Authority or other Person in respect of which the Buyer may be required to incur any Environmental Liabilities and Costs.

(C) None of the Real Property is, and neither the Seller nor any of its Affiliates has transported or arranged for transportation (directly or indirectly) of any Hazardous Substances relating to the Assets, the Real Property or the operation of the Business to any location that is, listed or, to the knowledge of the Seller, proposed for listing, under CERCLA, or on any similar state list, or the subject of federal, state or local enforcement actions or investigations or Remedial Action regardless of whether such location was listed or proposed for listing at the time of transportation.

(D) No work, repair, construction or capital expenditure is required or planned in respect of the Assets or the Business pursuant to or to comply with any Environmental Law, nor has the Seller or its Affiliates received any notice of any such requirement, except for such work, repair, construction or capital expenditure which is not material to the Business and is in the Ordinary Course of Business.

(E) There are no underground tanks and related pipes, pumps and other facilities regardless of their use or purpose whether active or abandoned at the Real Property.

- 21 -

(F) To the knowledge of the Seller, there is no asbestos nor any asbestos-containing materials used in, applied to or in any way incorporated in any building, structure or other form of improvement on the Real Property. The Business does not sell and has not sold any product containing asbestos or that utilizes or incorporates asbestos-containing materials in any way.

(v) The Seller has disclosed and made available to the Buyer all information, including all studies, analyses and test results, in the possession, custody or control of the Seller and its Affiliates relating to (A) the environmental conditions on, under or about the Real Property, (B) Hazardous Substances used, managed, handled, transported, treated, generated, stored or Released by the Seller or any other Person at any time on any Real Property, or otherwise in connection with the use or operation of the properties or assets used in or held for use in connection with the Business, (C) noncompliance with Environmental Laws in connection with the Business or the Real Property, and (D) liability or potential liability under Environmental Law in connection with wastes generated by the operation of the Business coming to be located at any location.

(t) <u>Employees, Labor Matters, etc</u>. Except as set forth in <u>Schedule 3.1(t)</u>, the Seller is not a party to or bound by any collective bargaining agreement and there are no labor unions or other organizations representing, purporting to represent or attempting to represent any employees employed in the operation of the Business. Since three years prior to the date of this Agreement there has not occurred or, to the knowledge of the Seller, been threatened any material strike, slowdown, picketing, work stoppage, concerted refusal to work overtime or other similar labor activity with respect to any employees employed in the operation of the Business. Except as set forth on <u>Schedule 3.1(t)</u>, there are no labor disputes currently subject to any grievance procedure, arbitration or litigation and there is no representation petition pending or, to the knowledge of the Seller, threatened with respect to any employee employed in the operation of the Business. The Seller has complied, in all material respects, with all provisions of Applicable Laws pertaining to the employment of employees, including all such Applicable Laws relating to labor relations, equal employment, fair employment practices, entitlements, prohibited discrimination or other similar employment practices or acts.

(u) <u>Employee Benefit Plans and Related Matters</u>. (i) <u>Schedule 3.1(u)</u> sets forth a complete list of (i) all "employee benefit plans," as defined in Section 3(3) of ERISA, (ii) all other severance pay, salary continuation, bonus, incentive, stock option, retirement, pension, profit sharing or deferred compensation plans, Contracts, programs, funds, or arrangements of any kind, and (iii) all other employee benefit plans, Contracts, programs, funds, or arrangements (whether written or oral, qualified or nonqualified, funded or unfunded, foreign or domestic, currently effective or terminated) and any trust, escrow, or similar agreement related thereto, whether or not funded, that are maintained or contributed to by the Seller or any Affiliate of the Seller in respect of any present or former employees, consultants, or independent contractors of the Seller who are employed in the Business (all of the above being hereinafter individually or collectively referred to as "<u>Plan</u>" or "<u>Plans</u>," respectively). The Seller has no liability with respect to any plan, arrangement or practice of the type described in the preceding sentence other than the Plans.

- 22 -

(ii) Copies of the following materials have been delivered or made available to Buyer: (i) all current and prior plan documents for each Plan or, in the case of an unwritten Plan, a written description thereof, (ii) all determination letters from the IRS with respect to any of the Plans, (iii) all current and prior summary plan descriptions, summaries of material modifications, annual reports, and summary annual reports, (iv) all current and prior trust agreements, insurance Contracts, and other documents relating to the funding or payment of benefits under any Plan, and (v) any other documents, forms or other instruments relating to any Plan reasonably requested by Buyer.

(iii) Each Plan has been maintained, operated, and administered in material compliance with its terms and any related documents or agreements and in material compliance with all Applicable Laws.

(iv) Each Plan intended to be qualified under Section 401(a) of the Code is so qualified and has heretofore been determined by the IRS to be so qualified or is entitled to rely upon an opinion letter issued to a prototype sponsor as to the qualified status of such Plan, and each trust created thereunder has heretofore been determined by the IRS to be exempt from tax under the provisions of Section 501(a) of the Code, and, to the knowledge of Seller, there are no circumstances that would reasonably be expected to adversely affect the qualified status of such Plan.

(v) The Seller does not currently have and at no time in the past has had an obligation to contribute to a "defined benefit plan" as defined in Section 3(35) of ERISA, a pension plan subject to the funding standards of Section 302 of ERISA or Section 412 of the Code, a "multiemployer plan" as defined in Section 3(37) of ERISA or Section 414(f) of the Code or a "multiple employer plan" within the meaning of Section 210(a) of ERISA or Section 413(c) of the Code.

(vi) With respect to each group health plan benefiting any current or former employee of the Seller or any trade or business (whether or not incorporated) (i) under common control within the meaning of Section 4001(b)(1) of ERISA with the Seller or (ii) that together with the Seller is treated as a single employer under Section 414(t) of the Code (the "<u>Controlled Group</u>") that is subject to Section 4980B of the Code, the Seller and each member of the Controlled Group has complied with the continuation coverage requirements of Section 4980B of the Code and Part 6 of Subtitle B of Title I of ERISA.

(v) <u>No Guarantees</u>. Except as set forth on <u>Schedule 3.1(v)</u>, none of the obligations or liabilities of the Business or of the Seller incurred in connection with the operation of the Business is guaranteed by or subject to a similar contingent obligation of any other Person. Except as set forth on <u>Schedule 3.1(v)</u>, there are no outstanding letters of credit, surety bonds or similar instruments of the Seller or any of its Affiliates in connection with the Business or the Assets.

- 23 -

(w) <u>Related Party Transactions</u>. Except as set forth on <u>Schedule 3.1(w)</u>, no Affiliate of the Seller (i) has or during the last two years has had any direct or indirect interest (A) in, or is or during the last two years was, a director, officer or employee of, any Person that is a client, customer, supplier, lessor, lessee, debtor, creditor or competitor of the Seller or the Business or (B) in any material property, asset or right that is owned or used by the Seller or the Business or (ii) is, or during the last two years has been, a party to any agreement or transaction with the Seller or the Business.

(x) <u>Brokers, Finders, etc</u>. Except as set forth on <u>Schedule 3.1(x)</u>, all negotiations relating to this Agreement, the Ancillary Agreements, and the transactions contemplated hereby and thereby, have been carried on without the participation of any Person acting on behalf of the Seller or its Affiliates in such manner as to give rise to any valid claim against the Buyer for any brokerage or finder's commission, fee or similar compensation, or for any bonus payable to any officer, director, employee, agent or sales representative of or consultant to the Seller or its Affiliates upon consummation of the transactions contemplated hereby or thereby.

3.2 <u>Representations and Warranties of the Buyer</u>. The Buyer represents and warrants to the Seller as follows:

(a) <u>Corporate Status</u>; <u>Authorization</u>. The Buyer is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation and has, or prior to the Closing Date will have, full corporate power and authority to execute and deliver this Agreement and the Ancillary Agreements, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The execution and delivery by the Buyer of this Agreement and the consummation of the transactions contemplated hereby have been, and on the Closing Date the execution and delivery by the Buyer of each of the Ancillary Agreements and the consummation of the transactions contemplated thereby will have been, duly authorized by all requisite corporate action of the Buyer. The Buyer has duly executed and delivered this Agreement and on the Closing Date will have duly executed and delivered the Ancillary Agreements. This Agreement is, and on the Closing Date each of the Ancillary Agreements will be, valid and legally binding obligations of the Buyer, enforceable against the Buyer in accordance with their respective terms subject to the General Enforceability Exceptions.

(b) <u>No Conflicts</u>. The execution, delivery and performance by the Buyer of this Agreement and each of the Ancillary Agreements, and the consummation of the transactions contemplated hereby and thereby, do not and will not (i) violate or conflict with the articles of incorporation or the code of regulations or other organizational documents of the Buyer or any Applicable Law, or (ii) violate, conflict with or result in a breach or termination of, or otherwise give any Person additional rights or compensation under, or the right to terminate or accelerate, or constitute (with notice or lapse of time or both) a default under the terms of any note, deed, mortgage, other Contract or other instrument or oral understanding to which the Buyer is a party that, in the case of either (i) or (ii), has a material adverse effect on the ability of the Buyer to consummate the transactions contemplated by this Agreement and the Ancillary Agreements. Except as specified in <u>Schedule 3.2(b)</u>, no Consent is required to be obtained or made by the Buyer in connection with the execution and delivery of this Agreement or the Ancillary Agreements or the consummation of the transactions contemplated hereby or thereby.

- 24 -

(c) <u>Litigation</u>. There is no action, claim, suit or proceeding pending, or to the Buyer's knowledge, threatened, by or against or affecting the Buyer in connection with or relating to the transactions contemplated by this Agreement.

(d) <u>Brokers, Finders, etc</u>. All negotiations relating to this Agreement, the Ancillary Agreements, and the transactions contemplated hereby and thereby, have been carried on without the participation of any Person acting on behalf of the Buyer in such manner as to give rise to any valid claim against the Seller for any brokerage or finder's commission, fee or similar compensation.

(e) <u>Financial Ability to Close</u>. At Closing, Buyer will have the financial ability to perform its obligations under this Agreement and the Ancillary Agreements. The obligation of Buyer to consummate the transactions contemplated hereby is not subject to any financing contingency. Buyer is, and immediately following the Closing will be, Solvent.

ARTICLE IV COVENANTS

4.1 Covenants of the Seller.

(a) <u>Conduct of Business</u>. From the date hereof to the Closing Date, except as expressly permitted or required by this Agreement or as otherwise expressly consented to by the Buyer in writing (including by electronic mail), the Seller will:

(i) carry on the Business in, and only in, the Ordinary Course of Business, and use all reasonable efforts to preserve intact its present business organization, maintain its properties in good operating condition and repair (normal wear and tear excepted), keep available the services of its present officers and significant employees and preserve its relationship with customers, suppliers and others having business dealings with it, to use its commercially reasonable best efforts to maintain its goodwill and going business in all material respects in the same condition as on the date of this Agreement and on the Closing Date;

(ii) pay accounts payable and other obligations of the Business in the Ordinary Course of Business unless disputed in good faith;

(iii) perform in all material respects all of its obligations under all Contracts and other agreements and instruments relating to or affecting the Business or the Assets, and comply in all material respects with all Applicable Laws applicable to the Assets or the Business;

(iv) not enter into or assume any Contract relating to the Business involving an amount in excess of \$50,000, or enter into or permit any material amendment, supplement, waiver or other modification in respect thereof, except for Contracts with customers entered into in the Ordinary Course of Business;

- 25 -

(v) make all maintenance capital expenditures for the upkeep and maintenance of the Assets in the Ordinary Course of Business, including capital expenditures and repair and replacement expenditures consistent in all material respects with those set forth on Schedule 4.1(a)(v); and

(vi) not take any action, or omit to take any action, which action or omission would reasonably be expected to result in (A) any representation or warranty of the Seller set forth in this Agreement ceasing to be true and correct in all material respects, or (B) any condition to the Closing set forth in <u>Article V</u> not being satisfied.

(b) <u>Confidentiality</u>. The Seller will keep confidential and cause its Affiliates to keep confidential all non-public information relating to the Business, except for disclosures required by Applicable Law or administrative process (including disclosures required in Tax Returns or in other required filings with Governmental Authorities), <u>provided</u> that the Seller shall provide the Buyer with reasonable notice of any required disclosure, to the extent practicable, and except for information that becomes public other than as a result of a breach of this <u>Section 4.1(b)</u>.

(c) Access and Information. (i) So long as this Agreement remains in effect, the Seller will (and will cause each of its Affiliates and its and its Affiliates' respective accountants, counsel, consultants, employees and agents) after reasonable prior notice give the Buyer and its accountants, counsel, consultants, employees and agents hours to, and furnish them with all documents, records, work papers and information with respect to, all of such Person's properties, assets, books, Contracts, reports and records relating to the Division or the Business, as the Buyer shall from time to time reasonably request. In addition, the Seller will permit the Buyer, and its accountants, counsel, consultants, employees and agents, reasonable access to such personnel of the Seller during normal business hours after reasonable prior notice as may be necessary or useful to the Buyer in its review of the properties, assets and business affairs of the Division and the Business and the above-mentioned documents, records and information. The Seller will keep the Buyer generally informed as to any material developments pertaining to the Business. Notwithstanding anything in this <u>Section 4.1(c)</u> to the contrary, Seller will not be required to share any competitively sensitive information with Buyer pursuant to this <u>Section 4.1(c)</u>.

(ii) The Seller will retain all books and records relating to the Division in accordance with the Seller's record retention policies as presently in effect. During the five-year period beginning on the Closing Date, the Seller shall not dispose of or permit the disposal of any such books and records not required to be retained under such policies without first giving 60 days' prior written notice to the Buyer offering to surrender the same to the Buyer at the Buyer's expense.

(d) <u>Public Announcements</u>. Except as required by Applicable Law, the Seller shall not, and it shall not permit any Affiliate to, make any initial public announcement in respect of this Agreement or the transactions contemplated hereby without the prior written consent of the Buyer.

- 26 -

(e) Further Actions. Except as otherwise provided in Section 4.3:

(i) The Seller agrees to use all reasonable good faith efforts to take all actions and to do all things necessary, proper or advisable to consummate the transactions contemplated hereby by the Closing Date, including causing the release of any and all Liens (other than Permitted Liens) on or affecting the Assets.

(ii) The Seller shall not, and shall not authorize or permit any of its respective officers, directors, employees, agents or representatives to, propose, announce or enter into any transaction that could reasonably be expected to materially adversely affect the Seller's ability to consummate the transactions contemplated hereby and to fulfill the conditions thereto.

(iii) The Seller will, as promptly as practicable, file or supply, or cause to be filed or supplied, all applications, notifications and information required to be filed or supplied by it pursuant to Applicable Law in connection with this Agreement, the Ancillary Agreements, the sale and Transfer of the Assets pursuant to this Agreement and the consummation of the other transactions contemplated thereby.

(iv) The Seller, as promptly as practicable, will use all commercially reasonable efforts to obtain, or cause to be obtained, all Consents (including all Governmental Approvals and any Consents required under any Contract) necessary to be obtained by it in order to consummate the sale and Transfer of the Assets pursuant to this Agreement and the consummation of the other transactions contemplated hereby.

(v) The Seller will, and will cause its Affiliates to, coordinate and reasonably cooperate with the Buyer in exchanging such information and supplying such assistance as may be reasonably requested by the Buyer in connection with the filings and other actions contemplated by <u>Section 4.2(b)</u>.

(vi) At all times prior to the Closing, the Seller shall promptly notify the Buyer in writing of any fact, condition, event or occurrence that will or may result in the failure of any of the conditions contained in <u>Sections 5.1</u> and <u>5.2</u> to be satisfied, promptly upon the Seller becoming aware of the same.

(vii) The Seller shall, as reasonably requested by the Buyer, deliver to the Buyer's title insurance company any customary affidavits reasonably required by such title insurance company in connection with any title insurance relating to the Owned Real Property proposed to be purchased by Buyer. The Buyer shall be responsible for any and all costs relating to any such purchase of title insurance.

(viii) The Seller shall be responsible for compliance with all applicable requirements of the New Jersey Industrial Site Recovery Act triggered by the transfer of ownership of the South Plainfield New Jersey Facility at Closing.

- 27 -

(f) <u>Further Assurances</u>. Following the Closing, the Seller shall, and shall cause each of its Affiliates to, from time to time, execute and deliver such additional instruments, documents, conveyances or assurances and take such other actions as shall be reasonably necessary, or otherwise reasonably requested by the Buyer, to confirm and assure the rights and obligations provided for in this Agreement and in the Ancillary Agreements, to render effective the consummation of the transactions contemplated hereby and thereby, and to comply with all Applicable Law.

(g) <u>Liability for Transfer Taxes</u>. The Seller shall be responsible for the timely payment of, and shall indemnify and hold harmless the Buyer against, all sales (including bulk sales, if applicable), use, value added, documentary, stamp, gross receipts, registration, transfer, conveyance, excise, license and other similar Taxes, levies, charges and fees, including all interest and penalties thereon and additions thereto, whether disputed or not ("<u>Transfer Taxes</u>"), arising out of or in connection with or attributable to the transactions effected pursuant to this Agreement and the Ancillary Agreements. The party required by Applicable Law to file all necessary Tax Returns relating to such Transfer Taxes shall prepare and file such Tax Returns and Buyer and Seller shall each, and shall each cause their Affiliates to, cooperate in the timely preparation and filing of, and join in the execution of, any such Tax Returns.

(h) <u>Certificates of Tax Authorities</u>. On or before the Closing Date, the Seller shall provide to the Buyer copies of certificates from the appropriate taxing authority stating that no Taxes are due to any state or other taxing authority for which the Buyer would reasonably be expected to have liability to withhold or pay Taxes with respect to the Transfer of the Assets or the Business.

(i) <u>Use of Intellectual Property</u>. Except for the Licensed IP, from and after the Closing, none of the Seller or any of its Affiliates shall have any rights or interest in and to the Business Intellectual Property Assets, and none of the Seller or any of its Affiliates will contest the exclusive ownership or validity of any rights of the Buyer in or to the Business Intellectual Property Assets. From and after the Closing, neither the Seller nor any of its Affiliates shall use any of the Business Intellectual Property Assets (other than the Licensed IP, subject to the terms of the License Agreement) for any purpose.

(j) <u>Insurance and Insurance Proceeds</u>. The Seller shall keep, or cause to be kept, all insurance policies of the Seller relating to the Business, or equivalent replacements therefor, in full force and effect through the Closing Date. To the extent that the Seller has in force any policies of property and casualty or similar insurance insuring any of the Assets, any proceeds of insurance payable (in excess of any deductible, retention or self-insurance amount) in respect of any event that occurs from and after the date of this Agreement and on or before the Closing Date, to the extent that the proceeds pertain to properties or assets that would have constituted Assets had they been in existence on the Closing Date, shall be received by the Seller in trust for the Buyer and, to the extent the Assets to which the proceeds pertain have not been replaced, repaired or restored, shall be paid over to the Buyer at the Closing or at any such later time at which the Seller receives such proceeds. If no proceeds have been received before the Closing, the Seller shall assign its claims thereto to the Buyer at the Closing.

- 28 -

(k) <u>Notice of Certain Events</u>. After the execution of this Agreement but prior to the Closing, the Seller shall, and shall cause its representatives to, promptly notify the Buyer in writing in reasonable detail of:

(i) any written notice or other written communication from any Person alleging that the Consent of such Person is or may be required in connection with the transactions contemplated by this Agreement; or

(ii) the damage or destruction by fire or other casualty of any material asset of the Business or any material part thereof or in the event that any material asset of the Business or material part thereof becomes the subject of any proceeding or, to the knowledge of the Seller, threatened proceeding for the taking thereof or any part thereof or of any right relating thereto by condemnation, eminent domain or other similar governmental action.

(1) <u>Monthly Financial Reports</u>. As promptly as practicable and in any event no later than fifteen (15) calendar days after the end of each calendar month ending after the date hereof and before the Closing Date, the Seller will deliver to the Buyer copies of monthly profit and loss statements with respect to the Business and a balance sheet and statement of cash flows for the Business.

(m) <u>Competitive Activity; Confidentiality; Non-Solicitation</u>. On the Closing Date, the Buyer, the Seller and the Principal Stockholders shall enter into the Non-Competition Agreement in the form attached as <u>Exhibit B</u> (the "<u>Non-Competition Agreement</u>").

(n) <u>No-Shop</u>. From and after the date of this Agreement and continuing until the Closing or the earlier termination of this Agreement pursuant to <u>Section 7.1</u>, each of the Seller and the Principal Stockholders hereby covenants and agrees that it, he or she will not, and will not authorize or permit any officer, director, manager, employee or agent of the Seller to, or authorize or permit any investment banker, attorney, accountant or other representative or agent retained by the Seller or any Principal Stockholder to, directly or indirectly, without the written consent of the Buyer (i) solicit, negotiate, or enter into material discussions regarding, any possible Business Combination, with any party other than the Buyer or (ii) following receipt of an unsolicited proposal for a Business Combination, approve, endorse or recommend such proposal. From and after the date hereof, and continuing to the Closing or earlier termination of this Agreement pursuant to <u>Section 7.1</u>, each of the Principal Stockholders covenants and agrees that it will vote for, approve and otherwise support the consummation of the transaction contemplated by this Agreement, and will not vote, for approve or otherwise support any Business Combination with any party other than the Buyer.

(o) <u>Accounts Receivable</u>. After the Closing, the Seller shall promptly remit to the Buyer all amounts received by the Seller in respect of accounts receivable that constitute Assets.

- 29 -

4.2 Covenants of the Buyer.

(a) <u>Public Announcements</u>. Prior to the Closing, except as required by Applicable Law, the Buyer shall not, and shall not permit any Affiliate to, make any public announcement in respect of this Agreement or the transactions contemplated hereby without the prior written consent of the Seller.

(b) <u>Confidentiality</u>. The Buyer will keep confidential and will cause its Affiliates to keep confidential all non-public information relating to the ongoing business of Seller, except for disclosures required by Applicable Law or administrative process (including disclosures required in Tax Returns or in other required filings with Governmental Authorities), <u>provided</u> that the disclosing party shall provide the non-disclosing party with reasonable notice of any required disclosure, to the extent practicable, and except for information that becomes public other than as a result of a breach of this <u>Section 4.2(b)</u>. In addition to the foregoing, in between the execution of this Agreement and the Closing Date, the Buyer will keep confidential and will cause its Affiliates to keep confidential all non-public information relating to the ongoing business of Seller that is disclosed by Seller regarding its customers, suppliers and the Business set forth in the Disclosure Schedules and make such information available only to those individuals who have a need to know that are set forth on <u>Schedule 4.2(b)</u>, except for disclosures required by Applicable Law or administrative process (including disclosures required disclosure, to the extent practicable, and except for information that becomes public other than as a result of a breach of this <u>Section 4.2(b)</u>. If this Agreement is terminated for any reason, Buyer shall cause all information referenced in the immediately preceding sentence to be returned to Seller within 10 Business Days of such termination along with a certification of an authorized officer of the Buyer as to compliance with this covenant.

(c) Further Actions. Except as otherwise provided in Section 4.3,

(i) The Buyer agrees to use all reasonable good faith efforts to take all actions and to do all things necessary, proper or advisable to consummate the transactions contemplated hereby by the Closing Date.

(ii) The Buyer shall not, and shall not authorize or permit any of its respective officers, directors, employees, agents or representatives to, announce or enter into any transaction to acquire or combine with, by merger or by purchase of stock, assets or otherwise, a provider of contract ethylene oxide sterilization services for the sterilization of medical products, devices, supplies, and equipment, which transaction could reasonably be expected to adversely affect the likelihood that the material Governmental Approvals required to consummate the transactions contemplated hereby will be received or delayed.

(iii) The Buyer will, as promptly as practicable, file or supply, or cause to be filed or supplied, all applications, notifications and information required to be filed or supplied by the Buyer pursuant to Applicable Law in connection with this Agreement, the Ancillary Agreements, the Buyer's acquisition of the Assets pursuant to this Agreement and the consummation of the other transactions contemplated hereby and thereby.

- 30 -

(iv) The Buyer will coordinate and cooperate with the Seller in exchanging such information and supplying such reasonable assistance as may be reasonably requested by the Seller in connection with the filings and other actions contemplated by <u>Section 4.1(e)</u>.

(v) At all times prior to the Closing, the Buyer shall promptly notify the Seller in writing of any fact, condition, event or occurrence that will or may result in he failure of any of the conditions contained in <u>Sections 5.1</u> and <u>5.3</u> to be satisfied, promptly upon becoming aware of the same.

(vi) After the Closing, the Buyer will promptly remit to the Seller all amounts received by Buyer that represent Excluded Assets.

(d) <u>Further Assurances</u>. Following the Closing, the Buyer shall, and shall cause its Affiliates to, from time to time, execute and deliver such additional instruments, documents, conveyances or assurances and take such other actions as shall be necessary, or otherwise reasonably requested by the Seller, to confirm and assure the rights and obligations provided for in this Agreement and in the Ancillary Agreements and render effective the consummation of the transactions contemplated hereby and thereby.

(e) <u>Use of Licensed IP and Excluded Non-Licensed IP</u>. From and after the Closing, none of the Buyer or any of its Affiliates will use the Excluded Non-Licensed IP or the Licensed IP (except, in the case of the Licensed IP, pursuant to the License Agreement), and none of the Buyer or its Affiliates will contest the ownership or validity of any rights of the Seller or its Affiliates in the Excluded Non-Licensed IP.

4.3 <u>Antitrust Notification</u>. Each of the Buyer and Seller will, if not done prior to the execution of this Agreement, as promptly as practicable, but in no event later than three Business Days following the execution and delivery of this Agreement, file with the United States Federal Trade Commission (the "<u>FTC</u>") and the United States Department of Justice (the "<u>DOJ</u>") the notification and report form required for the transactions contemplated hereby and any supplemental information required in connection therewith pursuant to the HSR Act. Each party hereto represents and warrants that such notification and report form and all such supplemental information submitted by such party or its ultimate parent, and any additional supplemental information filed by such party or its ultimate parent after the date of the original filing, will be in substantial compliance with the requirements of the HSR Act. Each of the Buyer and the Seller shall furnish to the other such necessary information (subject to redaction as reasonably necessary to protect competitively sensitive confidential business information) and reasonable assistance as the other may request in connection with its preparation of any filing or submission that is necessary under the HSR Act. The Buyer and the Seller shall keep each other apprised of the status of any communications with, and inquiries or requests for information from, the FTC, the DOJ or any state attorneys general, and shall use their reasonable best efforts to comply promptly with any such inquiry or request. Each of the Buyer and the Seller will use its reasonable best efforts to cause the expiration or early termination of the waiting period required

- 31 -

under the HSR Act as a condition to the purchase and sale of the Assets and shall use its reasonable best efforts to resolve any antitrust concerns raised by, and defend against any action to enjoin the sale of the Assets to the Buyer brought by the FTC, the DOJ or any state attorneys general. Each of the Buyer and the Seller will respond promptly under the circumstances to any requests for additional information by any Governmental Authority in connection with the transactions contemplated by this Agreement, including promptly filing a response to a "second request" from an applicable Governmental Authority in connection with its review of the transactions contemplated by this Agreement pursuant to the HSR Act. Nothing stated in this Agreement shall require the Buyer to agree or commit to divest, hold separate, offer for sale, abandon, limit its operation of or take similar action with respect to any assets of the Buyer or any of the Assets, or permit the Seller to agree or commit to take any such action with respect to the Assets in connection with or to obtain termination of the waiting period under the HSR Act or otherwise.

4.4 <u>Change Notices</u>. During the period from the date of this Agreement to the Closing Date or the earlier termination of this Agreement, the Seller shall advise the Buyer in writing of (i) the occurrence, or non-occurrence, of any event or development which has caused, or could reasonably be expected to cause, any representation or warranty made by the Seller to be untrue or inaccurate in any material respect at any time after the date of this Agreement and prior to the Closing Date; and (ii) any material failure by the Seller to comply with or satisfy any of its covenants or agreements hereunder. If any event or development (i) arises and occurs in the Ordinary Course of Business after the date hereof, (ii) was not within the Seller's Knowledge as of the date of this Agreement, (iii) is of a nature such that Schedules 3.1(f) (Taxes), 3.1(g) (Absence of Changes), 3.1(i) (Compliance with Laws and Governmental Approvals), 3.1(l) (Contracts), 3.1(p) (Intellectual Property), 3.1(q) (Insurance), 3.1(t) (Employees, Labor Matters), or 3.1(u) (Employee Benefit Plans and Related Matters) to this Agreement would need to be modified solely in order to make the representations and warranties in Section 3.1(f)(ii), Section 3.1(g)(xv), the first sentence of Section 3.1(i)(ii), Section 3.1(l)(i), the first sentence of Section 3.1(p)(i), Section 3.1(q), Section 3.1(u) true and accurate in all material respects, and (iv) does not represent, individually or in the aggregate with any other events or developments set forth in Change Notices pursuant to this <u>Section 4.4</u>, a Material Adverse Effect, the Seller shall promptly deliver to the Buyer written notice that sets forth the manner in which such schedules would need to be so modified (a "<u>Change Notice</u>"). The Disclosure Schedules shall be deemed amended as set forth in any Change Notices.

4.5 <u>Consent Decree</u>. Prior to the Closing, the Seller shall continue to negotiate the form of the Consent Decree with the United States Environmental Protection Agency. After the Closing, the Buyer and the Seller shall reasonably cooperate with each other, and shall each participate, in the negotiation and execution of the Consent Decree with the United States Environmental Protection Agency. From the date hereof until the Consent Decree is entered as a final judgment in the applicable U.S. District Court, each party shall promptly notify the other of any developments in, and shall promptly provide the other party with copies of any correspondence or other documents received in connection with, such negotiations, and shall reasonably consult with the other party in connection therewith. After the Closing Date, the Seller shall not execute or otherwise enter into any Consent Decree with respect to the Facilities or the Business without the prior written consent of the Buyer, which consent shall not be unreasonably withheld, delayed or denied, and the Buyer shall comply with the provisions of

- 32 -

Section 5 of the Consent Decree concerning the transfer of ownership of the Facilities to the extent provided elsewhere in this Agreement. The Seller shall promptly reimburse the Buyer for all of the documented reasonable out-of-pocket costs, expenses and fees (including reasonable attorneys' and other professionals' fees) actually incurred by the Buyer in connection with the obligations assumed by the Buyer pursuant to <u>Section 1.3(c)</u> hereof under Appendix B of the Consent Decree, but only as to the cost of the compliance audit work plans, compliance audits and compliance audit reports and as to violations of environmental compliance law and regulations in existence at the Real Property on or before the date of Closing which remain unresolved on the date of Closing (which shall include those evidenced by the compliance audits performed by the Buyer, a copy of each of which is attached hereto as Exhibit H) which remain unresolved on the date of Closing; provided, however, that Buyer shall only be liable for any stipulated penalties attributable to the Buyer's failure to meet compliance deadlines for such obligations after the date of Closing and for all costs, expenses, fees and penalties arising out of violations of Environmental Liabilities and Costs for Seller, shall be done as follows: at Buyer's option, (1) Buyer and Seller can mutually agree upon a third party consultant/firm; or (2) Buyer may submit written bids to Seller from at least three consultants/firms and Seller shall chose therefrom; provided that if Seller does not chose a consultant/firm within 30 days after the bids are submitted to it, Buyer may use any of the consultants/firms whose bid was submitted for the project covered by such bid.

4.6 <u>Environmental Permits</u>. Buyer shall take any and all commercially reasonable actions to procure any consents from Governmental Authorities to the transfer of the Environmental Permits for each Facility set forth on <u>Schedule 3.1(i)(ii)</u>, as may be required by the relevant Governmental Authority. Seller shall assist and shall cooperate with Buyer in Buyer's efforts with regard to the Environmental Permits to the extent commercially reasonable.

ARTICLE V CLOSING CONDITIONS

5.1 <u>Conditions to Obligations of Each Party</u>. The obligations of the parties to consummate the transactions contemplated hereby shall be subject to the fulfillment on or prior to the Closing Date of the following conditions:

(a) <u>HSR Act Notification</u>. In respect of the notifications of the Buyer and the Seller pursuant to the HSR Act, if any, the applicable waiting period and any extensions thereof shall have expired or been terminated.

(b) <u>No Injunction, etc</u>. Consummation of the transactions contemplated hereby shall not have been restrained, enjoined or otherwise prohibited by any Applicable Law, including any Order. No Governmental Authority shall have determined any Applicable Law to make illegal the consummation of the transactions contemplated hereby or by the Ancillary Agreements, and no proceeding with respect to the application of any such Applicable Law to such effect shall be pending.

5.2 <u>Conditions to Obligations of the Buyer</u>. The obligations of the Buyer to consummate the transactions contemplated hereby shall be subject to the fulfillment (or waiver

- 33 -

by the Buyer) on or prior to the Closing Date of the following additional conditions, which the Seller agrees to use reasonable good faith efforts to cause to be fulfilled:

(a) <u>Representations</u>, <u>Performance</u>. The representations and warranties of the Seller contained in this Agreement and in the Ancillary Agreements taken together with the Disclosure Schedules and any supplements to the Disclosure Schedules permitted pursuant to <u>Section 4.4</u> hereof, (i) shall be true and correct in all respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty without any materiality qualification) on and as of the Closing Date with the same effect as though made on and as of the Closing Date. The Seller shall have duly performed and complied in all material respects with all covenants, agreements and conditions required by this Agreement and each of the Ancillary Agreements to be performed or complied with by it prior to or on the Closing Date. The Seller shall have delivered to the Buyer a certificate, dated the Closing Date and signed by its duly

(b) <u>Consents</u>. The Seller shall have obtained and shall have delivered to the Buyer copies of (i) all Consents of Governmental Authorities required to be obtained by the Seller in connection with the execution and delivery of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby and (ii) all Consents required under any Contract (other than customer Contracts) listed, or required to be listed, on <u>Schedule 3.1(1)(i)</u> from third parties other than Governmental Authorities, except with regard to both (i) and (ii) above, to the extent such Consent is indicated on <u>Schedule 3.1(c)</u> as not being required as a condition to the obligations of the Buyer to consummate the transactions contemplated hereby.

(c) <u>Consents to Assignment of Customer Contracts</u>. Within 60 days following the execution of this Agreement, Seller shall deliver a true and correct notice to Buyer certifying that, with respect to customer Contracts representing in the aggregate at least 60% of the total revenues of the Business for the nine month period ending September 30, 2004, either (1) a Consent to assign such Contract upon the Closing has been obtained (in a form reasonably acceptable to the Buyer), or (2) such Contract requires no Consent to be assigned other than the giving of notice by the Seller, which notice does not give rise to any rights of termination or other rights in favor of the customer. Such notice shall indicate the percentage of the total revenues of the Business for the nine month period ending September 30, 2004 for which such Consents have been obtained or for which only such notices are required. If this condition is not satisfied, Buyer shall, within five Business Days of its receipt of such notice (or, if such notice is not given, within 65 days following the execution of this Agreement) either: (i) waive the condition; or (ii) terminate this Agreement pursuant to Section 7.1(a). The Seller shall be deemed to have agreed to the Buyer's termination pursuant to clause (ii) of the immediately preceding sentence.

- 34 -

(d) <u>No Material Adverse Effect</u>. No event, occurrence, fact, condition, change, development or effect shall have occurred, exist or come to exist since the date of this Agreement that, individually or in the aggregate, has constituted or resulted in, or would reasonably be expected to constitute or result in, a Material Adverse Effect.

(e) <u>Ancillary Agreements</u>. The Seller shall have delivered to the Buyer on the Closing Date executed counterparts of (i) a license agreement substantially in the form attached hereto as <u>Exhibit C</u> (the "<u>License Agreement</u>"), (ii) an equipment maintenance and transition services agreement substantially in the form attached hereto as <u>Exhibit D</u> (the "<u>Maintenance Agreement</u>"), (iii) an equipment supply agreement substantially in the form attached hereto as <u>Exhibit D</u> (the "<u>Maintenance Agreement</u>"), (iii) an equipment supply agreement substantially in the form attached hereto as <u>Exhibit E</u> (the "<u>Supply Agreement</u>"), and (iv) the Non-Competition Agreement.

(f) <u>Corporate Proceedings</u>. All corporate and other proceedings of the Seller in connection with this Agreement and the Ancillary Agreements and the transactions contemplated hereby and thereby, and all documents and instruments incident thereto, shall be reasonably satisfactory in substance and form to the Buyer and its counsel, and the Buyer and its counsel shall have received all such documents and instruments, or copies thereof, certified if requested at least three Business Days prior to the Closing, as may be reasonably requested.

(g) <u>Transfer Documents</u>. The Seller shall have delivered to the Buyer at the Closing all documents, certificates and agreements necessary to Transfer to the Buyer good and marketable, as the case may be, title to the Assets, free and clear of any and all Liens thereon, other than Permitted Liens, including:

(i) A bill of sale, assignment and general conveyance, substantially in the form of <u>Exhibit F</u> hereto (the "<u>Bill of Sale</u>"), dated the Closing Date, with respect to the Assets (other than any Asset to be Transferred pursuant to any of the instruments referred to in any other clause of this <u>Section 5.2(g)</u>);

(ii) An Assignment and Assumption Agreement, substantially in the form of <u>Exhibit G</u> hereto (the "<u>Assumption Agreement</u>"), assignments of all Business Intellectual Property Assets (other than the Licensed IP) in form and substance reasonably satisfactory to the parties, and assignments of any other agreements and instruments constituting Assets in form and substance reasonably satisfactory to the parties, dated the Closing Date, assigning to the Buyer all of the Seller's right, title and interest therein and thereto, with any required Consent endorsed thereon;

(iii) A warranty deed, dated as of the Closing Date, with respect to that certain parcel of Owned Real Property located in Rhode Island, and a special warranty deed, dated as of the Closing Date, with respect to that certain parcel of Owned Real Property located in New Jersey, each in form and substance substantially equivalent to the form of deed provided to the Seller upon the Seller's acquisition of such properties (true and complete copies of which have been provided to the Buyer prior to the date hereof), together with any necessary transfer declarations or other filings;

- 35 -

(iv) An assignment of leases, dated as of the Closing Date, with respect to each Lease, together with any necessary transfer declarations or other filings, each in form and substance reasonably satisfactory to the Buyer; and

(v) Certificates of title to all motor vehicles included in the Assets to be Transferred to the Buyer hereunder, duly endorsed for Transfer to the Buyer as of the Closing Date.

(h) <u>Estoppels</u>. The Buyer shall also have received estoppel certificates addressed to the Buyer from the lessor of each Lease, dated within 30 days of the Closing Date, identifying the Lease documents and any amendments thereto, stating that the Lease is in full force and effect and, to the actual knowledge of the lessor, that the tenant is not in default under the Lease and no event has occurred that, with notice or lapse of time or both, would constitute a default by the tenant under the Lease and containing any other information reasonably requested by the Buyer.

(i) <u>FIRPTA Certificate</u>. The Buyer shall have received a certificate of the Seller, dated as of the Closing Date and sworn to under penalty of perjury, setting forth the name, address and federal tax identification number of the Seller and stating that the Seller is not a "foreign person" within the meaning of Section 1445 of the Code, such certificate to be in the form set forth in the Treasury Regulations thereunder and reasonably satisfactory to Buyer.

5.3 <u>Conditions to Obligations of the Seller</u>. The obligation of the Seller to consummate the transactions contemplated hereby shall be subject to the fulfillment (or waiver by the Seller), on or prior to the Closing Date, of the following additional conditions, which the Buyer agrees to use reasonable good faith efforts to cause to be fulfilled:

(a) <u>Representations; Performance, etc</u>. The representations and warranties of the Buyer contained in this Agreement and the Ancillary Agreements (i) shall be true and correct in all respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty without any materiality qualification) at and as of the date hereof and (ii) shall be repeated and shall be true and correct in all respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty without any materiality qualification) on and as of the Closing Date with the same effect as though made at and as of such time. The Buyer shall have duly performed and complied in all material respects with all agreements, covenants and conditions required by this Agreement and the Ancillary Agreements to be performed or complied with by it prior to or on the Closing Date. The Buyer shall have delivered to the Seller a certificate, dated the Closing Date and signed by its duly authorized officer, to the foregoing effect.

(b) <u>Assumption Agreement</u>. The Seller shall have received from the Buyer the executed Assumption Agreement.

(c) <u>Corporate Proceedings</u>. All corporate proceedings of the Buyer in connection with this Agreement and the Ancillary Agreements and the transactions

- 36 -

contemplated hereby and thereby, and all documents and instruments incident thereto, shall be reasonably satisfactory in substance and form to the Seller, and its counsel, and the Seller and its counsel shall have received all such documents and instruments, or copies thereof, certified if requested, at least three Business Days prior to the Closing, as may be reasonably requested.

(d) <u>Consents</u>. The Buyer shall have obtained and shall have delivered to Seller copies of all Consents from Governmental Authorities required to be obtained by the Buyer in connection with the execution and delivery and the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements.

(e) <u>Ancillary Agreements</u>. The Buyer shall have delivered to the Seller on the Closing Date executed counterparts of (i) the License Agreement, (ii) the Maintenance Agreement, (iii) the Supply Agreement, and (iv) the Non-Competition Agreement.

(f) <u>Purchase Price</u>. The Buyer shall have delivered to the Seller on the Closing Date the Purchase Price less the Holdback Amount and the Chamber Holdback in immediately available funds wire transferred to an account designated by the Seller.

ARTICLE VI EMPLOYEE MATTERS

6.1 <u>Employment of the Seller's Employees</u>. The Seller will use all reasonable efforts to cause the employees listed on <u>Schedule 6.1</u> to make available their employment services to the Buyer. Buyer commits to offer employment to each of the persons set forth on <u>Schedule 6.1</u> who continue to be employed by the Seller immediately prior to the Closing Date upon terms and conditions of employment no less favorable than those provided to similarly situated employees of the Buyer and at an equal or greater base salary to that which they currently receive.

6.2 Employee and Related Matters.

(a) The Seller shall provide to the Buyer within a reasonable period prior to the Closing (and again on the Closing Date) a true, complete and accurate list of the projected employees of the Seller actively employed by the Business (the "<u>Current Employees</u>"). Current Employees who, on or immediately after the Closing Date, accept the Buyer's offer of employment and become employees of the Buyer shall be referred to herein as the "<u>Rehired Employees</u>."

(b) The Buyer shall not assume any Plan or any liability or obligation thereunder.

(c) Notwithstanding anything contained herein to the contrary, the Seller shall perform all of its obligations under the health care continuation requirements of Part 6 of Subtitle B of Title I of ERISA, Section 4980B of the Code or any similar state law ("<u>COBRA</u>") with respect to its employees, former employees and their covered dependents, whether or not such employees accept employment with the Buyer.

- 37 -

(d) Subject to all applicable legal restrictions, Buyer and Seller shall provide each other, in a timely manner, with any information which the other may reasonably request with respect to any employee of Seller or, after the Closing, any Rehired Employee, his employment with and compensation from Seller or Buyer, or rights or benefits under any Plan or any personnel policy of Seller or Buyer relating to the Business.

(e) As soon as practicable after the Closing Date, the Buyer shall cause all Rehired Employees to be eligible to participate in the retirement and welfare benefit plans, arrangements and programs maintained by the Buyer, subject to the terms and conditions of each such plan. The Buyer shall waive any pre-existing condition limitation under the Buyer's health plan for any condition of a Rehired Employee or an eligible beneficiary of a Rehired Employee that was covered under a health plan of the Seller as of the Closing Date. The Buyer shall recognize the service of each Rehired Employee for the Seller before the Closing Date for determining eligibility and vesting under the Buyer's benefit plans and shall give each Rehired Employee credit for service with the Seller for such purposes under the Buyer's benefit plans.

ARTICLE VII TERMINATION

7.1 Termination. This Agreement may be terminated at any time prior to the Closing Date:

(a) by the written agreement of the Buyer and the Seller;

(b) by either the Seller or the Buyer by written notice to the other party if the transactions contemplated hereby shall not have been consummated pursuant hereto by 5:00 p.m. Providence, Rhode Island time on December 31, 2004, unless (i) such date shall be extended by the mutual written consent of the Seller and the Buyer or (ii) the transactions contemplated hereby have not been consummated as the result of a failure of any of the conditions of the obligations of the parties set forth in <u>Article V</u> to be met, which failure is the result of circumstances outside of the reasonable control of the party not seeking to terminate this Agreement, in which case this Agreement may be terminated by either the Seller or the Buyer by written notice to the other party if the transactions contemplated hereby shall not have been consummated pursuant hereto by 5:00 p.m., Providence, Rhode Island time on July 8, 2005 (such date, if and as extended, the "<u>Drop Dead Date</u>");

(c) by the Buyer by written notice to the Seller if (i) the representations and warranties of the Seller shall not have been true and correct in all respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty without any materiality qualification) as of the date when made, or (ii) any of the conditions set forth in <u>Sections 5.1</u> or <u>5.2</u> shall not have been satisfied as of the day before the Drop Dead Date, or satisfaction of any such condition becomes impossible, unless such failure shall be due to the failure of the Buyer to perform or comply with any of the covenants, agreements or conditions hereof to be performed or complied with by it prior to the Closing; and

- 38 -

(d) by the Seller by written notice to the Buyer if (i) the representations and warranties of the Buyer shall not have been true and correct in all respects (in the case of any representation or warranty containing any materiality qualification) or in all material respects (in the case of any representation or warranty without any materiality qualification) as of the date when made or (ii) any of the conditions set forth in <u>Section 5.1</u> or <u>5.3</u> shall not have been satisfied as of the date before the Drop Dead Date, or satisfaction of any such condition becomes impossible, unless such failure be due to the failure of the Seller to perform or comply with any of the covenants, agreements or conditions hereof to be performed or complied with by it prior to the Closing.

7.2 <u>Effect of Termination</u>. In the event of the termination of this Agreement pursuant to the provisions of <u>Section 7.1</u>, this Agreement shall become void and have no effect, without any liability to any Person in respect hereof or of the transactions contemplated hereby on the part of any party hereto, or any of its directors, officers, employees, agents, consultants, representatives, advisers, stockholders or Affiliates, except as specified in <u>Section 10.1</u> with respect to Transaction Expenses and except for any liability resulting from such party's willful or intentional breach of this Agreement.

ARTICLE VIII DEFINITIONS

8.1 <u>Definition of Certain Terms</u>. The terms defined in this <u>Section 8.1</u>, whenever used in this Agreement (including in the Schedules), shall have the respective meanings indicated below for all purposes of this Agreement. All references herein to a Section, Article or Schedule are to a Section, Article or Schedule of or to this Agreement, unless otherwise indicated.

Accounts Receivable has the meaning set forth in Section 3.1(m).

Additional Chambers means those two certain 28 pallet chambers and all ancillary equipment related thereto, and configured in a manner that is equivalent to the configuration of similar equipment that is currently installed at the Seller's San Diego Facility, that are in the process of being manufactured by the Seller as of the date hereof.

<u>Affiliate</u> of a Person means a Person that directly or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, the first Person. "<u>Control</u>" (including the terms "<u>controlled by</u>" and "<u>under common control with</u>") means the possession, directly or indirectly, of the power to direct or cause the direction of the management policies of a person, whether through the ownership of voting securities, by Contract or otherwise, as trustee or executor, or otherwise.

Agreement means this Asset Purchase Agreement, including the Schedules and Exhibits hereto.

<u>Ancillary Agreements</u> means the License Agreement, the Maintenance Agreement, the Supply Agreement, Assumption Agreement, the Non-Compete Agreement, the Bill of Sale and such other agreements entered into between the Seller and the Buyer to consummate the transactions contemplated hereby.

- 39 -

<u>Applicable Law</u> means all applicable provisions of all (i) constitutions, treaties, statutes, laws (including the common law), rules, regulations, ordinances, codes or orders of any Governmental Authority, (ii) Governmental Approvals and (iii) Orders.

Arbitration Firm has the meaning set forth in Section 2.3(b).

Assets has the meaning set forth in Section 1.1.

Assumed Contracts has the meaning set forth in Section 1.1(c).

Assumption Agreement has the meaning set forth in Section 5.2(g).

Assumed Liabilities has the meaning set forth in Section 1.3(a).

Balance Sheet Date has the meaning set forth in Section 3.1(g).

<u>Benefit Liabilities</u> means all past, present and future obligations and liabilities of the Seller arising out of Applicable Law (i) with respect to each Plan and (ii) with respect to all employees and former employees of the Seller in connection with any event commencing, occurring or failing to occur on or prior to the Closing Date.

Bill of Sale has the meaning set forth in Section 5.2(g).

<u>Business Combination</u> means any merger, reorganization, share exchange, consolidation, business combination, recapitalization, liquidation, dissolution or similar transaction, or any sale or other disposition (or series of sales or dispositions) of all or any significant portion of the assets or 50% or more of the outstanding equity securities in the aggregate of, the Seller or other sale of the Business.

Business has the meaning set forth in the Recitals.

Business Day means a day other than a Saturday, Sunday or other day on which commercial banks in Cleveland, Ohio are authorized or required to close.

Business Intellectual Property Assets means all of the Intellectual Property are related rights to be Transferred by Seller to Buyer pursuant to Section <u>1.1</u> and the Licensed IP.

Buyer has the meaning set forth in the Preamble.

Buyer Indemnitees has the meaning set forth in Section 9.1.

<u>Buyer Material Adverse Effect</u> means any event, occurrence, fact, condition, change or effect that has or would reasonably be expected to have a materially adverse effect on the business, operations, results of operations, condition (financial or otherwise), properties (including intangible properties), assets (including intangible assets) or liabilities of the Buyer.

- 40 -

CERCLA means the Comprehensive Environmental Response, Compensation and Liability Act, as amended, 42 U.S.C. 9601 et seq.

<u>Chamber Holdback</u> means (a) zero if both Additional Chambers are Transferred to the Buyer at the Closing in accordance with <u>Section 1.1</u> and <u>Section 1.5</u>, (b) \$650,000 if one Additional Chamber has been transferred to the Buyer at the Closing in accordance with <u>Section 1.1</u> and <u>Section 1.5</u> as of the Closing Date, and (c) \$1,300,000 if no Additional Chambers are Transferred to the Buyer in accordance with <u>Section 1.1</u> and <u>Section 1.5</u> as of the Closing Date.

Change Notice has the meaning set forth in Section 4.4.

<u>Claims Notice</u> has the meaning set forth in <u>Section 9.3(a)</u>.

Closing has the meaning set forth in Section 2.1.

<u>Closing Date</u> has the meaning set forth in <u>Section 2.1</u>.

Closing Working Capital has the meaning set forth in Section 2.3(a).

COBRA has the meaning set forth in Section 6.2(c).

Code means the Internal Revenue Code of 1986, as amended.

<u>Commissioned</u> means a mutual determination by the parties that (i) the subject Additional Chamber has been completed and fully installed, (ii) the installation qualifications and operational qualifications of such Additional Chamber are acceptable based on recognized industry standards, (iii) such Additional Chamber is configured to perform in a manner equivalent to the other chambers in operation at the San Diego Facility, and (iv) such Additional Chamber is ready and acceptable for performance qualification work for medical device providers consistent with policies and procedures practiced by Cosmed of California.

<u>Consent</u> means any consent, approval, authorization, waiver, permit, franchise, concession, agreement, license, exemption or order of, registration, certificate, declaration, application or filing with, or report or notice to, any Person, including, but not limited to, any Governmental Authority.

<u>Consent Decree</u> means that certain draft Consent Decree between the Seller and the United States Government, acting on behalf of the Environmental Protection Agency.

Contracts has the meaning set forth in Section 1.1(c).

<u>Controlled Group</u> as defined in <u>Section 3.1(u)</u>.

Covered Returns has the meaning set forth in Section 3.1(f).

Covered Taxes has the meaning set forth in Section 3.1(f).

- 41 -

Current Employees has the meaning set forth in Section 6.2(a).

Disclosure Schedules has the meaning set forth in the first paragraph of Section 3.1.

Division has the meaning set forth in the Recitals.

DOJ has the meaning set forth in Section 4.3.

<u>Drop Dead Date</u> has the meaning set forth in Section 7.1(b).

<u>Environment</u> means soil, surface waters, groundwater, land, stream sediments, surface or subsurface strata, ambient air, indoor air or indoor air quality, including any material or substance used in the physical structure of any building or improvement.

Environmental Laws means all Applicable Laws relating to the protection of the Environment, to human health and safety, or to any emission, discharge, generation, processing, storage, holding, abatement, existence, Release, threatened Release or transportation of any Hazardous Substances, including (i) all requirements pertaining to reporting, licensing, permitting, investigation or remediation of emissions, discharges, Releases or threatened Releases of Hazardous Substances into the Environment or relating to the manufacture, processing, distribution, use, sale, treatment, receipt, storage, disposal, transport or handling of Hazardous Substances, and (ii) all other requirements pertaining to the protection of the health and safety of employees or the public.

Environmental Liabilities and Costs means all Losses, whether direct or indirect, known or unknown, current or potential, past, present or future, imposed by, under or pursuant to Environmental Laws, including all Losses and oversight costs of Governmental Authorities related to Remedial Actions, and all fees, disbursements and expenses of counsel, experts, personnel and consultants based on, arising out of or otherwise in respect of: (i) the ownership or operation of the Business, the Real Property or any other real properties, assets, equipment or facilities, by the Seller, or any of its predecessors or Affiliates on or prior to the Closing Date; (ii) the environmental conditions existing on or prior to the Closing Date on, under, above, or about any Real Property or any other real properties, assets, equipment or facilities currently or previously owned, leased or operated by the Seller, or any of its predecessors or Affiliates; (iii) the treatment, storage disposal or arrangement for disposal at, upon or to any location of Hazardous Substances generated by the Seller or any of its Affiliates in connection with the operation of the Business on or prior to the Closing Date; and (iv) expenditures necessary to address noncompliance with any and all requirements of Environmental Laws which noncompliance first existed on or prior to the Closing Date at any Real Property or in connection with any aspect of the Business, including all Environmental Permits issued under or pursuant to such Environmental Laws.

<u>Environmental Permits</u> means any permit, license, registration, consent, order, administrative consent order, certificate, approval, application or other authorization necessary for the lawful conduct of the Business as currently conducted or previously conducted under any Environmental Law.

- 42 -

ERISA means the Employee Retirement Income Security Act of 1974, as amended.

Excluded Assets has the meaning set forth in Section 1.2.

Excluded Liabilities has the meaning set forth in Section 1.4.

Excluded Non-Licensed IP has the meaning set forth in Section 1.2(h).

<u>Facilities</u> means the following facilities operated by the Seller: Coventry, Rhode Island, (d/b/a Cosmed of Rhode Island); South Plainfield, New Jersey (d/b/a Cosmed of New Jersey); Waukegan, Illinois (d/b/a Cosmed of Illinois); Grand Prairie, Texas (d/b/a Cosmed of Texas); and San Diego, California (d/b/a Cosmed of California).

<u>Financial Statements</u> has the meaning set forth in <u>Section 3.1(d)</u>.

FTC has the meaning set forth in Section 4.3.

GAAP has the meaning set forth in Section 2.3(a).

General Enforceability Exceptions has the meaning set forth in Section 3.1(a).

<u>Governmental Approval</u> means any permit, license, registration, consent, order, administrative consent order, certificate, approval, application or other authorization issued by any Governmental Authority, including any Environmental Permit.

<u>Governmental Authority</u> means any nation or government, any state or other political subdivision thereof, any governmental entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, including any government authority, agency, department, board, commission or instrumentality of the United States, any State of the United States or any political subdivision thereof, and any tribunal or arbitrator(s) of competent jurisdiction, and any self-regulatory organization.

<u>Guarantee</u> means, for any Person, any obligation, contingent or otherwise, of such Person directly or indirectly guaranteeing or otherwise supporting in whole or in part the payment of any Indebtedness or other obligation of any other Person and, without limiting the generality of the foregoing, any obligation, direct or indirect, contingent or otherwise, of such Person (a) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation of such other Person (whether arising by virtue of partnership arrangements, by agreement to keep-well, to purchase assets, goods, securities or services, to take-or-pay, or to maintain financial statement conditions or otherwise) or (b) entered into for the purpose of assuring in any other manner the obligee of such Indebtedness or other obligations of the payment of such Indebtedness or to protect such obligee against loss in respect of such Indebtedness (in whole or in part); <u>provided</u>, <u>however</u>, that the term Guarantee shall not include endorsements for deposit or collection in the Ordinary Course of Business. The term "Guarantee" used as verb has a correlative meaning.

- 43 -

<u>Hazardous Substances</u> means any substance that: (i) is or contains asbestos, urea formaldehyde foam insulation, polychlorinated biphenyls, petroleum or petroleum-derived substances or wastes, leaded paints, toxic mold, radon gas or related materials, (ii) requires investigation, removal or remediation under any Environmental Law, or is defined, listed or identified as a "hazardous waste," "hazardous material," "toxic substance" or "hazardous substance" thereunder, or (iii) is toxic, explosive, corrosive, flammable, infectious, radioactive, carcinogenic, mutagenic, or otherwise regulated by any Governmental Law.

<u>Holdback Amount</u> means \$7,300,000 plus interest accruing at an annual rate of 2%, compounded monthly, on the outstanding amount of such Holdback Amount from the date of Closing until the date the Holdback Amount is disbursed to Seller.

HSR Act means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

<u>Indebtedness</u> means, for any Person, either (a) any liability of any Person (i) for borrowed money (including the current portion thereof), or (ii) under any reimbursement obligation relating to a letter of credit, bankers' acceptance or note purchase facility, or (iii) evidenced by a bond, note, debenture or similar instrument (including a purchase money obligation), (iv) for the payment of money relating to leases that are required to be classified as a capitalized lease obligations in accordance with GAAP for all or any part of the deferred purchase price of property or services (other than trade payables), or (v) for all or any part of the deferred purchase price of property or services (other than trade payables), including any non-compete and earnout payments, or (b) any liability of others described in the preceding clause (a) that such Person has Guaranteed, that is recourse to such Person or any of its assets or that is otherwise its legal liability or that is secured in whole or in part by the assets of such Person.

Indemnified Party has the meaning set forth in Section 9.3(a).

Indemnifying Party has the meaning set forth in Section 9.3(a).

Intellectual Property means any and all United States and foreign: (a) patents (including design patents, industrial designs and utility models) and patent applications (including docketed patent disclosures awaiting filing, reissues, divisions, continuations-in-part and extensions), (b) trademarks, service marks, trade names, trade dress, logos, business and product names, slogans and registrations and applications for registration thereof; (c) copyrights (including software) and registrations thereof; (d) inventions, processes, designs, formulae, trade secrets, know-how, industrial models, confidential and technical information, manufacturing, engineering and technical drawings, product specifications and confidential business information, including any such matters under development and for which subsection (a) above is not applicable; (e) mask work and other semiconductor chip rights and registrations thereof; (f) intellectual property rights (including common law rights) similar to any of the foregoing; and (g) copies and tangible embodiments (in whatever form or medium, including electronic media) of any of the foregoing.

- 44 -

Interim Financial Statements has the meaning set forth in Section 3.1(d).

<u>Inventories</u> has the meaning set forth in <u>Section 1.1(b)</u>.

IRS means the Internal Revenue Service.

knowledge of the Seller means the actual knowledge of the following persons: each of the Principal Stockholders, Frances Sordellini, Marek Janasek, William V. Feldmann and Clark Houghtling.

Leased Real Property means all interests leased pursuant to the Leases.

<u>Leases</u> means the real property leases, subleases, licenses and occupancy agreements related to the Business and pursuant to which the Seller is the lessee, sublessee, licensee or occupant other than real property leases, subleases, licenses and occupancy agreements included in Excluded Assets.

Liability Claim has the meaning set forth in Section 9.3(a).

License Agreement has the meaning set forth in Section 5.2(e).

Licensed IP has the meaning set forth in Section 1.2(g).

Lien means any mortgage, pledge, hypothecation, claim, security interest, encumbrance, lease, sublease, license, occupancy agreement, adverse claim or interest, easement, covenant, encroachment, burden, title defect, title retention agreement, voting trust agreement, interest, equity, option, lien, right of first refusal, charge or other restrictions or limitations of any nature whatsoever, including those that may arise under any Contracts.

Litigation Conditions has the meaning set forth in Section 9.3(b).

Losses has the meaning set forth in Section 9.1.

Maintenance Agreement has the meaning set forth in Section 5.2(e).

<u>Material Adverse Effect</u> means any event, occurrence, fact, condition, change or effect that has or would reasonably be expected to have a materially adverse effect on the business, operations, results of operations, condition (financial or otherwise), properties (including intangible properties), assets (including intangible assets) or liabilities of the Buyer or on the Business or Assets of the Seller, as the case may be; provided, however, that the term Material Adverse Effect shall not include any event, occurrence, fact, condition, change or effect generally affecting (and that does not have a disproportionately adverse effect on the Seller or the Business, as applicable, compared to its effect on competitors of the Seller or the Business): (i) the industry in which the Business operates, (ii) interest rates, (iii) securities or capital markets, (iv) general economic conditions or (v) Applicable Laws.

Material Customers has the meaning set forth in Section 3.1(n)(i).

- 45 -

<u>Material Suppliers</u> has the meaning set forth in <u>Section 3.1(n)(ii)</u>.

<u>Migration</u> means seeping, leaching or other movement of Hazardous Substances through the soil, land, surface or subsurface strata, sediments, surface water or ground water.

Non-Competition Agreement has the meaning set forth in Section 4.1(m).

Non-Medical Customers means those certain customers identified on Schedule 8.1.

Orders means any order, judgment, injunction, award, decree, ruling, charge or writ of any Governmental Authority.

<u>Ordinary Course of Business</u> means the ordinary course of business consistent with past custom and practice (including with respect to quantity and frequency).

<u>Other Leases</u> means the leases, subleases, licenses and occupancy agreements pursuant to which the Seller is a lessor, sublessor or licensor of any part of the Real Property other than leases, if any, included in Excluded Assets.

Other Services means those certain services specified on Schedule 8.1.

Owned Business Intellectual Property has the meaning set forth in Section 3.1(p).

<u>Owned Real Property</u> means the real property owned by the Seller or its Affiliates and used in the Business, together with all structures, facilities, improvements, fixtures, systems, equipment and items of property located thereon, attached or appurtenant thereto and all easements, licenses, rights and appurtenances relating to the foregoing other than owned real property, if any, included in Excluded Assets.

Pending Claim Notice has the meaning set forth in Section 9.7.

<u>Permitted Liens</u> means (i) Liens for Taxes not yet due and payable or (ii) non-monetary Liens that, individually and in the aggregate, do not and would not reasonably be expected to materially detract from the value of any of the property or assets of the Business or interfere with the use thereof as currently used or contemplated to be used or otherwise.

Person means any natural person, firm, partnership, association, corporation, limited liability company, trust, business trust, Governmental Authority or other entity.

Personal Property Tax has the meaning set forth in Section 2.7.

Plan has the meaning set forth in Section 3.1(u).

Principal Stockholders has the meaning set forth in the recitals.

Purchase Price has the meaning set forth in Section 2.2.

- 46 -

<u>Qualifying Business Combination</u> means any Business Combination in which the Qualifying Consideration payable to the Seller and/or its stockholders, directly and indirectly, is equal to or greater than \$74 million.

<u>Qualifying Consideration</u> means, with respect to any Qualifying Business Combination, the total aggregate gross proceeds and other consideration paid to or received by the Company or the Company's stockholders and to be paid to or received by the Company or the Company's stockholders (which shall be deemed to include all amounts paid or to be paid into escrow, all amounts retained or otherwise held back, all amounts payable pursuant to earn-out provisions and all other contingent payments), including (i) cash, (ii) the full value of all notes, securities and other property, (iii) all liabilities and obligations (including any Indebtedness) of the Company and its stockholders assumed or to be assumed by the acquiror in connection with such transaction, and (iv) the full value of any securities or assets of the Company retained by the Company or its stockholders upon the consummation of the Qualifying Business Combination.

Real Property means the Owned Real Property and the Leased Real Property.

Real Property Laws has the meaning set forth in Section 3.1(r).

Real Property Tax has the meaning set forth in Section 2.7.

<u>Rehired Employees</u> has the meaning set forth in <u>Section 6.1(a)</u>.

<u>Release</u> means any releasing, disposing, discharging, injecting, spilling, leaking, leaching, pumping, dumping, emitting, escaping, emptying, seeping, dispersal, migration, transporting, placing and the like, or otherwise entering, into the Environment.

<u>Remedial Action</u> means all actions required by a Governmental Authority or by any Environmental Law to (i) clean up, remove, treat or in any other way remediate any Hazardous Substances; (ii) prevent the release or migration of Hazardous Substances so that they do not endanger or threaten to endanger public health or welfare or the environment; or (iii) perform studies, investigations, care or other Response related to any such Hazardous Substances.

<u>Response</u> means "response" as defined in 42 U.S.C. Section 9601(25), to the extent necessary to comply with Environmental Law, and includes all ancillary activities reasonably or necessarily required to investigate the nature and extent of Releases or threatened Releases of Hazardous Substances and to implement such response action, including but not limited to: (i) obtaining access agreements, Governmental Authority approvals, and permits required by federal, state or local law; (ii) preparing and submitting reports required by any Governmental Authority order or requirement; (iii) operation and maintenance activities associated with any treatment system; (iv) monitoring activities; (v) oversight costs of a Governmental Authority; and (vi) investigative, remedial and monitoring work in, on, under or about properties as to which actual or suspected Migration of Hazardous Substances has or may have occurred.

- 47 -

Seller has the meaning set forth in the Preamble.

Seller Indemnitees has the meaning set forth in Section 9.2.

Solvent means, at any time, that (i) the fair market value and present fair saleable value of the applicable entity's assets is greater than the amount of its stated and identified contingent liabilities; (ii) the applicable entity is able to realize upon its property and pay its debts and other liabilities as they mature in the normal course of business; and (iii) the applicable entity is not engaged in business or a transaction for which its property would constitute unreasonably small capital.

<u>Subsidiary</u> means any Person of which at least 50% of the outstanding shares or other equity interests having ordinary voting power for the election of directors or comparable managers of such Person are at the time owned by the Seller, by one or more directly or indirectly wholly or partially owned subsidiaries of the Seller or by the Seller and one or more such subsidiaries, whether or not at the time the shares of any other class or classes or other equity interests of such Person shall have or might have voting power by reason of the happening of any contingency.

Supply Agreement has the meaning set forth in Section 5.2(e).

<u>Target Working Capital</u> means an amount equal to the product of \$1,916,536 multiplied by the greater of (i) one, and (ii) a ratio equal to (A) the revenues of the Business during the last six full months immediately preceding the Closing Date, over (B) \$11,811,736.

Tax means any federal, state, provincial, local, foreign or other income, alternative, minimum, accumulated earnings, personal holding company, franchise, capital stock, net worth, capital, profits, windfall profits, gross receipts, value added, sales, use, goods and services, excise, customs duties, transfer, conveyance, mortgage, registration, stamp, documentary, recording, premium, severance, environmental (including taxes under Section 59A of the Code), real property, personal property, ad valorem, intangibles, rent, occupancy, license, occupational, employment, unemployment insurance, social security, disability, workers compensation, payroll, health care, withholding, estimated or other similar tax, duty or other governmental charge or assessment or deficiencies thereof (including all interest and penalties thereon and additions thereto whether disputed or not) and (ii) any liability for the payment of any amount of the types described in the immediately preceding clause (i) as a result of Seller (x) being a member of an affiliated, consolidated, combined or unitary group, (y) being a party to any agreement to indemnify any Person, or (z) being a successor or transferee of any other Person.

<u>Tax Return</u> means any return, report, declaration, form, claim for refund or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

Transaction Expenses has the meaning set forth in Section 10.1.

- 48 -

Transfer has the meaning set forth in the Recitals.

Transfer Taxes has the meaning set forth in Section 4.1(g).

Treasury Regulations means the regulations prescribed pursuant to the Code.

<u>Utility Charges</u> has the meaning set forth in <u>Section 2.7</u>.

Withholding Taxes has the meaning set forth in Section 3.1(f).

<u>Working Capital</u> means (a) the sum of the aggregate (i) inventories, (ii) accounts receivable, net of any reserves, (iii) pre-paid expenses and other credits, and (iv) petty cash held at the Facilities less (b) the sum of the aggregate (i) accounts payable, and (ii) accrued expenses, each as of the measurement date, and calculated in a manner consistent with the Financial Statements and <u>Schedule 2.3</u>. Working Capital does not include any Excluded Assets or Excluded Liabilities actually retained by the Seller.

Working Capital Statement has the meaning set forth in Section 2.3(a).

ARTICLE IX INDEMNIFICATION

9.1 <u>Indemnification By the Seller</u>. The Seller covenants and agrees to defend, indemnify and hold harmless the Buyer and its respective officers, directors, employees, agents, advisers, representatives and Affiliates (collectively, the "<u>Buyer Indemnitees</u>") from and against, and pay or reimburse the Buyer Indemnitees for, any and all claims, liabilities, obligations, losses, fines, penalties, costs, royalties, proceedings, deficiencies or damages (whether absolute, accrued, conditional or otherwise and whether or not resulting from third party claims), including expenses and reasonable attorneys' and accountants' fees incurred in the investigation or defense of any of the same or in asserting any of their respective rights hereunder (collectively, "Losses"), resulting from or arising out of:

(i) any inaccuracy of any representation or warranty made by the Seller herein or under any Ancillary Agreement or in connection herewith or therewith;

(ii) any failure of the Seller to perform any covenant or agreement hereunder or under any Ancillary Agreement or fulfill any other obligation in respect hereof or thereof; and

(iii) any Excluded Liabilities or Excluded Assets.

9.2 <u>Indemnification by the Buyer</u>. The Buyer covenants and agrees to defend, indemnify and hold harmless the Seller and its officers, directors, employees, agents, advisers, representatives and Affiliates (collectively, the "<u>Seller Indemnitees</u>") from and against and pay or reimburse the Seller Indemnities for any and all Losses resulting from or arising out of:

(i) any inaccuracy in any representation or warranty made by the Buyer herein or under any Ancillary Agreement or in connection herewith or therewith;

- 49 -

(ii) any failure of the Buyer to perform any covenant or agreement herein or under any Ancillary Agreement or fulfill any other obligation in respect hereof or thereof; and

(iii) the Assumed Liabilities.

9.3 Notice and Opportunity to Defend.

(a) <u>Notice of Asserted Liability</u>. As soon as is reasonably practicable after the Seller or the Buyer becomes aware of any claim that such party has under <u>Section 9.1</u> or <u>9.2</u> that may result in a Loss (a "<u>Liability Claim</u>"), such party (the "<u>Indemnified Party</u>") shall give written notice of such Liability Claim (a "<u>Claims Notice</u>") to the other party (the "<u>Indemnifying Party</u>"). A Claims Notice must describe the Liability Claim in reasonable detail and must indicate the amount (estimated, if necessary and to the extent feasible) of the Loss that has been or may be suffered by the Indemnified Party. No delay in or failure to give a Claims Notice by the Indemnified Party to the Indemnifying Party pursuant to this <u>Section 9.3</u> will adversely affect any of the other rights or remedies that the Indemnified Party has under this Agreement or alter or relieve the Indemnifying Party; provided, however, that all Liability Claims made in connection with this Agreement must be made within the relevant time periods set forth in <u>Section 9.4</u>.

(b) <u>Opportunity to Defend</u>. The Indemnifying Party has the right, exercisable by written notice to the Indemnified Party within 30 days after receipt of a Claims Notice from the Indemnified Party of the commencement or assertion of any Liability Claim in respect of which indemnity may be sought under this <u>Article IX</u>, to assume and conduct the defense of such Liability Claim in accordance with the limits set forth in this Agreement with counsel selected by the Indemnifying Party and reasonably acceptable to the Indemnified Party; <u>provided</u>, <u>however</u>, that (i) the defense of such Liability Claim by the Indemnifying Party will not, in the reasonable judgment of the Indemnified Party, have a Material Adverse Effect or Buyer Material Adverse Effect, as the case may be, on the Indemnified Party; (ii) the Indemnifying Party has sufficient financial resources, in the reasonable judgment of the Indemnified Party, to satisfy the amount of any adverse monetary judgment that is reasonably likely to result; (iii) the Liability Claim solely seeks (and continues to seek) monetary damages; and (iv) the Indemnifying Party expressly agrees in writing that as between the Indemnifying Party and the Indemnified Party, the Indemnifying Party may only satisfy and discharge the Liability Claim in accordance with the limits set forth in this Agreement (the conditions set forth in clauses (i) through (iv) are, collectively, the "<u>Litigation Conditions</u>"). If the Indemnifying Party does not assume the defense of a Liability Claim in accordance with this <u>Section 9.3(b</u>), the Indemnified Party may continue to defend the Liability Claim. If the Indemnifying Party has assumed the defense of a Liability Claim as provided in this <u>Section 9.3(b</u>), the Indemnifying Party will not be liable for any legal expenses subsequently incurred by the Indemnified Party in connection with the defense of the Liability Claim; <u>provided</u>, <u>however</u>, that if (i) any of the Litigation Conditions cease to be met or (ii) the Indemnifying Party fails to take

- 50 -

Indemnifying Party will be liable for all reasonable costs or expenses paid or incurred in connection with such defense. The Indemnifying Party or the Indemnified Party, as the case may be, has the right to participate in (but not control), at its own expense, the defense of any Liability Claim which the other is defending as provided in this Agreement. The Indemnifying Party, if it has assumed the defense of any Liability Claim as provided in this Agreement, may not, without the prior written consent of the Indemnified Party, consent to a settlement of, or the entry of any judgment arising from, any such Liability Claim that (i) does not include as an unconditional term thereof the giving by the claimant or the plaintiff to the Indemnified Party a complete release from all liability in respect of such Liability Claim, (ii) grants any injunctive or equitable relief, or (iii) would reasonably be expected to cause a Material Adverse Effect or Buyer Material Adverse Effect, as the case may be, on the affected business of the Indemnified Party. The Indemnified Party has the right to settle any Liability Claim, the defense of which has not been assumed by the Indemnifying Party using good faith and commercial reasonableness.

9.4 Survival.

(a) The representations and warranties in this Agreement shall survive the Closing as follows:

(i) the representations and warranties in Section 3.1(a) (Authorization), Section 3.1(b) (Corporate Status), Section 3.1(c) (No Conflicts), the first sentence of Section 3.1(k) (Assets) and Section 3.2(a) (Corporate Status; Authorization) shall survive the Closing indefinitely;

(ii) the representations and warranties in <u>Section 3.1(f)</u> (Taxes) and <u>Section 3.1(u)</u> (Employee Benefit Plans and Related Matters) shall survive the Closing and will not terminate until 30 days after the expiration of the statute of limitations (taking into account for this purpose, any extensions) applicable to the matters covered thereby;

(iii) the representations and warranties in <u>Section 3.1(r)</u> (Real Property) shall survive the Closing and will not terminate until the third anniversary of the Closing Date, and the representations and warranties in <u>Section 3.1(s)</u> (Environmental Matters) shall survive the Closing and will not terminate until the fifth anniversary of the Closing Date; and

(iv) all other representations and warranties in the Agreement shall survive the Closing and will not terminate until the twelve-month anniversary of the Closing Date.

(b) The covenants and agreements of the parties hereto contained in this Agreement shall, subject to the express terms thereof, survive the Closing indefinitely.

- 51 -

9.5 Limitations in Indemnification.

(a) Notwithstanding anything to the contrary contained in this <u>Article IX</u>, the Seller will not have any liability as a result of any breach of or inaccuracy in any representation or warranty referred to in this Agreement (other than those representations and warranties of the Seller referenced in clauses (i) and (ii) of <u>Section 9.4</u> for which such limitation shall not apply) until the aggregate amount of all such Losses sustained by the Buyer exceeds \$100,000.

(b) Notwithstanding anything to the contrary contained in this <u>Article IX</u>, the Seller shall not have any liability under this Agreement as a result of any breach of or inaccuracy in any of the representations and warranties contained in this Agreement (other than those representations and warranties of the Seller referenced in Clauses (i) and (ii) of <u>Section 9.4</u> for which such limitation shall not apply) in excess of 35% of the Purchase Price, in the aggregate.

9.6 Specific Performance; General Provisions.

(a) Each party's obligation under this Agreement is unique. If any party should breach its covenants under this Agreement, the parties each acknowledge that it would be extremely impracticable to measure the resulting damages; accordingly, the non-breaching party or parties, in addition to any other available rights or remedies, may sue in equity for specific performance, and each party expressly waives the defense that a remedy in damages will be adequate.

(b) Any amounts payable by the Indemnifying Party to or on behalf of the Indemnified Party pursuant to this Agreement shall be reduced by an amount equal to the greater of (i) the amount of any insurance proceeds actually received by the Indemnified Party as a result of the claim, loss or damage involved (not to exceed the amounts payable by the Indemnifying Party pursuant to such Liability Claim) multiplied by 95%, and (ii) the amount of any insurance proceeds actually received by the Indemnified Party as a result of the claim, loss or damage involved (not to exceed the amounts payable by the Indemnified Party as a result of the claim, loss or damage involved (not to exceed the amounts payable by the Indemnified Party as a result of the claim, loss or damage involved (not to exceed the amounts payable by the Indemnified Party as a result of the claim, loss or damage involved (not to exceed the amounts payable by the Indemnified Party as a result of the claim, loss or damage involved (not to exceed the amounts payable by the Indemnified Party seek insurance proceeds), to the extent that any claim or Loss is covered by insurance, no Indemnified Party may make a claim or receive payment for indemnification pursuant to this Article IX without first making, or undertaking to make, a claim under an applicable insurance policy, if any, and must use commercially reasonable efforts to pursue such claim. The Indemnified Party shall be entitled to payment for any Loss with respect to a Claims Notice to the Indemnifying Party if the insurance proceeds in respect of such Loss have not been received by the Indemnified Party within 60 days after a claim has been made by the Indemnified Party under any applicable insurance policy, provided however, that if insurance proceeds are subsequently received by the Indemnified Party in connection with such claim, the Indemnified Party as a result of the Indemnifying Party an amount equal to the greater of (i) the amount of any insurance proceeds actually received by the Indemnified P

- 52 -

amount of any insurance proceeds actually received by the Indemnified Party as a result of the claim, loss or damage involved (not to exceed the amounts payable by the Indemnifying Party pursuant to such Liability Claim) minus \$50,000 (in the case of both clauses (i) and (ii) of this sentence to the extent that the Indemnifying Party had previously made indemnification payments in respect of such Loss to the Indemnified Party).

(c) The parties shall cooperate with each other with respect to any claim or liability with respect to which one party is obligated to indemnify the other party hereunder and use commercially reasonable efforts to mitigate any such claim or liability under this Agreement, in each case, to the extent required by Applicable Law.

9.7 Holdback Amount

(a) On the twelve month anniversary of the Closing Date, the Buyer shall pay or cause to be paid by bank wire transfer of immediately available funds, to the account designated in writing by the Seller, the Holdback Amount, less the sum of the amounts, if any, specified in any then Pending Claim Notices or any claims previously paid to Buyer out of the Holdback Amount. For purposes of this Agreement, a "<u>Pending Claim Notice</u>" refers to a written notice provided by the Buyer to the Seller specifying in reasonable detail the nature and dollar amount (including a good faith estimate of the aggregate amount of Losses that may be suffered by the Buyer) of any claim the Buyer may have under <u>Section 9.1</u>.

(b) The Buyer shall pay or cause to be paid by bank wire transfer of immediately available funds to the account designated in writing by the Seller any portion of the Holdback Amount withheld pursuant to <u>Section 9.7(a)</u> with respect to a Pending Claim Notice if it is finally determined pursuant to <u>Section 9.7(c)</u> that no indemnification payment was required to be made by the Seller in respect of such Pending Claim Notice or such indemnification payment is less than originally estimated in such Pending Claim Notice.

(c) In the event of any dispute regarding the final amount payable under a Pending Claim Notice, Buyer and Seller shall negotiate in good faith to resolve such dispute. If Buyer and Seller, not withstanding such good faith effort, are unable to mutually agree to the amount of any claims subject to a Pending Claim Notice within thirty (30) days after the Indemnifying Party has notified the Indemnified Party of its objections thereto, and any underlying lawsuit has been resolved, then either Party shall have the right to demand that such dispute be resolved through binding arbitration. Upon such a demand, the Parties shall submit the dispute to an arbitrator jointly selected by Buyer and Seller, and the arbitrator shall apply the Rules for Commercial Arbitration of the American Arbitration Association to the extent applicable. (If Buyer and Seller cannot agree on an arbitrator, then each shall appoint an arbitrator and those two arbitrators shall select an arbitrator to handle the dispute.) The arbitrator's determination of the final amount due under such Pending Claim Notice shall be final and binding on Buyer and Seller. The expenses of the arbitrator shall be paid by Buyer and/or Seller as directed by the arbitrator.

- 53 -

ARTICLE X MISCELLANEOUS

10.1 Expenses. Except as otherwise provided in this Agreement, the Seller and the Buyer shall bear their respective expenses, costs and fees (including attorneys' and auditors' fees) in connection with the transactions contemplated hereby, including the preparation, execution and delivery of this Agreement and compliance herewith (the "<u>Transaction Expenses</u>"), whether or not the transactions contemplated hereby shall be consummated. In the event (a) that this Agreement is terminated (1) by the Buyer prior to the Closing in accordance with <u>Section 7.1(c)</u>, or (2) by the Seller in accordance with <u>Section 7.1(b)</u>, and (b) the Seller or any of its stockholders, prior to December 31, 2005, enter into any agreement or other binding obligation or understanding with respect to, or consummates, any Qualifying Business Combination (or series of transactions that constitute, in the aggregate, a Qualifying Business Combination), the Seller shall pay all of the documented out-of-pocket expenses actually incurred by the Buyer in connection with the transactions contemplated however, if this Agreements (including the fees and expenses of the Buyer's legal counsel and other advisors) up to a maximum of \$1 million; provided, however, if this Agreement is terminated by the Buyer prior to the Closing in accordance with <u>Section 7.1(c)</u> because the satisfaction of a condition set forth in <u>Section 5.1</u> becomes impossible and such failure is not due to the Seller's failure to perform or comply with any of its covenants or agreements, the Buyer shall not be entitled to the expense reimbursement set forth in this sentence. The Seller shall give prompt written notice to the Buyer in the event that it, or its stockholders, enter into any such agreement to other binding obligation or understanding with respect to, or if it or they consummate, any Qualifying Business Combination. Any payment due pursuant to this <u>Section 10.1</u> shall be paid within five business days after receipt by the Seller from the Buyer of a reasonably detailed s

10.2 <u>Severability</u>. If any provision of this Agreement, including any phrase, sentence, clause, section or subsection is inoperative or unenforceable for any reason, such circumstances shall not have the effect of rendering the provision in question inoperative or unenforceable in any other case or circumstance, or of rendering any other provision or provisions herein contained invalid, inoperative, or unenforceable to any extent whatsoever.

10.3 <u>Notices</u>. All notices, requests, demands, waivers and other communications required or permitted to be given under this Agreement shall be in writing and shall be deemed to have been duly given if (a) delivered personally, (b) mailed by first-class, registered or certified mail, return receipt requested, postage prepaid, or (c) sent by next-day or overnight mail or delivery or (d) sent by facsimile.

- 54 -

(i) if to the Buyer to:

STERIS Corporation 5960 Heisley Road Mentor, Ohio 44060-1834 Attention: General Counsel Fax Number: (440) 392-7344

with a copy to:

Jones Day 901 Lakeside Avenue Cleveland, Ohio 44114 Attention: Thomas C. Daniels, Esq. Fax Number: (216) 579-0212

(ii) if to the Seller to:

Cosmed Group, Inc. 28 Narragansett Avenue Jamestown, Rhode Island 02835 Attention: Mr. Michael Howe Fax Number: (401) 423-2301

with a copy to:

Edwards & Angell, LLP 2800 Financial Plaza Providence, Rhode Island 02903 Attention: Douglas G. Gray, Esq. Fax Number: (401) 276-6611

or, in each case, at such other address as may be specified in writing to the other parties hereto.

All such notices, requests, demands, waivers and other communications shall be deemed to have been received (w) if by personal delivery on the day after such delivery, (x) if by certified or registered mail, on the seventh business day after the mailing thereof, (y) if by next-day or overnight mail or delivery, on the day delivered, (z) if by facsimile, on the next day following the day on which such facsimile was sent, provided that a copy is also sent by certified or registered mail.

10.4 <u>Headings</u>. The headings contained in this Agreement are for purposes of convenience only and shall not affect the meaning or interpretation of this Agreement.

10.5 <u>Entire Agreement</u>. This Agreement (including the Schedules and Exhibits hereto), the Ancillary Agreements (when executed and delivered), that certain Guarantee Agreement between the Buyer and the Principal Stockholders executed on the date hereof, and

- 55 -

any other documents executed pursuant to this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, between the parties with respect to the subject matter hereof.

10.6 <u>Counterparts</u>. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall together constitute one and the same instrument.

10.7 <u>Governing Law; Consent to Jurisdiction</u>. This Agreement shall be governed in all respects, including as to validity, interpretation and effect, by the internal laws of the State of New York, without giving effect to the conflict of laws rules thereof. Each of the parties hereto (a) elects, except as otherwise expressly set forth herein, that the sole judicial forum for the adjudication of any dispute arising out of or in connection with this Agreement, the Ancillary Agreements or any of the transactions contemplated by this Agreement shall be any U.S. federal court located in the State of New York, as well as any court to which appeal may be taken from such courts, (b) consents to submit itself to the personal jurisdiction of any federal court located in the State of New York or any New York state court in the event any dispute arises out of this Agreement, the Ancillary Agreements, or any of the transactions contemplated by this Agreement, and (c) agrees that it will not attempt to object to, deny or defeat such personal jurisdiction by motion or other request for leave from any such court.

10.8 <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, successors and permitted assigns.

10.9 <u>Assignment</u>. This Agreement shall not be assignable or otherwise transferable by any party hereto <u>provided</u> that the Buyer may assign this Agreement to any Affiliate of the Buyer, and <u>provided</u>, <u>further</u>, that no assignment shall in any way affect the Buyer's obligations or liabilities under this Agreement and Buyer shall unconditionally and absolutely guaranty any obligations of such Affiliate pursuant to this Agreement or any Ancillary Agreements.

10.10 <u>No Third Party Beneficiaries</u>. Nothing in this Agreement shall confer any rights upon any person or entity other than the parties hereto and their respective heirs, successors and permitted assigns.

10.11 <u>Amendment: Waivers, etc</u>. This Agreement may be amended by the mutual agreement of parties at any time. No amendment, modification or discharge of this Agreement, and no waiver hereunder, shall be valid or binding unless set forth in writing and duly executed by the party against whom enforcement of the amendment, modification, discharge or waiver is sought. Any such waiver shall constitute a waiver only with respect to the specific matter described in such writing and shall in no way impair the rights of the party granting such waiver in any other respect or at any other time.

10.12 <u>Construction</u>. The parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement. Any reference to any federal, state, local or foreign statute or law

- 56 -

will be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise. The word "including" shall mean including without limitation. Any reference to the singular in this Agreement shall also include the plural and vice versa. The word "knowledge" shall mean knowledge obtained or obtainable after due inquiry and reasonable investigation.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the parties have duly executed this Asset Purchase Agreement as of the date first above written.

COSMED GROUP, INC.

By: /s/ Michael L. Howe

Name:Michael L. HoweTitle:Chief Executive Officer

STERIS CORPORATION

By: /s/ Les C. Vinney

Name:Les C. VinneyTitle:President and Chief Executive Officer

The following Persons are parties to this Agreement solely with respect to the obligations set forth in <u>Section 4.1(m)</u> and <u>Section 4.1(n)</u> hereof:

PRINCIPAL STOCKHOLDERS

/s/ Michael L. Howe

Name: Michael L. Howe

/s/ David G. Howe

Name: David G. Howe

STERIS CORPORATION

NONQUALIFIED STOCK OPTION AGREEMENT

This Agreement is between STERIS Corporation ("STERIS") and **[Optionee's Name]** ("Optionee"), with respect to the grant of a Nonqualified Stock Option by STERIS to Optionee pursuant to the STERIS Corporation 2002 Stock Option (the "Plan"). (Capitalized terms used in this Agreement and not otherwise defined have the meanings assigned to them in the Plan.)

1. *Grant of Option*. STERIS hereby grants to Optionee, as of **[Date of Grant]**, an option (the "Option") to purchase all or any number of an aggregate of **[Number of Shares Granted – referred to below as "X"]** STERIS Common Shares, at an exercise price of **[Grant Price]** per share, upon and subject to the terms of this Agreement and the Plan.

2. Documents Delivered with Agreement. STERIS has delivered to Optionee, along with two copies of this Agreement, the following documents: (a) a copy of STERIS's Policy Prohibiting the Improper Use of Material Non-Public Information (the "Policy"); (b) a copy of the Plan and its related Prospectus, (c) two execution copies of a Nondisclosure and Noncompetition Agreement to be entered into between STERIS and Optionee (the "Nondislosure Agreement"); (d) two copies of an acknowledgment form (the "Acknowledgment Form"); and (e) a copy of STERIS's most recent integrated Annual Report to Shareholders and Form 10-K (the "Annual Report"). By executing this Agreement, Optionee acknowledges receipt of these documents.

3. *Terms and Conditions of Option*. The Option is a Nonqualified Option and shall not be treated as an Incentive Stock Option. Except as otherwise provided in this Agreement, the Option shall be subject to all of the terms and conditions of the Plan. As a condition to the effectiveness of the Option, Optionee must return to STERIS signed copies of (a) this Agreement, (b) the Nondisclosure Agreement, and (c) the Acknowledgment Form. If Optionee violates the terms of the Policy, or this Agreement, the Nondisclosure Agreement, or any similar agreement previously entered into (collectively "Prior Agreements"), any and all options to purchase Common Shares that were granted by STERIS to Optionee (including the Option granted by this Agreement or any Prior Agreements) shall be forfeited, void, and of no further force and effect.

4. *Term of Option*. Unless earlier terminated, the Option shall terminate at the close of business on, and shall not be exercisable at any time after, **[Ten Years and One Month from Date of Grant]**.

5. *Vesting.* So long as Optionee remains in the employ of STERIS, but subject to the terms of this Agreement and the Plan, the Option shall vest and first become exercisable in equal increments (excluding any fractional shares) as follows:

From and after this date:

[First Anniversary of Grant] [Second Anniversary of Grant] [Third Anniversary of Grant] [Fourth Anniversary of Grant] The Option is Exercisable with Respect to:

[X/4] Common Shares
 an additional [X/4] Common Shares
 an additional [X/4] Common Shares
 the remaining [X/4] Common Shares

If Optionee's employment terminates before **[Fourth Anniversary of Grant]** and Optionee is entitled to exercise the Option after the Employment Termination Date by reason of the application of Section 6 of the Plan, a portion of the Option shall also vest and first become exercisable, as of the Employment Termination Date (in addition to the extent to which it has already vested pursuant to the foregoing schedule), to the extent of an additional percentage of the full number of Common Shares subject to the Option determined by multiplying 25% by a fraction the numerator of which is the number of days between the Employment Termination Date and the immediately preceding **[Anniversary of Grant]** and the denominator of which is 365. 6. *Exercise of Vested Option*. Except as otherwise provided in Section 6 of the Plan, the Option shall be exercisable only while Optionee is in the employ of STERIS. To the extent exercisable under the Agreement, the Option may be exercised from time to time in whole or in part.

7. *Method of Exercise*. A request to exercise the Option requires delivery of (a) the Exercise Price payable in accordance with Section 5.6 of the Plan and (b) a written notice to STERIS identifying this Agreement and specifying the number of Common Shares as to which the Option is being exercised. Subject to the terms of the Plan and this Agreement, STERIS shall deliver to Optionee certificates representing the Option shares as soon as administratively feasible following such exercise.

8. *Certain Determinations*. Application, violation, or other interpretation of the terms of this Agreement, the Nondisclosure Agreement, the Policy, any Prior Agreement, or any STERIS policy shall be determined by the Committee, in its sole discretion, and its determination shall be final and binding on Optionee and STERIS.

9. *Termination of the Plan; No Right to Future Grants; No Right of Employment; Extraordinary Item of Compensation.* By entering into this Agreement, the Optionee acknowledges: (a) that the Plan is discretionary in nature and may be suspended or terminated by STERIS at any time; (b) that each grant of an Option is a one-time benefit which does not create any contractual or other right to receive future grants of Options, or benefits in lieu of Options; (c) that all determinations with respect to any such future grants, including, but not limited to, the times when the Option shall be granted, the number of shares subject to each Option, the Option price, and the time or times when each Option shall be exercisable, will be at the sole discretion of STERIS; (d) that the Optionee's participation in the Plan shall not create a right to further employment with the Optionee's employer and shall not interfere with the ability of the Optionee's employer to terminate the Options is an extraordinary item of compensation which is outside the scope of the Optionee's employment contract, if any; (g) that the Option is not part of normal and expected compensation for purposes of calculating any severance, resignation, redundancy, end of service payments, bonuses, long-service awards, pension or retirement benefits or similar payments; (h) that the right to purchase stock ceases upon termination of employment for any reason except as may otherwise be explicitly provided in the Plan or this Agreement; (i) that the future value of the Optionee's employer or that the Optionee's employer is a Subsidiary or affiliate of STERIS, the Option has been granted to the Optionee in the Optionee's status as an employee of such Subsidiary or affiliate, and can in no event be understood or interpreted to mean that STERIS is the Option will have no value.

10. *Employee Data Privacy*. By entering into the Agreement, and as a condition of the grant of the Option, the Optionee consents to the collection, use and transfer of personal data as described in this Section 10. The Optionee understands that STERIS and its Subsidiaries hold certain personal information about the Optionee, including, but not limited to, the Optionee's name, home address and telephone number, date of birth, social insurance number, salary, nationality, job title, any shares of stock or directorships held in STERIS, details of all Options or other entitlement to shares of stock awarded, canceled, exercised, vested, unvested or outstanding in the Optionee's favor, for the purpose of managing and administering the Plan ("Data"). The Optionee further understands that STERIS and/or its Subsidiaries will transfer Data among themselves as necessary for the purposes of implementation, administration and management of the Optionee's participation in the Plan, and that STERIS and/or its Subsidiaries may each further transfer Data to any third parties assisting STERIS in the implementation, administration and management of the Plan ("Data Recipients"). The Optionee understands that these Data Recipients may be located in the Optionee's country of residence, the European Union, or elsewhere, such as the United States. The Optionee authorizes the Data Recipients to receive, possess, use, retain and transfer Data in electronic or other form, for the purposes of implementing, administering and managing the Plan, including any transfer of such Data, as may be necessary or appropriate for the administration of the Plan and/or the subsequent holding of shares of stock on the Optionee's behalf, to a broker or third party with whom the shares acquired on exercise may be deposited. The Optionee understands that he or she may, at any time, review the Data, require any necessary amendments to it or withdraw the consent herein by notifying STERIS in

writing. The Optionee further understands that withdrawing consent may affect the Optionee's ability to participate in the Plan, at the sole discretion of the Committee.

11. *Miscellaneous*. Nothing contained in this Agreement shall be understood as conferring on Optionee any right to continue as an employee of STERIS or any Subsidiary or affiliate. STERIS reserves the right to correct any clerical, typographical, or other error in this Agreement or otherwise with respect to this grant. This Agreement shall inure to the benefit of and be binding upon its parties and their respective heirs, executors, administrators, successors, and assigns, but the Option shall not be transferable by Optionee other than as provided in Section 8 of the Plan.

IN WITNESS WHEREOF, STERIS has caused this Agreement to be executed on its behalf by its duly authorized officer, and Optionee has executed this Agreement, all as of the day and year first written above.

STERIS Corporation

By: /s/ Mark D. McGinley

Mark D. McGinley Vice President, General Counsel, and Secretary Optionee

[Optionee's Name]

STERIS CORPORATION

NOTICE OF RESTRICTED STOCK GRANT

This Notice is given as of **[Date of Grant]** by STERIS Corporation ("STERIS") to **[Director's Name]**, a nonemployee Director of STERIS ("Participant"), to memorialize the grant of Restricted Shares to Participant pursuant to the STERIS Corporation 1994 Nonemployee Directors Equity Compensation Plan (the "Plan") as approved by STERIS shareholders at their annual meeting held on August 25, 1994 (the "1994 Annual Meeting") and as subsequently amended. (Capitalized terms used in this Agreement and not otherwise defined have the same meanings assigned to them in the Plan.)

1. *Grant of Restricted Shares*. As of **[Date of Grant]**, STERIS automatically granted to Participant **[Number of Shares Granted – referred to below as "X"]** Restricted Shares, which grant is hereby memorialized and affirmed. The Restricted Shares are granted as partial payment of the first \$14,000 of the annual retainer for services to be rendered by Participant during the Annual Term commencing on the date of the **[Current Year]** Annual Meeting and continuing through the date of the Annual Meeting of the shareholders of STERIS to be held in **[Following Year]**. The Restricted Shares are subject to all of the terms and conditions of the Plan. Participant hereby acknowledges receipt of a copy of the Plan, enclosed herewith.

2. *Restrictions on Restricted Share Grant*. Participant shall not sell, transfer, otherwise dispose of, or pledge or otherwise hypothecate any Restricted Shares (a) at any time before **[Six Months after Date of Grant]** and (b) thereafter, until Participant's rights with respect to those particular Restricted Shares have vested in accordance with Section 5.3(b) of the Plan.

3. Vesting of Restricted Shares. The Restricted Shares shall vest in increments as follows, provided that Participant remains a director through the dates indicated:

Restricted Shares	Vest On
[2/7 of X]	January 31, [Following Year]
[1/7 of X]	February 28, [Following Year]
[1/7 of X]	March 31, [Following Year]
[1/7 of X]	April 29, [Following Year]
[1/7 of X]	May 31, [Following Year]
[1/7 of X]	June 30, [Following Year]

If Participant ceases to be a Director of STERIS before June 30, **[Following Year]** for any reason other than death, any Restricted Shares that have not vested will be forfeited. Nothing in this Agreement shall be deemed to confer upon Participant any right to continue as a Director of STERIS.

4. *Federal Income Tax Consequences*. Whether Participant will recognize income (a) as of the **[Date of Grant]** grant of Restricted Shares or (b) as and when the restrictions on the Restricted Shares lapse in accordance with the table set forth above, will depend on whether Participant files an election under Section 83(b) of the Internal Revenue Code. If Participant does file such an election, the income will be recognized in **[Current Year]**. If Participant does not file such an election, the income will be recognized in **[Following Year]** as and when the restrictions lapse.

IN WITNESS WHEREOF, STERIS has caused this Notice to be executed on its behalf by its duly authorized officer, and Participant has hereunto set his hand, all as of the day and year first written above.

STERIS Corporation

By: /s/ Mark D. McGinley

Mark D. McGinley Vice President, General Counsel, and Secretary [Director's Name]

STERIS CORPORATION

NONQUALIFIED STOCK OPTION AGREEMENT

This Agreement is made as of **[Date of Grant]**, between STERIS Corporation ("STERIS") and **[Director's Name]**, a Director of STERIS ("Optionee"), to memorialize the grant of a Nonqualified Stock Option by STERIS to Optionee pursuant to the STERIS Corporation 1997 Stock Option Plan (the "Plan"). (Capitalized terms used in this Agreement and not otherwise defined have the same meanings assigned to them in the Plan.)

1. *Grant of Option*. STERIS hereby grants to Optionee, as of **[Date of Grant]**, an option (the "Option") to purchase **[Number of Shares Granted]** STERIS Common Shares, at an exercise price of **[Grant Price]** per share. The Option is granted as additional consideration for services to be rendered by Optionee during the Annual Term commencing on the date of the **[Current Year]** Annual Meeting and continuing through the date of the Annual Meeting of the shareholders of STERIS to be held in **[Following Year]**. The Option is subject to all of the terms and conditions of the Plan.

2. *Documents Delivered with Agreement.* STERIS has delivered to Optionee, along with two copies of this Agreement, the following documents: (a) a copy of the Plan, (b) a copy of STERIS's Policy Prohibiting the Improper Use of Material Non-Public Information (the "Non-Public Information Policy"), and (c) two copies of an acknowledgment form (the "Acknowledgment Form"). By executing this Agreement, Optionee acknowledges receipt of these documents.

3. *Terms and Conditions of Option*. The Option is a Nonqualified Option and shall not be treated as an Incentive Stock Option. Except as otherwise provided in this Agreement, the Option shall be subject to all of the terms and conditions of the Plan. As a condition to the effectiveness of the Option, Optionee must return to STERIS signed copies of (a) this Agreement and (b) the Acknowledgment Form. If Optionee violates the terms of this Agreement, the Non-Public Information Policy, or any similar agreement previously entered into (collectively "Prior Agreements") any and all options to purchase Common Shares that were granted by STERIS to Optionee (including the Option granted by this Agreement or any Prior Agreements) shall be forfeited, void, and of no further force and effect.

4. *Term of Option*. The Option shall be exercisable not earlier than [Six Months after Date of Grant] and shall terminate at the close of business on and shall not be exercisable at any time after [Ten Years and One Month from Date of Grant], except as provided in Section 7.3 of the Plan.

5. *Exercise of Option*. Except as provided in the last sentence of this Section 5 or in Section 6 below, the option shall be exercisable only by Optionee and only during the continuance of Optionee's service as a Director or within three months after termination of such service. If Optionee dies while serving as a Director of STERIS, the Option shall be exercisable for a period of one year after the date of Optionee's death by Optionee's executor or administrator or the person or

persons to whom Optionee's rights under the Option are transferred by will or the laws of descent and distribution.

6. *Continued Exercise after Qualifying Retirement*. If Optionee retires from service as a director of STERIS (I) with the consent of or under guidelines approved by the Committee, (II) before **[Ten Years and One Month from Date of Grant]**, (III) after having attained age 55, and (IV) after having served as a director of STERIS for at least five consecutive years, as defined below (a retirement in such circumstances being a "Qualifying Retirement"), then, so long as Optionee remains in "Good Standing," as defined below Optionee will be entitled to exercise vested Options from time to time on any date during the period (the "Extended Exercise Period") that begins on the date of retirement and ends on the first to occur of (i) **[Ten Years and One Month from Date of Grant]** (which, as set forth in Section 6 above, is the scheduled termination date of the Option), and (ii) the fifth anniversary of the date of retirement.

If, at any time during the Extended Exercise Period, Optionee fails to remain in Good Standing, any and all options then outstanding and held by Optionee (whether granted under this or any other agreement) shall be forfeited and of no force or effect. For these purposes, in order for Optionee to remain in Good Standing during the Extended Exercise Period, Optionee must not violate either the Policy or any other STERIS policy and must refrain from acting in a manner detrimental to the interests of STERIS. If, at any time during the Extended Exercise Period, Optionee directly or indirectly materially competes with STERIS, Optionee will be deemed to have acted in a manner detrimental to the interests of STERIS and will, therefore, be deemed not to be in Good Standing for purposes of this Section 6.

If Optionee dies during the Extended Exercise Period and while in Good Standing, the Option will thereafter be exercisable to the same extent and at the same times (for so long and only so long after Optionee's death) as if Optionee had continued in service as a director of STERIS through the date of Optionee's death. Unless otherwise determined by the Committee, Optionee will be deemed to have "served as a director of STERIS for at least five consecutive years" only if Optionee served continuously as a director of STERIS throughout the five year period ending on the Service Termination Date.

7. *Method of Exercise*. The Option shall be exercised by delivery of (a) the Exercise Price payable in accordance with Section 6.4 of the Plan and (b) a written notice to STERIS identifying this Agreement and specifying the number of Common Shares as to which the Option is being exercised. STERIS shall deliver to Optionee certificates representing the Option shares as soon as administratively feasible following such exercise.

8. *Miscellaneous*. Nothing contained in this Agreement shall be understood as conferring on Optionee any right to continue as a Director of STERIS. This Agreement shall inure to the benefit of and be binding upon its parties and their respective heirs, executors, administrators, successors, and assigns, but the Option shall not be transferable by Optionee otherwise than by will or the laws of descent and distribution.

IN WITNESS WHEREOF, STERIS has caused this Agreement to be executed on its behalf by its duly authorized officer, and Optionee has hereunto set his hand, all as of the day and year first written above.

STERIS Corporation

Optionee

By: /s/ Mark D. McGinley

Mark D. McGinley Vice President, General Counsel, and Secretary [Director's Name]

EXHIBIT 15.1 LETTER REGARDING UNAUDITED INTERIM FINANCIAL INFORMATION

Board of Directors and Shareholders

STERIS Corporation

We are aware of the incorporation by reference in the following Registration Statements and related Prospectuses of our report dated January 24, 2005 relating to the unaudited consolidated interim financial statements of STERIS Corporation and subsidiaries that are included in its Form 10-Q for the quarter ended December 31, 2004:

Registration Number	Description
333-101308	Form S-8 Registration Statement — STERIS Corporation 2002 Stock Option Plan
333-91302	Form S-8 Registration Statement — Nonqualified Stock Option Agreement between STERIS Corporation and Mark D. McGinley
333-63774	Form S-8 Registration Statement — Nonqualified Stock Option Agreement between STERIS Corporation and Peter A. Burke
333-63770	Form S-8 Registration Statement — Nonqualified Stock Option Agreement between STERIS Corporation and Charles L. Immel and Restricted Shares Agreement between STERIS Corporation and Charles L. Immel
333-40082	Form S-8 Registration Statement — Nonqualified Stock Option Agreement between STERIS Corporation and Laurie Brlas and the Nonqualified Stock Option Agreement between STERIS Corporation and David L. Crandall
333-40058	Form S-8 Registration Statement — Nonqualified Stock Option Agreement between STERIS Corporation and Les C. Vinney
333-65155	Form S-8 Registration Statement — STERIS Corporation 1998 Long-Term Incentive Compensation Plan
333-55839	Form S-8 Registration Statement — Nonqualified Stock Option Agreement between STERIS Corporation and John Masefield and the Nonqualified Stock Option Agreement between STERIS Corporation and Thomas J. DeAngelo
333-32005	Form S-8 Registration Statement — STERIS Corporation 1997 Stock Option Plan
333-06529	Form S-3 Registration Statement — STERIS Corporation
333-01610	Post-effective Amendment to Form S-4 on Form S-8 — STERIS Corporation
33-91444	Form S-8 Registration Statement — STERIS Corporation 1994 Equity Compensation Plan
33-91442	Form S-8 Registration Statement — STERIS Corporation 1994 Nonemployee Directors Equity Compensation Plan
33-55976	Form S-8 Registration Statement — STERIS Corporation 401(k) Plan
33-55258	Form S-8 Registration — STERIS Corporation Amended and Restated Non-Qualified Stock Option Plan
	/s/ Ernst & Young LLP

Cleveland, Ohio

February 8, 2005

CERTIFICATION OF THE CHIEF EXECUTIVE OFFICER

- I, Les C. Vinney, President and Chief Executive Officer of STERIS Corporation ("registrant"), certify that:
 - 1. I have reviewed this quarterly report on Form 10-Q of STERIS Corporation;
 - 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
 - 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations, and cash flows of the registrant as of, and for, the periods presented in this report;
 - 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - c. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
 - 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize, and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: February 8, 2005

/s/ Les C. Vinney

Les C. Vinney President and Chief Executive Officer

CERTIFICATION OF THE CHIEF FINANCIAL OFFICER

I, Laurie Brlas, Senior Vice President and Chief Financial Officer of STERIS Corporation ("registrant"), certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of STERIS Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations, and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - c. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize, and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: February 8, 2005

/s/ Laurie Brlas

Laurie Brlas Senior Vice President and Chief Financial Officer

Certification Pursuant to § 906 of the Sarbanes-Oxley Act of 2002

Pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, in connection with the filing of the Form 10-Q of STERIS Corporation for the quarter ended December 31, 2004, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), each of the undersigned officers of the Company certifies, that, to such officer's knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company as of the dates and for the periods expressed in the Report.

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to STERIS Corporation and will be retained by STERIS Corporation and furnished to the Securities and Exchange Commission or its staff upon request.

/s/ Les C. Vinney

Name:Les C. VinneyTitle:President and Chief Executive Officer

/s/ Laurie Brlas

Name: Laurie Brlas

Title: Senior Vice President and Chief Financial Officer

Dated: February 8, 2005